A preliminary model-based assessment of the cost-utility of a screening programme for early age-related macular degeneration

J Karnon, C Czoski-Murray, K Smith, C Brand, U Chakravarthy, S Davis, N Bansback, C Beverley, A Bird, S Harding, I Chisholm and YC Yang

E

June 2008

Health Technology Assessment NHS R&D HTA Programme www.hta.ac.uk







How to obtain copies of this and other HTA Programme reports.

An electronic version of this publication, in Adobe Acrobat format, is available for downloading free of charge for personal use from the HTA website (http://www.hta.ac.uk). A fully searchable CD-ROM is also available (see below).

Printed copies of HTA monographs cost £20 each (post and packing free in the UK) to both public **and** private sector purchasers from our Despatch Agents.

Non-UK purchasers will have to pay a small fee for post and packing. For European countries the cost is $\pounds 2$ per monograph and for the rest of the world $\pounds 3$ per monograph.

You can order HTA monographs from our Despatch Agents:

- fax (with credit card or official purchase order)
- post (with credit card or official purchase order or cheque)
- phone during office hours (credit card only).

Additionally the HTA website allows you **either** to pay securely by credit card **or** to print out your order and then post or fax it.

Contact details are as follows:

HTA Despatch c/o Direct Mail Works Ltd 4 Oakwood Business Centre Downley, HAVANT PO9 2NP, UK Email: orders@hta.ac.uk Tel: 02392 492 000 Fax: 02392 478 555 Fax from outside the UK: +44 2392 478 555

NHS libraries can subscribe free of charge. Public libraries can subscribe at a very reduced cost of $\pounds 100$ for each volume (normally comprising 30–40 titles). The commercial subscription rate is $\pounds 300$ per volume. Please see our website for details. Subscriptions can only be purchased for the current or forthcoming volume.

Payment methods

Paying by cheque

If you pay by cheque, the cheque must be in **pounds sterling**, made payable to *Direct Mail Works Ltd* and drawn on a bank with a UK address.

Paying by credit card

The following cards are accepted by phone, fax, post or via the website ordering pages: Delta, Eurocard, Mastercard, Solo, Switch and Visa. We advise against sending credit card details in a plain email.

Paying by official purchase order

You can post or fax these, but they must be from public bodies (i.e. NHS or universities) within the UK. We cannot at present accept purchase orders from commercial companies or from outside the UK.

How do I get a copy of HTA on CD?

Please use the form on the HTA website (www.hta.ac.uk/htacd.htm). Or contact Direct Mail Works (see contact details above) by email, post, fax or phone. *HTA on CD* is currently free of charge worldwide.

The website also provides information about the HTA Programme and lists the membership of the various committees.

A preliminary model-based assessment of the cost-utility of a screening programme for early age-related macular degeneration

J Karnon,^{1*} C Czoski-Murray,¹ K Smith,¹ C Brand,² U Chakravarthy,³ S Davis,¹ N Bansback,¹ C Beverley,¹ A Bird,⁴ S Harding,⁵ I Chisholm⁶ and YC Yang⁷

- ¹ School of Health and Related Research (ScHARR), University of Sheffield, UK
- ² Sheffield University Teaching Hospitals, Sheffield, UK
- ³ Royal Victoria Hospital, Belfast, UK
- ⁴ Moorfields Hospital, London, UK
- ⁵ Royal Liverpool University Hospital, Liverpool UK
- ⁶ Southampton University Hospitals NHS Trust, Southampton, UK
- ⁷ Wolverhampton Eye Infirmary, Wolverhampton, UK

* Corresponding author. Present address: School of Population Health and Clinical Practice, University of Adelaide, Australia

Declared competing interests of authors: C Czoski-Murray, C Brand, S Davis and N Bansback have received funding from Novartis Pharma for research. A Bird has been an advisor to Genentec, Geneera, Ictherapy, Eyecyte and Pfizer. C Brand has been an advisor to Pfizer.

Published June 2008

This report should be referenced as follows:

Karnon J, Czoski-Murray C, Smith K, Brand C, Chakravarthy U, Davis S, *et al*. A preliminary model-based assessment of the cost–utility of a screening programme for early age-related macular degeneration. *Health Technol Assess* 2008;**12**(27).

Health Technology Assessment is indexed and abstracted in Index Medicus/MEDLINE, Excerpta Medica/EMBASE and Science Citation Index Expanded (SciSearch[®]) and Current Contents[®]/Clinical Medicine.

NIHR Health Technology Assessment Programme

The Health Technology Assessment (HTA) Programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The research findings from the HTA Programme directly influence decision-making bodies such as the National Institute for Health and Clinical Excellence (NICE) and the National Screening Committee (NSC). HTA findings also help to improve the quality of clinical practice in the NHS indirectly in that they form a key component of the 'National Knowledge Service'.

The HTA Programme is needs-led in that it fills gaps in the evidence needed by the NHS. There are three routes to the start of projects.

First is the commissioned route. Suggestions for research are actively sought from people working in the NHS, the public and consumer groups and professional bodies such as royal colleges and NHS trusts. These suggestions are carefully prioritised by panels of independent experts (including NHS service users). The HTA Programme then commissions the research by competitive tender.

Secondly, the HTA Programme provides grants for clinical trials for researchers who identify research questions. These are assessed for importance to patients and the NHS, and scientific rigour.

Thirdly, through its Technology Assessment Report (TAR) call-off contract, the HTA Programme commissions bespoke reports, principally for NICE, but also for other policy-makers. TARs bring together evidence on the value of specific technologies.

Some HTA research projects, including TARs, may take only months, others need several years. They can cost from as little as £40,000 to over £1 million, and may involve synthesising existing evidence, undertaking a trial, or other research collecting new data to answer a research problem.

The final reports from HTA projects are peer-reviewed by a number of independent expert referees before publication in the widely read journal series *Health Technology Assessment*.

Criteria for inclusion in the HTA journal series

Reports are published in the HTA journal series if (1) they have resulted from work for the HTA Programme, and (2) they are of a sufficiently high scientific quality as assessed by the referees and editors. Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search,

appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

The research reported in this issue of the journal was commissioned by the HTA Programme as project number 03/06/01. The contractual start date was in April 2004. The draft report began editorial review in March 2006 and was accepted for publication in February 2008. As the funder, by devising a commissioning brief, the HTA Programme specified the research question and study design. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the referees for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

The views expressed in this publication are those of the authors and not necessarily those of the HTA Programme or the Department of Health.

Editor-in-Chief:	Professor Tom Walley
Series Editors:	Dr Aileen Clarke, Dr Peter Davidson, Dr Chris Hyde,
	Dr John Powell, Dr Rob Riemsma and Professor Ken Stein
Programme Managers:	Sarah Llewellyn Lloyd, Stephen Lemon, Kate Rodger,
8	Stephanie Russell and Pauline Swinburne

ISSN 1366-5278

© Queen's Printer and Controller of HMSO 2008

This monograph may be freely reproduced for the purposes of private research and study and may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising.

Applications for commercial reproduction should be addressed to: NCCHTA, Alpha House, Enterprise Road, Southampton Science Park, Chilworth, Southampton SO16 7NS, UK.

Published by Gray Publishing, Tunbridge Wells, Kent, on behalf of NCCHTA.

Printed on acid-free paper in the UK by St Edmundsbury Press Ltd, Bury St Edmunds, Suffolk.



A preliminary model-based assessment of the cost-utility of a screening programme for early age-related macular degeneration

J Karnon,^{1*} C Czoski-Murray,¹ K Smith,¹ C Brand,² U Chakravarthy,³ S Davis,¹ N Bansback,¹ C Beverley,¹ A Bird,⁴ S Harding,⁵ I Chisholm⁶ and YC Yang⁷

¹ School of Health and Related Research (ScHARR), University of Sheffield, UK

² Sheffield University Teaching Hospitals, Sheffield, UK

³ Royal Victoria Hospital, Belfast, UK

- ⁴ Moorfields Hospital, London, UK
- ⁵ Royal Liverpool University Hospital, Liverpool UK
- ⁶ Southampton University Hospitals NHS Trust, Southampton, UK
- ⁷ Wolverhampton Eye Infirmary, Wolverhampton, UK

* Corresponding author. Present address: School of Population Health and Clinical Practice, University of Adelaide, Australia

Objectives: To estimate the cost-effectiveness of screening for age-related macular degeneration (AMD) by developing a decision analytic model that incorporated and assessed all of the National Screening Committee criteria. A further objective was to identify the major areas of uncertainty in the model, and so inform future research priorities in this disease area. Data sources: Major databases were searched in March 2004 and updated in January 2005. **Review methods:** Systematic literature reviews covered the epidemiology and natural history of AMD, the screening and treatment effectiveness and healthrelated quality of life relating to AMD. A hybrid cohort-individual sampling model was implemented to describe the range of pathways between the incidence of age-related maculopathy (ARM) and death via clinical presentation and treatment at different stages of the disease. As significant shortfalls in the data available from the literature were apparent, so a range of primary data sources were also used to populate the model. To obtain estimates for the value of parameters deemed to be within an expert's remit, data describing some parameters were elicited from relevant experts. The data identified informed probability distributions describing the uncertainty around the model parameters. To incorporate joint parameter uncertainty (i.e. correlations between parameters), the AMD natural history model was calibrated probabilistically. Randomly sampled sets of input parameters were assigned weights representing the accuracy of their predictions of a set of observed model outputs. The

analysis of the AMD screening model estimated the costs, numbers of quality-adjusted life-years (QALYs) and cases of blindness in a general population sample of 50-year-olds over the remainder of their lifetime, for 16 alternative screening options (including no screening). The reference case analysis incorporated current treatment options of laser photocoagulation and photodynamic therapy. Sensitivity analyses describing six alternative sets of intervention strategies, based on horizon scanning of potential future treatments for AMD, were also undertaken. **Results:** There remains significant uncertainty about whether any form of screening for AMD is costeffective. However, annual screening from age 60 years seems to provide the highest mean net benefits, but this is based on a cost-effectiveness estimate that has very poor precision (high levels of uncertainty). The probabilistic sensitivity analysis shows that the 95% credible interval for annual screening from age 60 years ranges from this option dominating the previous option to an incremental cost per QALY of over £0.5 million. Plotting a cost-effectiveness acceptability frontier shows that although annual screening from age 60 years has the highest net benefits at a value of QALY of £30,000, the associated probability of this option being the most cost-effective option is only around 20%. The sensitivity analyses around potential future treatment options indicate that screening may become more costeffective with the new treatments. Conclusions: The conclusions focus on the interpretation of the results from the perspective of

defining the major areas of uncertainty, which were defined as disease progression, rates of clinical presentation, screening test and optician effectiveness, treatment effectiveness, and costs of blindness. Future research may be best targeted at assessing how routine data may be used to describe clinical presentation rates of ARM. Other potential studies include a pilot study of the effectiveness of screening and opticians' referral patterns for AMD and a costing study of blindness as a continuum of association with deterioration in vision.



	List of abbreviations	vii
	Executive summary	ix
I	Introduction	1
	Epidemiology	1
	General rationale for screening	1
	Study aims and objectives	2
	Research methods	2
	Literature review methods	3
	Inclusion and exclusion criteria	4
2	The AMD screening model	5
	General framework	5
	Lesion states	6
	Lesion progression	6
	Modelling clinical presentation and screen	
	detection	7
	Modelling visual acuity	8
	Modelling treatment	8
	Modelling costs	9
3	Natural history parameters	11
	Risk factors	11
	Incidence	12
	Natural history	24
	Progression to subfoveal wet AMD	31
	Visual acuity	32
4	Screening and treatment parameters	37
	Screening parameters	37
	Treatment parameters	40
5	Costs and utility parameters	47
•	Screening diagnosis and surveillance	17
	costs	47
	Treatment costs	48
	Costs associated with blindness and rapidly	10
	deteriorating vision	50
	Utility values	50
	Curry values	50
6	Model implementation and analysis	53

	Calibration outputs Calibration analyses Calibration results Model analysis Future treatment options sensitivity	53 56 57 58
	analyses	59
7	Results Reference case results Sensitivity analyses results Summary	61 61 68 70
8	Discussion Areas of uncertainty Conclusions	73 74 77
	Acknowledgements	79
	References	81
	Appendix I Literature search plans and results	87
	Appendix 2 Studies of prevalence of AMD	91
	Appendix 3 Expert elicitation questionnaire	115
	Appendix 4 Survey of ophthalmologists	121
	Appendix 5 National Screening Committee's criteria for appraising the viability, effectiveness and appropriateness of a screening programme	198
	Health Technology Assessment reports published to date	125
	Health Technology Assessment Programme	143

List of abbreviations

AMD	age-related macular degeneration	EBRT	external beam radiation therapy	
ANCHOR	NCHOR ANti-VEGF Antibody for the Treatment of Predominantly Classic CHORoidal Neovascularization in AMD		extrafoveal	
			EuroQol 5 Dimensions	
anti-VEGF	anti-vascular endothelial-derived	EVI	expected value of information	
AREDS	Age-Related Eye Disease Study	FOCUS (trial)	RhuFab V2 Ocular Treatment Combining the Use of Verteporfin to Evaluate Safety	
AREDSRG	Age-Related Eye Disease Study Research Group	GA	geographic atrophy	
ARM	age-related maculopathy	GAMD	geographical age-related macular	
ARR	absolute risk reduction		degeneration	
BCVA	best corrected visual acuity	HRQoL	health-related quality of life	
BMD	bilateral macular drusen	HUI-3	Health Utility Index 3	
BMES	Blue Mountains Eye Study	ICER	incremental cost-effectiveness	
CDSR	Cochrane Database of Systematic Reviews	IVSI	intra-vitreal steroid injections	
CEAC	cost-effectiveness acceptability curve	JF	juxtafoveal	
CENTRAL	Cochrane Central Register of	MC	minimally classic	
	Controlled Trials	MPS	Macular Photocoagulation Study	
CI	confidence interval	NICE	National Institute for Health and	
CNV	choroidal neovascularisation		Clinical Excellence	
CNVPT	CNV Prevention Trial	NSC	National Screening Committee	
CS	contrast sensitivity	NVAMD	neovascular age-related macular	
DARE	Database of Abstracts of Reviews		degeneration	
	of Effects	OR	odds ratio	
DLTV	daily living tasks dependent on vision	PC	predominantly classic	

continued

viii

List of a	List of abbreviations continued						
PDT	photodynamic therapy	SFRADS	SubFoveal Radiotherapy Study				
QALY	quality-adjusted life-year	SSCI	Social Sciences Citation Index				
QoL	quality of life	ТАР	Treatment of AMD with Photodynamic Therapy				
RAP	retinal angiomatous proliferation	TRIP	Turning Research into Practice				
ROT	randomised controlled trial	ТТО	time trade-off				
RR	relative risk	TTT	trans-papillary thermo-therapy				
SAILOR	Safety Assessment Intravitreal	VA	visual acuity				
	Ranubizimab for AMD	VEGF	vascular endothelial growth factor				
SCI	Science Citation Index	VIP	Verteporfin in Photodynamic Therapy				
SF	subfoveal	VISION	VEGF Inhibition Study in Ocular				
SF-6D	Short Form 6 Dimensions		Neovascularisation				
A 11 - 1-1	and the former former and the shift many set and first different former and the state of the sta	hono unloss the1	Lucietien is seell her seen (s. s. NUTC) and				

All abbreviations that have been used in this report are listed here unless the abbreviation is well known (e.g. NHS), or it has been used only once, or it is a non-standard abbreviation used only in figures/tables/appendices in which case the abbreviation is defined in the figure legend or at the end of the table.

Executive summary

Introduction

Age-related macular degeneration (AMD) is the leading cause of blindness in people aged over 60 years in the western world. It is estimated that around 25% of the over-60s in the UK have some degree of visual loss because of AMD. AMD is divided into early and late. In early disease, visual acuity is normal. Late disease consists of choroidal neovascularisation (wet) and geographic atrophy (dry). Treatment options for dry AMD are still at a relatively early stage of evaluation and at present no treatment for dry AMD is routinely available. Wet AMD occurs as a result of bleeding and scarring and leads to more rapid sight loss. Current treatments for wet AMD seek to prevent further visual loss rather than cure or restore vision, although newer interventions have shown promising results that indicate that vision improvement may be possible.

The National Screening Committee have defined a set of criteria to inform the suitability of screening for a condition. The condition must be important, and the natural history and epidemiology must be understood. The screening test should be simple, safe, precise and acceptable to the general population, and there should be a defined diagnostic process following a positive test. Treatment for screen-detected disease should lead to better outcomes than treatment provided at the point of clinical diagnosis.

Aims and objectives

The aim of this study was to estimate the costeffectiveness of screening for AMD by developing a decision analytic model that incorporated and assessed all of the above criteria. At the outset it was recognised that there was likely to be significant uncertainty in key areas of the model, and an objective of the study was to identify the major areas of uncertainty, and so inform future research priorities in this disease area.

Methods

Systematic literature reviews of the major electronic databases took place in March 2004 and were updated in January 2005. These reviews covered the epidemiology and natural history of AMD, the screening and treatment effectiveness and health-related quality of life relating to AMD. A hybrid cohort–individual sampling model was implemented to describe the range of pathways between the incidence of age-related maculopathy (ARM) and death via clinical presentation and treatment at different stages of the disease.

Significant shortfalls in the data available from the literature were apparent, so a range of primary data sources were also used to populate the model. To obtain estimates for the value of parameters deemed to be within an expert's remit, data describing some parameters were elicited from relevant experts. The data identified informed probability distributions describing the uncertainty around the model parameters.

To incorporate joint parameter uncertainty (i.e. correlations between parameters), the AMD natural history model was calibrated probabilistically. Randomly sampled sets of input parameters were assigned weights representing the accuracy of their predictions of a set of observed model outputs.

The analysis of the AMD screening model estimated the costs, numbers of quality-adjusted life-years (QALYs) and cases of blindness in a general population sample of 50-year-olds over the remainder of their lifetime, for 16 alternative screening options (including no screening). The reference case analysis incorporated current treatment options of laser photocoagulation and photodynamic therapy. Sensitivity analyses describing six alternative sets of intervention strategies, based on horizon scanning of potential future treatments for AMD, were also undertaken.

Results

There remains significant uncertainty about whether any form of screening for AMD is costeffective. However, annual screening from age 60 years seems to provide the highest mean net benefits, but this is based on a cost-effectiveness estimate that has very poor precision (high levels of uncertainty). The probabilistic sensitivity analysis shows that the 95% credible interval for annual screening from age 60 years ranges from this option dominating the previous option to an incremental cost per QALY of over £0.5 million. Plotting a cost-effectiveness acceptability frontier shows that although annual screening from age 60 years has the highest net benefits at a value of QALY of £30,000, the associated probability of this option being the most cost-effective option is only around 20%.

The sensitivity analyses around potential future treatment options indicate that screening may become more cost-effective with the new treatments.

Conclusions

The conclusions focus on the interpretation of the results from the perspective of defining the major areas of uncertainty, which were defined as:

- Disease progression (due to the available data, the model was built around progression of visual acuity, despite a preference for contrast sensitivity).
- Rates of clinical presentation (informed by local data from the Sheffield photodynamic therapy (PDT) clinic and responses from a survey of

general ophthalmologists). Problems with this approach included a small sample of patients, the fact that the PDT database was not validated, a limited response to the survey of ophthalmologists and inconsistencies in the responses received.

- Screening test and optician effectiveness (elicited data described the probability that individuals undertaking the simple screening test at home who notice an abnormality would then present at an optician's). The model assumes that optometrists accurately refer all cases of dry and wet AMD on to hospital ophthalmologists, while not referring any cases of early ARM.
- Treatment effectiveness (a lack of long-term follow-up data inevitably requires the use of weak assumptions to extrapolate the observed effectiveness data).
- Costs of blindness (a binary threshold for costs associated with blindness was incorporated, but such costs would be more appropriately described on a continuum).

Future research may be best targeted at assessing how routine data may be used to describe clinical presentation rates of ARM. Other potential studies include a pilot study of the effectiveness of screening and opticians' referral patterns for AMD and a costing study of blindness as a continuum of association with deterioration in vision.

Chapter I Introduction

ge-related macular degeneration (AMD) is a ${
m A}$ progressive and degenerative disease of the retina which may ultimately lead to blindness. AMD is divided into early and late. In early disease, visual acuity (VA) is normal. Late disease consists of choroidal neovascularisation (wet) and geographic atrophy (dry).¹ Wet AMD involves the vascular membrane from the choroid sprouting under and through the retinal pigment epithelium (RPE) and Bruch's membrane to spread beneath the retina. The leaking and bleeding from these vessels cause exudative or haemorrhagic retinal detachments or both.² The bleeding results in scarring that replaces the outer layers of the retina, which is the cause of the loss of visual function. The cause is still unknown, and most treatment options seek to arrest decline rather than cure or restore vision.

Epidemiology

The Framingham Eye Study in 1980 stated that although only 10% of the cases of AMD were wet, this form of the disease accounted for 80% of cases with severe visual loss.³ However, Owen and colleagues⁴ challenge this view as they estimated the prevalence of wet AMD to be higher than that of dry AMD (245,000 versus 172,000 in the UK). It is thought that studies identifying dry AMD as the more common form of late AMD have usually misidentified early AMD as dry AMD [information provided by one of the authors (AB)].

As the current study is concerned with populationbased screening for AMD, we looked mainly for population-based epidemiological studies which were either UK centred or were similar to the UK population. There is some evidence to suggest that AMD is more prevalent in white European populations and less common in black⁵ or Chinese⁶ populations. These distinctions are less important in the context of screening in the UK, as the prompt to take part in any programme would be related to age and not to ethnicity.

The incidence of AMD increases with the age of the population, and the majority of cases are found in the over-60 age group. It is not uncommon to see changes within the eye in the 50–60 age group, which could indicate that AMD may develop, even though there are no symptoms of vision loss at this stage.

The early stage of this disease is also referred to as age-related maculopathy (ARM). The appearance of drusen, which are deposits found between the RPE and Bruch's membrane, of the hard type is not necessarily associated with worrying pathological changes; indeed, drusen were noted to disappear in the Waterman study.⁷ The less defined and less common soft drusen are thought to be more indicative of the development of a more serious maculopathy.²

General rationale for screening

The purpose of screening is to identify individuals who have a greater or lesser risk of developing a particular condition. A screening programme should alert those at greater risk to seek further investigation and reassure those with a lower risk. There are well-established criteria for screening programmes,⁸ which have been adapted by the UK National Screening Committee (NSC)⁹ to guide the provision of screening programmes in general, and also to inform the specification of accepted screening programmes. The criteria address four broad factors: the condition, the test, the treatment and the screening programme.

The condition, it is stated, must be important, and the natural history and epidemiology must be understood. The screening test should be simple, safe, precise and acceptable to the general population, and there should be a defined diagnostic process following a positive test. Treatment for screen-detected disease should lead to better outcomes than treatment provided at the point of clinical diagnosis. Regarding the screening programme as a whole, it is stated that plans for monitoring the programme should be defined, adequate staffing and facilities should be available to cope with expected demand and the programme should provide value for money, as compared with other areas of medical expenditure.

The last criterion states the need for screening to be cost-effective and implicitly, if screening is cost-effective, that the most cost-effective form of screening should be implemented. Each of the preceding criteria describes factors that must be defined in order to estimate the cost-effectiveness of screening, or that will permit the confirmation of cost-effectiveness, namely monitoring. The criteria recognise the need for cost-effectiveness to be defined in terms of a generic outcome measure to allow comparison with other areas of medical expenditure, which in practice requires the estimation of quality-adjusted life-years (QALYs).

The criteria also state that "there must be evidence from high quality Randomised Controlled Trials that the screening programme is effective in reducing mortality or morbidity". If such data were available, one could assess the cost-utility of alternative screening programmes alongside the relevant trials. However, there are areas in which no clinical trial evidence is available, for example, cervical cancer screening, and it is always the case that trial evidence is not available to inform all possible specifications of a screening programme, for example, with respect to the eligible population, the combination of screening tests, the interval between screening rounds and the issues over the generalisability of trial results.

It is necessary, therefore, to use some form of modelling to inform cost–utility analyses of screening programmes.

Rationale for screening for AMD

The purpose of this research was to define, with the attendant uncertainty, the extent to which screening for AMD meets the NSC's criteria. To that end, the research addressed each criterion.

Study aims and objectives

This study aimed to examine the three broad issues around the evaluation of any screening programme, relating to the natural history of the disease, the relative effectiveness of early versus late treatment and the diagnostic accuracy of the screening test in identifying early cases of the disease. In all three of these areas there exists significant uncertainty around AMD.

Our objective was to:

1. Develop a model of natural history for each relevant subcategory of the disease, that is, for dry and wet AMD, in addition to differentiating between the alternative specifications of the wet form of the disease [classic and occult, and subfoveal (SF), juxtafoveal (JF) and extrafoveal (EF)]. The description of natural history required estimates of the likelihood and timing of loss of vision in both eyes, and the impact of AMD (and loss of vision) on people's quality of life (QoL) (utility).

- 2. Analyse the effectiveness of alternative therapy options for alternative forms of AMD, by estimating the impact of the therapies on preventing progression, and possibly encouraging regression of the disease.
- 3. Identify a range of potential screening programmes, and with clinical input further evaluate those programmes that appear to be the most viable, effective and cost-effective options using the defined decision analytic modelling framework.

Research methods

The planned methods for the evaluation of a screening programme for early AMD consisted of the following main sections, which are presented in order of their presentation in the report.

Chapter 2 describes the process of development and the final structure of the model used to estimate the cost-effectiveness of screening for AMD. The model structure was iteratively developed based on the findings from the literature review and discussions with expert ophthalmologists. This chapter is presented before the review and elicitation chapters to provide the reader with a reference point for the conduct and analysis of the literature review.

Chapters 3, 4 and 5 present the results of the literature reviews that were undertaken for each of the following areas:

- 1. observational evidence and no treatment arms of clinical trials to describe the natural history of the different forms of AMD (Chapter 3)
- 2. trial evidence, clinical guidelines and systematic reviews on the effectiveness of alternative treatment options for AMD (Chapter 4)
- evidence describing the diagnostic accuracy of alternative screening tests for AMD (Chapter 4)
- 4. evidence on the acceptability of screening tests in general, and for AMD in particular (Chapter 4)
- 5. empirical estimates of health-related quality of life (HRQoL) and patient utilities relating to AMD and screening (Chapter 5).

The analysis of the natural history of AMD was significantly enhanced by the availability of several primary datasets that contained more information than available in the secondary literature. These datasets, and their analysis, are also described in Chapter 3.

In addition to the results of the literature review for screening and treatment parameters, Chapter 4 describes the conduct and outputs from a primary elicitation study that obtained parameter estimates from experts in the field of AMD, mainly for the model's screening parameters. The parameters included in the elicitation study were those that were not adequately informed by the literature, but which were absolutely necessary to estimate the costeffectiveness of screening for AMD.

A full systematic review of treatment effectiveness was not undertaken because the study had access to the patient-level data from the two pivotal trials of the predominant current treatment options for AMD. Data from the Treatment with AMD with Photodynamic Therapy (TAP) and Verteporfin in Photodynamic Therapy (VIP) trials of photodynamic therapy (PDT) were analysed to estimate the effectiveness parameters for PDT. Evidence relating to other current treatment options, primarily laser photocoagulation, was limited to high-quality systematic reviews. Future, potential treatment options were also included in the analysis, which were informed by the ophthalmologist authors of this report, with data obtained through a nonsystematic review of key data sources, primarily the Internet.

Chapter 5 describes the utility values used in the model, but also reports on the data sources and assumptions used to populate the model's screening and treatment cost parameters.

Chapter 6 describes the process of implementing and calibrating the model against estimates of some key model outputs (e.g. age-specific numbers of AMD cases presenting clinically). The calibration process accounts for the potential correlation of all the model's input parameters by assigning probabilities to full sets of input parameters, reflecting the likelihood that each set is the most accurate set of parameter values.

Chapter 7 reports the results of the reference case probabilistic analysis, and also a number of sensitivity analyses that incorporate alternative scenarios. Chapter 8 interprets the methods and results of the study to identify the key areas of uncertainty by identifying the input parameters that had the greatest effect on the results, and at which future research should be targeted.

Literature review methods

Separate systematic reviews of the literature were undertaken in the following areas relating to AMD: natural history, epidemiology, risk factors, screening tests, treatments and QoL. The results of these reviews are reported in separate chapters, but the methodologies for each review were similar and so are described here.

Search strategies

A comprehensive literature search was undertaken during March 2004 (updated in January 2005) to identify relevant literature pertaining to screening for AMD (Appendix 1). Four major searches were conducted which were designed to retrieve:

- high-level evidence [i.e. guidelines, systematic reviews and randomised controlled trials (RCTs)] concerning AMD
- 2. papers describing the epidemiology of AMD
- 3. papers describing the diagnosis and diagnostic tests associated with AMD
- 4. cost-effectiveness and health utility literature in the field.

The following electronic bibliographic databases were searched:

- Cochrane Database of Systematic Reviews (CDSR)
- Cochrane Central Register of Controlled Trials (CENTRAL)
- EMBASE
- MEDLINE
- NHS Database of Abstracts of Reviews of Effects (DARE)
- NHS Health Technology Assessment (HTA) database
- Science Citation Index (SCI)
- Social Sciences Citation Index (SSCI)
- Turning Research into Practice (TRIP) database.

Attempts were also made to identify 'grey' literature by searching appropriate databases (e.g. Health Management Information Consortium, Index to Theses, Dissertation Abstracts), current research registers (e.g. National Research Register, Research Findings Register, Current Controlled Trials) and relevant websites (e.g. Macular Disease Society, Moorfields Eye Hospital, Royal National Institute for the Blind). The reference lists of included studies and relevant review articles were also checked.

Inclusion and exclusion criteria

The team devised the criteria for inclusion and exclusion. References were retrieved by one reviewer selecting from a specific subject heading and then checked by a second reviewer.

In the first round of sifting the retrieved references, we used the following criteria; as the work progressed, it became apparent that there were significant gaps in the literature in some of these categories, which we will report later.

The searches were conducted as described above. When the team began sifting the papers by category it became evident that there was often a mismatch between the key word description and, for example, the abstract. We made a decision to re-classify where possible as we sifted. The papers which we identified as being potentially useful in this review were separated into the categories listed below. We found this to be the most efficient method of making use of papers which did not match their search criteria or were relevant in more than one category.

- 1. Prevalence and incidence
 - (a) Inclusion: primary research, systematic review or high-quality review
 representative populationbased sample
 (b) Exclusion: non-AMD
- 2. Risk factors

(a)	Inclusion:	primary research, systematic
		review or high-quality
		review
(b)	Exclusion:	non-AMD
		risk factors not applicable to a
		screening programme, e.g.
		genetic risk markers
		pathogenesis studies

- 3. Natural history progression
 - (a) Inclusion: primary research, systematic review or high-quality review
 data reported, including baseline measurement and measurement(s) at subsequent time point(s)
 - (b) Exclusion: non-AMD drusen-based prognosis highly selected population
- 4. Quality of life studies
 - (a) Inclusion: primary research utility data or appropriate HRQoL measures used data relates to alternative states of visual impairment
- 5. Economic studies (a) Inclusion: screening evaluations resource use and cost data presented
- 6. Screening studies

(a)	Inclusion:	potential screening test
		data reported on sensitivity
		and specificity

7. Diagnostic test studies

(a) Inclusion: potential diagnostic test data reported: accuracy, cost oftests.

The search resulted in 388 studies. The number of studies identified in each category is presented in Appendix 1. The data were extracted by one reviewer and checked by a second reviewer using specially developed data extraction tables.

Chapter 2 The AMD screening model

ecision modelling is about representing a disease or condition as a number of health states. Important clinical events are demonstrated by transitions between these states. The preferred, and most common, modelling approach to evaluating screening programmes is to describe the progression of disease in a population without screening, that is the natural history of the disease to the point of clinical presentation, followed by the post-diagnosis (and treatment) prognosis of the disease given current treatment options. Disease progression is generally described with respect to factors that influence treatment eligibility and treatment effectiveness. Possible screening programme options are then 'laid on top' of the natural history model in order to predict the effects of screening on disease progression via the potential diagnosis of the condition at earlier stages.

All models are simplifications of reality, which may be due to an incomplete understanding of the process being modelled (e.g. the pathogenesis of a disease) or practical constraints on the level of detail that can be incorporated within a computerbased model. The more complex the model, the more time is required to build, verify and analyse the model. Given a fixed period for a model-based evaluation, the general modelling objective is to develop a feasible model structure that is based on as strong a set of assumptions as possible.

General framework

The development of the model structure for the evaluation of screening for AMD was refined in an iterative process that involved frequent conversation with expert ophthalmologists, improving understanding of the disease area through review of the literature and assessment of optimal methods for incorporating the available data for populating or calibrating the model.

The aim of the AMD model is to describe the progression of both eyes with respect to AMD over the lifetime of a general population. The principal outcome measure for the evaluation of screening for AMD is the QALY, which is estimated as a function of the VA of the best seeing eye. Costs are associated with the act of screening, diagnosis, treatment for ARM and the consequences of blindness. Therefore, the model must describe the VA of both eyes at all ages and differentiate between alternative states of ARM that affect treatment eligibility and treatment effectiveness. It was recognised that current treatment options for AMD are limited, but the developed model was intended to provide a tool for updating estimates of the cost-effectiveness of screening as and when new treatment options become available. A range of baseline assumptions are stated to place a boundary around the scope of the model:

- No-one under the age of 50 years develops any form of ARM.
- All individuals in a population have two eyes that are at risk of developing ARM.
- There are no practical risk factors that could be used to define a population subgroup at whom screening could be targeted.
- Individuals live to a maximum of 100 years.
- ARM does not affect life expectancy; all individuals have the same probability of dying at any given age.
- The model moves forward in cycles of 1 year.

The above assumptions inform a model structure that describes the progression of two eyes in a homogeneous set of individuals from the age of 50 years. The homogeneity refers to the probabilities of disease incidence and progression, not to the actual experience of disease.

Risk factors for AMD are identifiable (e.g. smoking), but it is not practically or ethically feasible to target screening based on risk factors. It may be hypothesised that individuals in different risk factor subgroups may have alternative propensities to screen for AMD, or to comply with treatment, but given the scarcity of data to inform the additional input parameters that would be required to incorporate such hypotheses, the research team decided against their inclusion.

The assumption of constant life expectancy is a conservative assumption, but because the evidence regarding the impact of sight on mortality is not definitive, we felt it to be a suitable assumption. The model's cycle length of 1 year was chosen for pragmatic reasons to ease the implementation of the model. Although it is recognised that ARM may progress between the defined health states (see the next section) more than once within a year, the model allows for transitions between nonsequential vision states, and the estimated transition probabilities account for the annual cycles. The main impact is that the model may overestimate the time between transitions slightly, but this is unlikely to have a material effect on the model's results.

Lesion states

The next stage of the development of the model structure involved the definition of the disease states to be included. The three broad stages of ARM may be defined as early ARM, dry AMD and wet AMD, although within each of these stages multiple and varying subcategories have been defined in the literature. The following ARM states are described within the model:

- early ARM
- dry AMD
- EF occult wet AMD
- EF minimally classic (MC) wet AMD
- EF predominantly classic (PC) wet AMD
- JF occult wet AMD
- JF MC wet AMD
- JF PC wet AMD
- SF occult wet AMD
- SF MC wet AMD
- SF PC wet AMD.

A range of other options was discussed by the research team, such as the subcategorisation of early ARM, which is commonly disaggregated in the literature; for example, the Rotterdam study defines five separate categories of early ARM, ranging from the presence of soft distinct drusen to the presence of soft, indistinct or reticular drusen with pigmentary irregularities.² However, there were few data describing progression rates between substages of early ARM, or between substages of early ARM and the AMD states. Also, the limited evidence around potential treatment options for early ARM did not distinguish between different stages of early ARM.

The main option for subcategorising dry AMD was into two states describing dry AMD affecting the fovea and dry AMD not affecting the fovea. However, few studies differentiate between these states, with respect to disease progression to wet AMD, or with regard to potential treatment options.

The description of alternative wet AMD states is most detailed because the type (occult, MC or PC) and location (EF, JF, or SF) of wet AMD is important in determining treatment options and treatment effectiveness. However, additional subcategories of wet AMD could have been included in the model. Predominantly classic lesions may be defined as either 100% classic or PC, although these lesion types were combined as treatment eligibility and effectiveness is generally defined for the combined group of lesions.

The terms classic and occult are clinical, morphological descriptions defined by fluorescein angiography characteristics. A broader spectrum of occult wet AMD may be defined to include entities such as polypoidal choroidal neovascularisation and retinal angiomatous proliferation (RAP).¹⁰ RAP is associated with proliferation of intraretinal capillaries in the paramacular area and a contiguous telangiectatic response, which itself has three defined stages in a progressive vasogenic sequence. At this stage, no data on differential rates of disease progression and treatment effectiveness for RAP and non-RAP occult lesions were identified, so the model retains the single category of occult lesions.

The primary additional factor that was considered for inclusion in the AMD screening model was lesion size. Lesion size has been defined as a potentially relevant criterion informing treatment effectiveness (PDT may be less effective in larger lesions), and some primary data were identified that described lesion size progression. However, analyses of the data informing disease progression of wet AMD (described in more detail in Chapter 3, 'Visual acuity', pp. 33-34) consistently identified size as a statistically insignificant covariate when estimating deterioration in VA for individuals with wet AMD. A lack of data describing the size of lesions at the point of clinical presentation also reduced the usefulness of including lesion size as a further prognostic indicator in the model and, as the inclusion of size would increase the complexity of the model significantly, the decision was taken to restrict the categorisation of wet AMD to the above-listed states.

Lesion progression

Figure 1 describes the possible pathways of progression between the main categories of ARM



FIGURE I Lesion pathways between broad ARM categories

states: early ARM, dry AMD and wet AMD. The figure shows that the model does not assume a strict linear pathway where early ARM is the precursor to all subsequent ARM states. This approach is due to theoretical uncertainties about the true pathogenesis of ARM as elaborated by the expert ophthalmologist members of the research team, but also owing to practical considerations due to the format of the data available to populate the model. Even if all dry and wet AMD pass through an early ARM stage, some of these transitions are not observed by prospective studies that examine populations at sequential time points. The reported direct incidence rates for dry and wet AMD are used in the model to represent direct transitions from 'no ARM' to dry or wet AMD. Similarly, direct transitions from early ARM to wet

AMD are reported in the literature, and used directly to populate the AMD screening model.

The same lesion pathways are used to describe lesion incidence and progression in both eyes, although incident ARM may only be described in eye 2 from the age at which eye 1 first develops some form of ARM.

Figure 2 describes the model's representation of possible pathways between alternative states of wet AMD. Patients may enter the wet AMD model in any of the nine vision states, from which point the possible pathways that an eye may follow are demonstrated by the presented arrows. The assumptions incorporated in the structure include:

- Lesion location may change from EF to JF or SF or from JF to SF.
- Lesion type may change from occult to MC or PC or from minimally classic to PC.
- Lesion type and location may change separately or simultaneously.

Modelling clinical presentation and screen detection

The model assumes that the impact of any costs and benefits associated with individuals presenting with visual symptoms unrelated to AMD are neutral, that is, the costs equal the benefits. Screening may, for example, lead to an increase in the number of individuals undergoing NHSfunded eye examinations, which would lead to increased resources being spent on providing eye



FIGURE 2 Lesion pathways between wet AMD lesion states

examinations. However, it is assumed that eye examinations are provided because they lead to some form of benefit to the population. As no readily available data were identified describing the costs and benefits of such interventions, and primary investigation of such effects was beyond the scope of this study, the simplifying assumption of equal costs and benefits was made. Therefore, the clinical presentation of individuals without any form of ARM is not described by the model.

Individuals with one or both eyes with any form of ARM may present in the year of incidence, or in any year in which one or both lesions progress. The simplifying assumption that individuals will not present clinically in years in which existing ARM lesions are no different to the previous year was made, as a year was considered to be a long time to go without noticing symptoms.

In the analyses in which screening programme are included, the model applies a probability that individuals respond to a reminder to screen for AMD (screening intervals of 1, 2, 3 and 5 years are tested). For the proportion of individuals who selftest for AMD, probabilities that screening identifies different states of preclinical ARM in the worst affected eyes are specified for early ARM, dry AMD, EF wet AMD, JF wet AMD and SF wet AMD.

Modelling visual acuity

The principal outcome measure for the evaluation of screening for AMD is QALYs, which are estimated by applying utility weights to the years of life of each individual in the model. The utility weights may be a function of the age of an individual and their VA, so the model must capture VA at each year of life. VA is measured in logMAR units in the model, where a logMAR score of 0 is assumed to be equivalent to 6/6 vision using the metric version of the Snellen chart. The following assumptions are made in the model with respect to measuring VA:

- The best possible vision is defined as a logMAR score of 0.
- All eyes with no ARM or early ARM have a logMAR score of 0.
- Upon incidence of dry or wet AMD, eyes are assigned a logMAR score worse than 0.
- VA associated with dry AMD may deteriorate over time even without progression to wet AMD.
- VA associated with EF or JF wet AMD deteriorates upon progression to alternative wet AMD states (by either type or location).
- VA associated with SF wet AMD deteriorates over time based on VA and lesion type at the point of incidence of SF wet AMD.

The health impact of changes in VA levels was estimated by converting the VA level into utility values that reflect the HRQoL of the patient, as depicted in *Figure 3*. Utility values are generally described as values between 0 and 1 that are used to weight the remaining lifetime of patients to reflect their QoL; for example, patients living for 10 years with a constant utility value of 0.8 are said to gain 8 (0.8 × 10) QALYs. Algorithms are available for converting VA scores to utility values (see the section 'Utility values', p. 50 for details of the algorithms).

Modelling treatment

The reference case model describes only treatment options considered to be current practice in the NHS, which limits treatment to PDT for some JF



and SF lesions, and laser photocoagulation for a proportion of EF lesions (see the section 'Calibration outputs', p. 53, for more details). For the treatment of JF and SF lesions, a probability distribution describing the probability that treatment maintains VA at the same level at which treatment is instigated was defined for each relevant disease category. For each iteration of the model, a probability is sampled from each treatment effectiveness distribution. This is implemented by having each eye sample a random number between 0 and 1 to determine whether treatment is effective in maintaining VA. If the sampled value is below the sampled probability that VA is maintained, then treatment is effective and VA in that eye is maintained at that level for the remainder of the patient's life. If treatment is ineffective, VA progresses as if treatment had not been received.

For the treatment of EF wet AMD, a similar process to that described in the previous paragraph is defined whereby VA is either maintained for the remainder of the patient's lifetime, or continues as if treatment had not been received. However, if laser photocoagulation is unsuccessful, an eye may receive subsequent intervention with PDT if the eye progresses to a treatable form of JF or SF wet AMD (as determined by the natural history of the disease).

In additional analyses of the model, where potential interventions for ARM are included, the impact of treatment on disease progression is modelled using two basic approaches. First, some interventions are targeted at preventing progression between the broad ARM states, such as from early ARM to dry or wet AMD, or from dry AMD to wet AMD. In these cases, effective treatments (as defined by a similar process to that described for the main model analysis) are assumed to maintain the eye in the state, and at the same level of VA, at which the intervention was implemented. Other potential interventions are aimed at maintaining or improving VA in eyes with wet AMD, primarily JF or SF. Eyes for which treatment is sampled as being effective are assigned maintained VA or improved VA over the remainder of their lifetime, based on the reported proportions of eyes experiencing either outcome.

Modelling costs

Costs are incurred due to screening, diagnosis and surveillance, treatment and blindness and rapidly deteriorating vision. Further details of the assumptions around the costs included in the model are described in Chapter 7. The costs of screening include only the cost of organising and distributing reminders to individuals aged over 50 years to self-test for AMD. The model attaches a unit cost of screening to each living individual at each age at which a screening reminder is due (e.g. at 3-yearly intervals after the age of 50 years).

Costs associated with diagnosis are attached at the point at which individuals present with ARM, the costs depending on the form of ARM at presentation. If no treatment is offered, the costs of surveillance are applied at each point at which VA deteriorates.

Aggregate (discounted) treatment costs are applied at the point at which treatment is instigated. Eyes treated for EF wet AMD may receive subsequent treatment for JF or SF wet AMD. No further costs are applied to eyes with JF or SF wet AMD until the point at which both eyes have 6/60 vision or worse, at which point the annual costs associated with blindness and rapidly deteriorating vision are applied to the surviving proportion of individuals each year.

Chapter 3 Natural history parameters

This chapter presents the results from the reviews of the literature aimed at informing the model's natural history parameters, including first eye and fellow eye incidence of AMD, and progression from early ARM, from dry AMD and between wet AMD states. Following presentation of the identified data, the relative benefits of the alternative data sources are discussed and the estimation of the relevant model parameters is described.

The first section describes the findings from a review of risk factors for AMD, which was undertaken in order to identify any relevant factors that could be used to define high-risk groups at whom screening could be targeted.

Risk factors

Several studies have been conducted looking at risk factors for the development of AMD. In reviewing these studies, we were interested in identifying risk factors that could be used to (1) identify high-risk groups for targeted screening and (2) identify high-risk activities which needed to be included in the analysis of the model.

Identifying high-risk groups for targeted screening

In adult screening programmes currently in place in the UK, a call and recall service is used to notify and recruit individuals for screening. In this model, routinely identifiable data are used. For example, cervical cancer screening is targeted at women aged 25–50 years and breast cancer screening is targeted at women aged 50 years and over. The population health record includes age and sex, so identification of the target group is easily possible.

Women who have not had penetrative sex are at a much lower risk (some would say no risk at all) of cervical cancer. These women are invited to decline screening based on their sexual history but the invitation is still sent. Similarly, women who have had hysterectomies or breast excision surgery are identified by GPs in the pre-invitation phase.

For AMD screening, we considered a notification system being used to remind people to test their

vision (however this would be done). For this we needed to consider which groups within the population register would be notified.

Several large studies have shown the link between advanced age and onset of AMD. In Smith and colleagues' study,¹¹ the combined analysis of three cohort studies was used to calculate the prevalence of AMD at various ages. This gave a prevalence of 0.2% in the population aged 55–64 years, rising to 13% in those older than 85 years. Most other and subsequent trials have presented data adjusted for age. This same study reported no significant difference by gender.

Identifying high-risk activities for targeted screening

Several high-risk activities were identified through the systematic review, as described in *Table 1*. Recent changes in general practice in the UK have led to the recording of several of these factors as a routine. However, none of the studies quoted relates to a British population. As such, the link between factors identified in the research setting and those identified through routine care within general practice is not clear.

It was decided to model screening as though none of these factors was highlighted on notification. As such, patients with these conditions are modelled as equally likely to screen themselves as others without that factor in the population. This may not be the case, since smoking, raised BMI and hypertension may be suggestive of an individual that would not take up health-promoting activities.^{11–19}

However, with no studies looking at uptake of any of the screening methods within the UK population, we contend that such differential uptake will need to be addressed if and when screening is introduced, rather than at this stage.

Since we did not factor in these activities and risks in the uptake stage of the model, we also did not use these factors to determine likely presentation, success of treatment or mortality. For these we applied figures that are appropriate for individuals at that age and sex – some of whom will have the risk factors, whereas others will not. To take one example: a smoker may be seen to be more likely

Activity or risk	OR or RR	Study reference		
Smoking (current and past versus never smoked)	$OR = 1.7 (95\% \text{ CI } 1.1 \text{ to } 2.6)^{a}$ $OR = 1.3 (95\% \text{ CI } 1.01 \text{ to } 1.69)^{b}$ $OR = 1.91 (95\% \text{ CI } 1.57 \text{ to } 2.33)^{a}$ (Former vs never) OR = 2.8 (95% CI 1.1 to $6.9)^{a}$ (Former vs never) RR = 2.1 (95% CI 1.1 to $3.9)^{c}$ (Current vs never) OR = 3.12 (95% CI 2.10 to $4.64)^{a}$	Seddon and colleagues ¹² Christen and colleagues ¹³ Age-Related Eye Disease Study Research Group ¹⁴ Delcourt and colleagues ¹⁵ Vingerling and colleagues ¹⁶ Smith and colleagues ¹¹		
BMI	(BMI >30) OR = 2.29 (95% CI 1.00 to 5.23) ^c (BMI >31 vs <23.6) OR = 1.43 (95% CI 1.08 to 1.91) ^a	Delcourt and colleagues ¹⁷ Age-Related Eye Disease Study Research Group ¹⁴		
Raised blood pressure	(Treated normal) OR = 2.29 (95% CI 1.12 to 4.69) ^d (Treated high) OR = 3.29 (95% CI 1.24 to 8.79)n ^d (Hypertension) OR = 1.45 (95% CI 1.20 to 1.76) ^a (Hypertension) no association	Klein and colleagues ¹⁸ Age-Related Eye Disease Study Research Group ¹⁴ Klein and colleagues ¹⁹		
Lens opacity	$OR = 1.32 (95\% CI 1.08 \text{ to } 1.60)^{a}$ (Any cataract) OR = 1.30 (95% CI 1.04 to 1.63) ^e (Lens opacity or previous cataract surgery) OR = 1.44 (95% CI 1.09 to 1.89) ^a	Age-Related Eye Disease Study Research Group ¹⁴ Klein and colleagues ²⁰ Chaine and colleagues ²¹		
 BMI, body mass index; CI, confidence interval; OR, odds ratio; RR, relative risk. ^a Multivariate analysis. ^b Age and treatment adjusted. ^c Age and sex adjusted. ^d Adjusted for age, sex, smoking, drinking, vitamin intake. 				

TABLE I Potential risk factors for the development of AMD

^e Generalised estimating equations.

to develop AMD, but less likely to present. If they were equally likely to self-screen, they would benefit from earlier detection of their lesion and consequent benefits of early treatment. However, if they were less likely to take up screening, this benefit would not be achieved. Similarly, if they continued to smoke, the benefit of treatment in arresting AMD in one eye might be cancelled out by the development of AMD in the other eye as a consequence of smoking.

It is clear that there is a complex relationship between risks and outcomes which the available evidence does not allow us to model for the UK population. We took the view that men and women of advanced age should be targeted for screening, but will allow for various lower age limits to be tested in the model. For the purposes of the model, we applied standard distributions of risk of prevalence, presentation and treatment effectiveness based on age but on no other factor.

Incidence

We retrieved and reviewed 39 studies (summary tables are presented in Appendix 2). With respect

to populating the AMD screening model, data were required that described incidence by disease type and by first eye and fellow eye. Single point estimates of disease prevalence were considered of limited applicability as these data do not permit the estimation of age-specific incidence rates, which are also required by the screening model. The following sections describe the studies describing relevant data on the incidence of ARM in the first and fellow eyes, including commentaries on the use of these data to inform age- and stagespecific incidence rates for both eyes.

First eye incidence

Seven studies were identified that presented some form of relevant data to inform the incidence of ARM in a first eye. The data extracted from these studies, and the re-analyses of the reported data, are presented in this section.

The Rotterdam study

The Rotterdam Study was a prospective study of over 10,000 people aged 55 years or older living in a middle-class suburb of Rotterdam, The Netherlands.² Individuals were examined at baseline, 2 years and 6.5 years. The exact data are used to estimate the age-specific incidence of early ARM, and confidence intervals (CIs) are estimated on the basis of person-years at risk. The use of person-years as the denominator informing the CIs will underestimate the true uncertainty, as person-years are estimated from baseline to followup for each participant, whereas the events of interest occurred prior to the follow-up point. However, as the proportion of cases is small, the underestimate is assumed to be insignificant. *Table 2* presents the age-specific annual probabilities of early ARM and the associated 95% CIs.

The estimation of the age-specific probabilities for the incidence of AMD (from no ARM) requires the combination of the presented incidence rates for individuals initially in stages 0 and 1. The AMD incidence rate presented in *Table 3* is estimated as a weighted proportion of the separately presented rates for ARM categories 0 and 1, based on the distribution of patients between categories 0 and 1 at follow-up (the distribution of ARM stages at baseline is not presented). To inform CIs for the AMD incidence rate, the age-specific numbers of cases of AMD developing in persons previously in stages (0 + 1) and (2 + 3)are estimated from the combined incidence rates and the implied person-years at risk. The combined estimates of the number of AMD cases in each age group occurring in persons in categories '0 + 1' and 2 + 3 (see the section 'Visual acuity', p. 32, for a description of progression rates from early ARM) are compared with the observed aggregate numbers of cases of AMD occurring in each age group. The results of this comparison are presented in *Table 4*, which shows a reasonable degree of agreement. As these data only inform the CIs, the effect of the difference between the observed and predicted values is limited.

Table 5 presents the age-specific distribution of AMD between dry and wet AMD, which can be applied to the aggregate AMD rate to estimate the individuals rates for dry and wet AMD.

The Copenhagen study

The Copenhagen study was a population-based cohort study that included 946 residents (age range 60–80 years) of Copenhagen who were examined between 1986 and 1988 and 359 persons (97.3% of survivors) were re-examined between

TABLE 2 Annual incidence of early ARM (per 1000 personyears) in the Rotterdam study 2

Age	Person-	Cases	Mean	95% CI	
(years)	years			Lower	Upper
55–59	2179	3	1.4	0.4	4.3
60–64	6085	32	5.3	3.7	7.4
65–69	6376	69	10.8	8.6	13.7
70–74	5102	97	19.0	15.6	23.2
75–79	3212	102	31.8	26.2	38.6
80+	2159	110	50.9	42.3	61.4

TABLE 3 Annual incidence of AMD (per 1000 person-years) in persons with no ARM

Age	e Person- Cases ^a Mean ^b		Mean ^b	95% CI	
(years)	years	years	Lower	Upper	
60–69 70–79 80+	246 83 4 2 59	3.6 6.5 3.0	0.289 0.78 1.399	0.1 0.3 0.3	0.7 1.6 4.1

^{*a*} The numbers of cases are estimated from the reported mean rate and implied person-years in order to inform the Cls.

^b The mean rates are estimated as weighted proportions of the separately presented rates for ARM categories 0 and 1, based on the distribution of patients in categories 0 and 1 at follow-up.

TABLE 4 Results of validation check for predicted number ofAMD cases occurring from individuals with no ARM or early ARM

Age	Total	Predicted			Predicted	
(years)	observed	Category 0 + I	Category 2 + 3	Total		
60–69	6	3.6	3.6	7.2		
70–79	24	6.5	16.7	23.2		
80+	17	3.0	11.7	14.8		
Total	47	13.1	32.1	45.2		

TABLE 5	Distribution of dry ar	nd wet AMD in	incident cases of
AMD in pe	ersons without early A	RM: the Rotte	rdam study ²

Age	Cases of	of AMD	Proportion Wet AMD				
(years)	Dry	Wet	Mean	95% CI			
				Lower	Upper		
60–64	0	I	I	0.05	1		
65–69	3	2	0.4	0.0527	0.8534		
70–74	3	7	0.7	0.3475	0.9333		
75–79	5	9	0.6429	0.3514	0.8724		
80+	8	9	0.5294	0.2781	0.7702		
Total	19	28	0.5957	0.4427	0.7363		

2000 and 2002.²² It is stated that "Because the correlations are high between eyes, the following analyses for lesions associated with ARM are presented for the right eye only". It is reported that aggregate incidence for late ARM was 14.8% for the right eye and 16.9% for either eye, and 31.5% and 37.8% for early ARM in the right or either eye, respectively.

It was not possible to estimate separate incidence rates for AMD for persons with and without early ARM at baseline, although we know that 254 eyes were at risk of early ARM and 297 were at risk of AMD, so presumably 43 (14.5%) had early ARM at baseline. *Table 6* presents the estimated incidence rates for early ARM and AMD. The Copenhagen study did not report person-years at risk, and CIs were estimated on the basis of persons at risk. Person-years may be estimated as the number of persons in each group multiplied by 14.

The Blue Mountains Eye Study

The Blue Mountains Eye Study (BMES) examined 3654 residents between 1992 and 1994 and re-examined 2335 (75.1% of survivors) between 1997 and 1999.²³ As in the Copenhagen study, incidence rates of AMD are not presented separately by initial stage, that is, the rates of late ARM include people with early ARM at baseline, which is 10% in 70–80-year-olds and 20% in 80+-year-olds. *Table* 7 presents the incidence rates with CIs for early ARM and AMD, which were based on the reported cases and the stated number of persons at risk of each outcome.

The Beaver Dam study

The Beaver Dam (Wisconsin) study was a population-based cohort study that initially examined 4926 persons (age range 43–86 years) between 1988 and 1990;²⁴ 3684 participated in a 5-year follow-up examination and 2764 participated in a 10-year follow-up. As in the Copenhagen and Blue Mountains studies, incidence rates of AMD were not presented separately by initial stage, that is, the rates of AMD include people with early ARM at baseline, which is 17% in 55–64-year-olds and 41% in 75+-year-olds. *Table 8* presents the incidence rates with CIs for early ARM and AMD, which were based on the reported cases and the stated number of persons at risk of each outcome.

Separate incidence rates for dry and wet AMD were presented only for the right eye, as a function of drusen type and drusen size. These data were used to estimate the proportion of wet AMD cases, as presented in *Table 9*. The alternative approaches to estimating the proportion result in a similar estimate of around 60% choroidal neovascularisation (CNV) cases.

The Age-Related Eye Disease Study

The Age-Related Eye Disease Study (AREDS) was a randomised, double-blind, placebo-controlled evaluation of the AREDS formulation of high-dose antioxidants and zinc.²⁵ Eleven clinical centres enrolled 4757 participants (aged 55–80 years) between 1992 and 1998; 1117 patients in category 1 (no ARM) at baseline had few if any drusen.

Age (years)	At risk	14	-year incide	nce	Ar	nual incide	nce
		Mean	95% CI		Mean	95% CI	
			Lower	Upper		Lower	Upper
Early ARM							
60–64	92.5	0.25	0.174	0.349	0.020	0.014	0.030
65–69	92.5	0.33	0.240	0.430	0.028	0.019	0.039
70–74	34.5	0.36	0.218	0.529	0.031	0.017	0.052
75–80	34.5	0.5	0.346	0.668	0.048	0.030	0.076
Dry AMD							
60 <u>–</u> 64	92.5	0.015	0.003	0.059	0.001	0.000	0.004
65–69	92.5	0.015	0.003	0.059	0.001	0.000	0.004
70–74	34.5	0.135	0.049	0.271	0.010	0.004	0.022
75–80	34.5	0.135	0.049	0.271	0.010	0.004	0.022
Wet AMD							
60–64	92.5	0.055	0.024	0.122	0.004	0.002	0.009
65–69	92.5	0.075	0.031	0.136	0.006	0.002	0.010
70–74	34.5	0.135	0.049	0.271	0.010	0.004	0.022
75–80	34.5	0.235	0.127	0.407	0.019	0.010	0.037

TABLE 6 Incidence rates in the right eye for early ARM and AMD in the Copenhagen study²³

Age (years)		5-year incidenc	e		Annual incidend	e
	Mean	2.5th Cl	97.5th Cl	Mean	2.5th Cl	97.5th Cl
Early ARM						
<60	0.032	0.020	0.048	0.006	0.004	0.010
60–69	0.074	0.058	0.093	0.015	0.012	0.019
70–79	0.183	0.150	0.220	0.040	0.032	0.049
80+	0.148	0.081	0.239	0.032	0.017	0.053
Dry AMD						
<60	0.000	0.000	0.004	0.000	0.000	0.001
60–69	0.005	0.002	0.012	0.001	0.000	0.002
70–79	0.017	0.008	0.031	0.003	0.002	0.006
80+	0.036	0.010	0.090	0.007	0.002	0.019
Wet AMD						
<60	0.000	0.000	0.004	0.000	0.000	0.001
60–69	0.005	0.002	0.012	0.001	0.000	0.002
70–79	0.024	0.013	0.041	0.005	0.003	0.008
80+	0.036	0.010	0.090	0.007	0.002	0.019
AMD ^a						
<60	0.000	0.000	0.004	0.000	0.000	0.001
60–69	0.006	0.002	0.014	0.001	0.000	0.003
70–79	0.024	0.013	0.041	0.005	0.003	0.008
80+	0.054	0.020	0.114	0.011	0.004	0.024
^a Late ARM includes g	geographic atroph	y only if it involv	es the fovea.			

TABLE 7 Incidence rates for early ARM and AMD in the Blue Mountains study²³

TABLE 8 Incidence rates for early and late ARM in the Beaver Dam study²⁴

Age (years)	At risk	Cases	10-year incidence			Annual incidence			
			Mean	2.5th Cl	97.5th Cl	Mean	2.5th Cl	97.5th Cl	
Early ARM									
43–54	1136	47	0.041	0.031	0.055	0.004	0.003	0.006	
55–64	856	92	0.107	0.088	0.130	0.011	0.009	0.014	
65–74	654	154	0.236	0.203	0.270	0.027	0.022	0.031	
75+	185	68	0.367	0.298	0.441	0.045	0.035	0.057	
AMD									
43–54	1250	I	0.001	0.000	0.004	0.000	0.000	0.000	
55–64	1032	10	0.010	0.005	0.018	0.001	0.000	0.002	
65–74	899	40	0.044	0.032	0.060	0.004	0.003	0.006	
75+	315	30	0.095	0.065	0.133	0.010	0.007	0.014	

Upon contacting the AREDS team, we obtained unpublished data describing the detailed progression of persons in category 1 at baseline. *Table 10* presents the age-specific 5-year and annual incidence rates, with corresponding CIs, for early and late ARM.

The Melton Mowbray study

The Melton Mowbray study was undertaken between 1980 and 1990 in a small English town whose entire population was served by a single general practice.²⁶ Sparrow and colleagues specify three categories of AMD (none, minor and major). The main problems with these data are that the analysis is presented by eye and the ARM state of the fellow eye is not reported, and that the age distribution of the responders is not presented, as shown in *Table 11*.

Eye I incidence commentary

Estimates of the age-specific incidence of early ARM, dry AMD and wet AMD in the first eye are required for the model. It is currently accepted that the presence of drusen and early ARM is a

Parameter		Dry AMD				Proportion	
	At risk	10-year incidence	Estimated cases	At risk	10-year incidence	Estimated cases	CNV
Max. drusen diameter (μn	1)						
0	381	0	0	355	0	0	
<63	2191	0	0	2158	0	0	
63–125	527	0.003	1.58	517	0.01	5.17	
125–250	229	0.044	10.08	229	0.045	10.31	
>250	59	0.117	6.90	64	0.244	15.62	
Total			18.56			31.09	0.63
Drusen type							
None or hard indistinct	378	0	0	362	0	0	
Hard distinct	2465	0	0	2447	0	0	
Soft distinct	246	0.012	2.95	239	0.02	4.78	
Soft indistinct	241	0.065	15.67	248	0.097	24.06	
Total			18.62			28.84	0.61

TABLE 9 Estimation of the distribution of dry and wet AMD from the Beaver Dam study²⁴

TABLE 10 Incidence rates for early and late ARM for persons with no ARM in the placebo arm of the AREDS trial²⁵

			5-year risks	5			Annual risks	5
	Cases	At risk	Mean	95%	% CI	Mean	95%	% CI
				Lower	Upper		Lower	Upper
Age 55–59								
Categories 2 or 3	0	2	0.000	0.000	0.776	0.000	0.000	0.259
Category 4	0	2	0.000	0.000	0.776	0.000	0.000	0.259
Age 60–64								
Categories 2 or 3	30	151	0.199	0.138	0.271	0.043	0.029	0.061
Category 4	0	151	0.000	0.000	0.020	0.000	0.000	0.004
Age 65-69								
Categories 2 or 3	33	187	0.176	0.125	0.239	0.038	0.026	0.053
Category 4	I	187	0.005	0.0001	0.029	0.001	0.000	0.006
Age 70-74								
Categories 2 or 3	27	118	0.229	0.157	0.315	0.051	0.033	0.073
Category 4	I	118	0.008	0.0002	0.046	0.002	0.000	0.009
Age 75-80								
Categories 2 or 3	9	30	0.300	0.147	0.494	0.069	0.031	0.127
Category 4	0	30	0.000	0.000	0.095	0.000	0.000	0.020

 TABLE II
 Incidence rates for minor and major ARM for persons in the Melton Mowbray study²⁶

	Cases	At risk	7-year	95% Cl Annual		Annual	95% CI	
			incidence	Lower	Upper	Incidence	Lower	Upper
Minor AMD	13	27	0.481	0.269	0.681	0.090	0.044	0.150
Major AMD	3	27	0.111	0.024	0.292	0.017	0.003	0.048

risk factor for AMD, but that persons may progress directly to AMD (dry or wet) without experiencing early ARM [information provided by one of the authors (UC)]. A problem with data describing the incidence of early ARM and AMD (from no ARM) in the first eye is that it is not possible to observe whether incident cases of AMD progressed through early ARM prior to developing AMD. The longer the period between follow-up points, the more likely it is that a proportion of incident AMD cases experienced prior early ARM. The assumption that all observed incident cases of AMD occur directly without prior early ARM provides the upper estimate for the number of direct cases of AMD.

A wide range of estimates of the incidence of ARM is presented in the literature. The Rotterdam study, based on follow-up at 2 years and 6.5 years, reports annual incidence rates of between 0.0014 (ages 55–59 years) and 0.051 (over 80 years) for early ARM. Annual incidence rates for AMD are between 0.00029 (60–69 years) and 0.0014 (over 80 years).

The 14-year period between baseline and followup in the Copenhagen study¹¹ is likely to lead to the greatest overestimate of the number of cases of directly incident AMD (and underestimate of incident early ARM). It is also not possible to distinguish the cases of AMD that occurred in patients with no ARM, and early ARM, at baseline. One can only estimate that around 14.5% of the population had early ARM at baseline. Agespecific incidence rates are presented for the right eye, which is not helpful as these incidence rates include cases in which the left eye developed ARM previously. In summary, the only useful data presented by the Copenhagen study are the aggregate annual incidence of early ARM and AMD in either eye, which are estimated to be 0.033 and 0.013, respectively. The combined annual incidence of early ARM in the Copenhagen study (0.033) is higher than the reported rate in the Rotterdam study in all age categories other than the oldest (over 80 years). The combined annual incidence of AMD is higher than the observed incidence in all age categories of the Rotterdam study.

The Australian BMES²³ incorporated one followup point 5 years after the baseline observation, and reported similar levels of follow-up to the Rotterdam study (75% of survivors attended follow-up). The estimated annual incidence rates of both early ARM and AMD are slightly higher in the younger age groups (e.g. 0.006 for early ARM in the under-60 years category compared with 0.0014 in the 55–59 years category of the Rotterdam study), but lower in the oldest age group (0.032 for early ARM compared with 0.051 in the over-80 years category). Part of the reason for the generally higher incidence rates for AMD is that, as in the Copenhagen study, the incidence rates for AMD include persons with early ARM at baseline, which is estimated to be around 5% in total (although up to 20% in the oldest age group).

The Beaver Dam study²⁴ included two follow-up points, at 5 and 10 years, and also reports agespecific incidence rates for early ARM that are higher than the Rotterdam study. The AMD incidence rates include cases in which early ARM was present at baseline (estimated to be around 19%), which, combined with the longer intervals between follow-up points, provide methodological reasons for the higher reported rates of AMD. However, van Leeuwen and colleagues² believe that the observed differences may be due to differences in the underlying risk of the respective populations.

The US-based prospective cohort study, AREDS, followed up participants at annual visits beginning at year 2, and had a median follow-up of 6.3 years.²⁷ The reported incidence rates of early ARM are higher than the above reported studies, although the CIs from AREDS are wide and overlap in most cases (although not with the Rotterdam study). Compared with the Rotterdam study, the definition of early ARM is broader; for example, it includes eyes with pigment abnormalities alone that are not included in the Rotterdam definition of early ARM.

Only two incident cases of AMD were observed in AREDS. Combined with AREDS having the shortest intervals between follow-up points and higher incidence rates of early ARM, this implies that a significant proportion of incident cases of AMD observed in the previous studies experienced early ARM prior to AMD.

The only UK study informing incidence rates is the Melton Mowbray study,²⁶ which reports 7-year incidence rates for minor and major AMD. Minor AMD includes all forms of drusen and pigment abnormalities, other than indistinct hard drusen. Major AMD includes more severe representations of drusen as well as geographic atrophy (GA) and CNV. In addition to the variable definitions, the sample is small and no age-specific data are presented. The problems noted with the Melton Mowbray and other studies mean that the age-specific incidence rates presented by the Rotterdam study appear of most relevance to a UK study, given the similar environmental and demographic characteristics. However, the Rotterdam study reports the lowest incidence rates of all the identified studies, which is unlikely to be explained fully by differences in underlying risk. Buch and colleagues²² recognised the higher observed rate in the Copenhagen study, and discussed potential explanatory factors. A key factor may be differences in participation rates as the Copenhagen study achieved a 97.3% follow-up rate, with 38.5% of incident late ARM cases being found through home examinations. The Rotterdam study, for example, achieved only 82% (range 57-87% across age groups) and 71% (range 29-79% across age groups) follow-up rates of living patients at the 2- and 6.5-year follow-up points, respectively. Differences in study design, sample variability, geographic location and methodology are also cited as possible sources of disparity among studies.

Buch and colleagues²² also stated that incidence does not remain constant over time, especially for AMD, such that the longer follow-up in the Copenhagen study is likely to result in higher estimated annual rates. This issue is addressed by the estimation of age-specific incidence rates, which allows the model to increase the probability of incidence as patients with early ARM reach sequential age thresholds.

More generally, the competing risk of death may affect the accuracy of the reported incidence rates. Persons dying between follow-up points may be more likely to develop ARM prior to death, but such cases are not observed.

On the basis of the above comparison, the input parameter estimates for the incidence of first eye early ARM and AMD were based on the Rotterdam data, although adjustments were initially made to reflect the lower follow-up rates in this study. The incidence rates in the age categories 60–69, 70–79 and 80+ years were adjusted upwards by factors of 1.2, 1.4 and 1.6, respectively. The relative adjustments were based on the reported follow-up rates in the different age categories (the older age groups had poorer follow-up rates) and the aim of increasing the weighted mean incidence rate by 40% (slightly higher than the 38.5% of incident late ARM cases that were found through home examinations in the Copenhagen study as only 16% of follow-up

was undertaken at home, i.e. some additional detection may be due to increased follow-up of persons attending the eye clinic).

To represent the uncertainty in the model's first eye incidence parameters, beta probability distributions were specified for each age category for early ARM incidence, and AMD incidence, based on the estimated number of cases and person-years at risk. The only exception is the incidence of AMD in the age category 55-59 years, in which no cases were observed in approximately 2179 person-years at risk. Based on the mean value of zero and the estimated upper 95% CI (0.0014), a standard deviation of 0.0007 is fitted to a normal distribution with mean zero so that the upper 95% CI equals 0.0014. The model can then sample a probability of progression such that if a value between 0 and 0.5 is sampled, the probability of incidence is zero, whereas values above 0.5 result in a positive probability of incidence.

Uncertainty around the likelihood of a case of AMD being dry or wet is also handled by beta distributions informed by the reported numbers of total cases of dry and wet AMD in each age category (i.e. including cases of AMD that progressed from early ARM). The only exception is for the age category 60–64 years, in which one case of wet AMD and no cases of dry AMD were observed. Lower and upper 95% CIs are estimated (0.05 and 1, respectively), which, combined with the mean probability that the AMD is wet, were used to define a triangular distribution.

Fellow eye incidence

Nine studies presenting data describing the incidence of ARM in the fellow eye were identified. The data extracted from these studies, and the re-analyses of these data, are presented in this section.

The Rotterdam study

The Rotterdam study describes outcomes for 56 patients with uniocular AMD at baseline, although only 25 participated at the first follow-up and seven at the second follow-up.² *Table 12* presents the estimated annual incidence rates and 95% CIs for the possible transitions, based on the number of cases in each category and the presented person-years at risk data.

The Blue Mountains Eye Study

The only data relating to second eye incidence identified in reports of the BMES state that the 5-year incidence rate for the development of AMD

Transition	Annual incidence	95% CI	
		Lower	Upper
Ist eye dry AMD, fellow eye dry AMD Ist eye dry AMD, fellow eye wet AMD Ist eye wet AMD, fellow eye dry AMD Ist eye wet AMD, fellow eye wet AMD	0.079 0.026 0.000 0.088	0.017 0.001 0.000 0.029	0.214 0.138 0.051 0.193

TABLE 12 Annual incidence rates of fellow eye AMD by initial and subsequent AMD: the Rotterdam study²

(dry AMD defined as involving the fovea) in the second eye of participants with AMD in one eye at baseline was 28.6% (six of 21).²³

The AREDS study

The AREDS study report no. 19^{27} presents data describing the incidence of AMD (dry AMD involving the centre of the macula) in the second eye of patients with unilateral AMD over a mean follow-up time of 6.3 years. Data from patients in all four treatment groups (placebo, antioxidants, zinc, antioxidants + zinc) are combined, which requires consideration of the fact that zinc and antioxidants + zinc were shown to reduce significantly the incidence of wet AMD in the fellow eye of patients with unilateral AMD.

Data describing the number at risk and the number of cases of dry AMD and wet AMD are presented. In addition, odds ratios (ORs) for the development of wet or dry AMD in patients aged 65–69 versus <65 years and age >70 versus <65 years are presented. As the paper does not present the age-specific event rates, the number of cases of dry and wet AMD occurring in each age group are estimated by solving equations for the presented ORs simultaneously so that the total number of cases equals the observed cases. The outputs from these estimates are presented in *Table 13*. **TABLE 13** Annual incidence rates of second eye AMD in patients with unilateral AMD: AREDS²⁷

Age (year) ^a	Mean	95% CI						
		Lower	Upper					
Wet AMD								
<65	0.043	0.030	0.060					
65–69	0.065	0.051	0.081					
>70	0.074	0.064	0.085					
Central dry AMD								
<65	0.012	0.006	0.023					
65–69	0.015	0.010	0.024					
>70	0.020	0.015	0.026					
^a Age-specific rates are estimated by solving equations for								

the presented ORs simultaneously so that the total number of cases equals the observed cases.

The Beaver Dam study

The Beaver Dam study²⁴ reports the incidence of either dry or wet AMD in the fellow eyes of eyes with dry or wet AMD at baseline. Dry AMD is defined as being in any subfield. *Table 14* presents the estimated annual incidence rates and their corresponding 95% CIs. Eight cases of dry AMD were observed, although two cases that were observed at the 5-year follow-up point developed wet AMD by the 10-year point. These persons are counted twice as the sequential observation of dry and then wet AMD represents the true incidence.

TABLE 14 Incidence of dry and wet AMD in eyes with uniocular late ARM at baseline: the Beaver Dam $study^{24}$

	At risk	Cases	10-year	9 5%	6 CI	Annual	95% CI	
			incidence	Lower	Upper	incidence	Lower	Upper
2nd eye dry AMD								
Ist eye dry AMD	11	5	0.455	0.168	0.766	0.059	0.018	0.135
Ist eye wet AMD	15	3	0.200	0.043	0.481	0.022	0.004	0.063
2nd eye wet AMD								
Ist eye dry AMD	11	I	0.091	0.002	0.413	0.009	0.000	0.052
Ist eye wet AMD	15	3	0.200	0.043	0.481	0.022	0.004	0.063

© Queen's Printer and Controller of HMSO 2008. All rights reserved.

The CNV Prevention Trial

The CNV Prevention Trial (CNVPT)²⁸ was a randomised trial that compared laser photocoagulation with observation. The eligible population included eyes without AMD and VA >20/40, but with >10 large drusen (>63 μ m diameter), whose fellow eyes had evidence of CNV or serious RPE detachment. Laser treatment was found to be ineffective and potentially harmful, so data from the observed group were analysed for use in the natural history model. Unlike the population prevalence studies, patients in the CNVPT presented clinically with the CNV in the first eye.

Data describing patients followed up at different time points and graphs of the proportion of patients developing dry AMD (within 1500 µm of the centre of the fovea) or wet AMD over time were used to estimate the respective annual incidence rates by year of follow-up. To inform CIs, the numbers of cases of AMD at each followup point were estimated by multiplying the overall incidence rate by the number of patients at risk. Table 15 presents the estimated annual incidence rates and associated CIs. The most striking result is the difference in the estimated annual incidence rate based on the 1-year and the 2-, 3- or 4-year follow-up data, which shows that the incidence rate increases sharply after the first year, although the CIs do overlap.

The Macular Photocoagulation Study

The Macular Photocoagulation Study (MPS) group have published three papers that describe different data on the incidence of second eye wet AMD.^{29–31} A 1996 paper presents 5-year follow-up data on fellow eyes of patients from two trials covering patients with unilateral EF and JF wet AMD.³⁰ Fellow eyes did not have wet AMD at baseline but could have had early ARM or dry AMD.

Separate results are presented for patients aged under 50 years, JF first eye lesions, EF first eye lesions, the laser treatment group and the observation group. The rates in the treatment and observation group are not significantly different (p = 0.28), although the mean rate of second eye involvement is lower in the observation group. The results are not presented by location and treatment group. *Table 16* presents the estimated annual incidence rates and associated CIs.

A 1997 paper by the MPS group published 5-year follow-up data for fellow eyes of patients from three clinical trials covering new JF wet AMD, new SF wet AMD and recurrent SF CNV secondary to AMD.³¹ Again, fellow eyes did not have wet AMD at baseline but could have had early ARM or dry AMD. It is not explicit, but it appears that the reported JF trial is the same as that reported in the previous paper.³⁰ However, this study presents a range of useful data including incidence rates of wet AMD in all fellow eyes over time,³¹ which allows an assessment of whether incidence rates are constant over time. Table 17 presents these data, which show that the estimated annual incidence is almost identical based on the data at each of the 5 years of follow-up.

The 1997 paper also presents data describing the 5-year incidence of fellow eye wet AMD by the drusen characteristics of the central macula in the

Follow-up (years)	At risk	Cases	Cumulative	95% CI		Annual	95% CI	
			Incidence	Lower	Upper	Incldence	Lower	Upper
Wet AMD								
1	58	2.999	0.052	0.011	0.119	0.052	0.011	0.119
2	47	12.220 ^a	0.260	0.156	0.404	0.140	0.081	0.228
3	38	11. 780 ª	0.310	0.139	0.380	0.116	0.049	0.147
4	37	14.001	0.378	0.225	0.552	0.112	0.062	0.182
Dry AMD								
	58	3.480	0.060	0.019	0.144	0.060	0.019	0.144
2	47	3.760	0.080	0.024	0.175	0.041	0.012	0.092
3	38	6.460	0.170	0.077	0.313	0.060	0.026	0.117
4	37	7.770	0.210	0.098	0.352	0.057	0.026	0.103

TABLE 15 Incidence of dry and wet AMD in fellow eyes with early ARM of first eyes with CNV: the CNVPT²⁸

^a The reduction in the number of cases between follow-up at 2 and 3 years is an anomaly. The data are not adjusted as the key parameter is the baseline incidence rate, which is not affected by the anomaly.

	Cases	At risk	5-year	95%	o CI	p-Value	Annual	95% CI	
			incidence	Lower	Upper		incidence	Lower	Upper
All	35	394	0.089	0.063	0.121		0.018	0.013	0.026
>50 years old	6	115	0.052	0.019	0.110	0.28	0.011	0.004	0.023
lst eye EF	20	181	0.110	0.069	0.166		0.023	0.014	0.036
Ist eye JF	15	213	0.070	0.040	0.114	0.18	0.014	0.008	0.024
Laser treatment	21	202	0.104	0.066	0.155		0.022	0.013	0.033
Observation	14	192	0.073	0.040	0.119	0.28	0.015	0.008	0.025

TABLE 16 Incidence of wet AMD in the fellow eye of patients with first eye wet AMD

TABLE 17 Incidence of wet AMD in the fellow eye of patients with unilateral JF or SF wet AMD: the MPS³¹

Follow-up (years)	Cumulative incidence	Annual incidence		
I	0.1	0.100		
2	0.19	0.100		
3	0.28	0.104		
4	0.36	0.106		
5	0.42	0.103		

fellow eye.³¹ These results are presented in *Table 18* and indicate that fellow eyes with early ARM (defined as drusen size >63 μ m) have a significantly increased rate of developing wet AMD compared with fellow eyes with no ARM or small drusen. The data also indicate that eyes with dry

AMD have similar wet AMD incidence rates to those with early ARM.

The third form of data describes age-specific incidence rates of wet AMD in fellow eyes, although these data are not presented by drusen characteristic. The data presented in *Table 19* show an upward trend in the rate of fellow eye wet AMD with increasing age.

The third study by the MPS group presents 5-year follow-up of fellow eyes of patients from a trial of argon laser photocoagulation for EF wet AMD.²⁹ The dates and patient numbers of the trials of argon laser photocoagulation for eyes with EF AMD reported in this study and in the above-mentioned study³⁰ are similar but not identical, implying that they are reporting results from

	At risk	Cases	Cases	5-year	95 %	6 CI	Annual	95% CI	
			incidence	Lower	Upper	incidence	Lower	Upper	
Largest druse (µm)									
None	23	5	0.230	0.075	0.437	0.051	0.015	0.109	
<63	173	48	0.280	0.212	0.351	0.064	0.047	0.083	
64–125	158	90	0.570	0.489	0.648	0.155	0.126	0.188	
>125	85	45	0.530	0.418	0.639	0.140	0.103	0.184	
Dry AMD									
None	464	204	0.440	0.394	0.486	0.109	0.095	0.125	
Dry AMD	20	10	0.490	0.272	0.728	0.126	0.062	0.229	

TABLE 18 Incidence of wet AMD in the fellow eye of patients with unilateral JF or SF wet AMD by drusen characteristics: the MPS³¹

TABLE 19 Incidence of wet AMD in the fellow eye of patients with unilateral JF or SF wet AMD by age: the MPS³¹

Age (years)	At risk	Cases	5-year	95% CI		Annual	95 %	% CI
			incidence	Lower	Upper	Incidence	Lower	Upper
50–69	237	83	0.350	0.290	0.415	0.083	0.066	0.102
70–74	168	76	0.450	0.376	0.531	0.113	0.090	0.140
>75	265	133	0.500	0.440	0.564	0.129	0.110	0.153

 $\ensuremath{\mathbb{C}}$ Queen's Printer and Controller of HMSO 2008. All rights reserved.

separate trials. Fellow eyes did not have wet AMD at baseline but could have had early ARM or dry AMD.

Table 20 presents data describing the incidence of dry or wet AMD in the fellow eye of patients with unilateral EF wet AMD by drusen characteristics, which show that the incidence of dry AMD is low. The data support the noted similarity of wet AMD progression in fellow eyes with early ARM or dry AMD (as observed in fellow eyes of eyes with JF or SF AMD). The reported zero incidence of wet AMD in fellow eyes with only small drusen or no ARM is surprising. Although the numbers are relatively small, the estimated upper 95% CI for the annual incidence rate is only 2.5%.

Baltimore studies

In the Baltimore studies, Bressler and colleagues³² present data on fellow eyes of patients with first eyes with poorly defined angiographic leakage, which was assumed to represent wet AMD. All fellow eyes are defined as having drusen, although further details are not provided. *Table 21* presents the incidence rates, which are broadly in

line with the data presented by the MPS group^{29,30} (considering that a proportion of patients may have had small drusen in the current study).

In another Baltimore study, Sunness and colleagues³³ reports the incidence of second eye wet AMD in 46 patients with first eye wet AMD, and second eye dry AMD at baseline. The criteria for dry AMD were at least one area of discrete retinal pigment epithelial atrophy of at least 500 mm in diameter within one disc diameter of the fovea. *Table 22* presents the estimated annual incidence rates and CIs at years 1–4, which show very similar annual rates of progression across the years.

Munster study

Pauleikhoff and colleagues³⁴ in the Munster study present data on fellow eyes of patients with unilateral wet AMD, in which fellow eyes were required to have VA of 20/30 or better. It is implied that all fellow eyes had drusen. *Table 23* presents the estimated annual incidence rates by type of wet AMD in the first eye (classic and

95% CI At risk Cases 5-year Annual 95% CI incidence incidence Lower Lower Upper Upper Dry AMD I + large drusen or focal 53 2 0.038 0.005 0.130 0.008 0.001 0.027 hyperpigmentation 0 0.025 Small drusen 24 0 0 0.117 0 0 Other significant macular 11 0 ٥ 0 0.238 0 0 0.053 disease None of the above 2 0 0 0 0.776 0 0 0.259 Wet AMD Dry AMD 5 0.455 0.168 0.766 0.114 0.036 0.252 11 I + large drusen or focal 53 20 0.377 0.248 0.521 0.090 0.055 0.137 hyperpigmentation 0 0 0.025 Small drusen 24 0 0 0.117 0 Other significant macular н 2 0.182 0.023 0.518 0.039 0.005 0.136 disease None of the above 2 0 0 0 0.776 0 0 0.259

TABLE 20 Incidence of wet or dry AMD in the fellow eye of patients with unilateral EF wet AMD by drusen characteristics: the MPS³¹

TABLE 21 Incidence of wet AMD in the fellow eye of eyes with wet AMD: the Baltimore study³²

	At risk	Cases	28-month	95% CI		Annual	95 %	6 CI
			Incidence	Lower	Upper	Incidence	Lower	Upper
Drusen	31	6	0.194	0.075	0.375	0.088	0.033	0.182

Follow-up (years)	Cumulative	95 %	% CI	Annual	95% CI	
	incidence	Lower	Upper	Incidence	Lower	Upper
1	0.08	0	0.17	0.080	0.000	0.170
2	0.18	0.04	0.3	0.094	0.020	0.163
3	0.22	0.06	0.36	0.079	0.020	0.138
4	0.34	0.12	0.51	0.099	0.031	0.163

TABLE 22 Progression from dry to wet AMD in fellow eyes of first eyes with wet AMD: the Baltimore study³³

TABLE 23 Incidence of CNV in the fellow eye of eyes with CNV by first eye CNV type: the Munster study³⁴

Year of follow-up	At risk	Cases	Cumulative	95% CI		Annual	95%	6 CI
			Incidence	Lower	Upper	incidence	Lower	Upper
All patients								
1	187	6	0.032	0.012	0.069	0.032	0.012	0.069
2	178	27	0.152	0.102	0.213	0.079	0.053	0.113
3	142	37	0.261	0.191	0.341	0.096	0.068	0.130
4	123	50	0.407	0.319	0.499	0.122	0.092	0.159
Ist eye classic wet AMI	D							
1	130	2	0.015	0.002	0.055	0.015	0.002	0.055
2	125	13	0.104	0.057	0.171	0.053	0.029	0.090
3	107	20	0.187	0.118	0.274	0.067	0.041	0.101
4	94	30	0.319	0.227	0.423	0.092	0.062	0.129
Ist eye occult wet AME)							
1	57	4	0.070	0.020	0.170	0.070	0.020	0.170
2	53	14	0.264	0.153	0.403	0.142	0.079	0.228
3	35	17	0.486	0.314	0.660	0.199	0.118	0.302
4	29	20	0.690	0.492	0.847	0.254	0.156	0.375

occult), which indicates a higher incidence of wet AMD in the fellow eyes of first eyes with occult wet AMD. The CIs do overlap, but this may be due to the sample size, which is small.

These data also indicate a non-constant rate of second eye involvement, showing a steadily increasing annual incidence rate over the 4-year follow-up period. This result is at odds with the observed time sequence in the CNVPT,²⁸ which showed a decreased incidence rate in the first year followed by seemingly constant rates in subsequent years. It is also contrary to the results presented by the MPS group, who reported a constant incidence rate for fellow eye wet AMD over time.

Danish study

In a Danish study, Baun and colleagues³⁵ report data on 45 patients with wet AMD in one eye at baseline; results are presented by the state of the fellow eye at baseline. The small numbers produce wide CIs, but the data presented in *Table 24* indicate low rates of wet AMD in fellow eyes with no ARM. That the highest rate is in the small drusen group is surprising, although, as noted, the CIs are wide.

Fellow eye incidence commentary

Separate estimates of ARM incidence rates in the second eye are required for different eye 1 ARM states, that is, eye 2 incidence given early ARM, GA or CNV, in the first eye. The following sections describe the respectively defined input parameter values.

Fellow eye incidence given early ARM in the first eye

No data were identified describing the incidence of ARM in the fellow eye of eyes with early ARM. Though it may tend to overestimate incidence, similar fellow eye incidence rates to those estimated for fellow eyes of eyes with dry AMD were assumed.

Fellow eye incidence given dry AMD in the first eye

The Rotterdam study presents aggregate data based on 38 person-years at risk, which can be

Fellow eyes at baseline	At Cases		4-year	95% CI		Annual	95% CI	
	risk		Incidence	Lower	Upper	incidence	Lower	Upper
No changes	5	0	0	0	0.451	0	0	0.139
Cluster(s) of small drusen	8	5	0.625	0.245	0.915	0.217	0.068	0.460
Medium-sized drusen	17	4	0.235	0.068	0.499	0.065	0.017	0.159
Large confluent drusen	10	3	0.300	0.067	0.653	0.085	0.017	0.232
Dry AMD	5	2	0.400	0.053	0.853	0.120	0.013	0.381
Total	45	14	0.311	0.182	0.467	0.089	0.049	0.145

TABLE 24 Incidence of wet AMD in the fellow eye of eyes with wet AMD by fellow eye characteristics: the Danish hospital study 35

used to estimate annual incidence rates of dry AMD (0.079) and wet AMD (0.026) in the second eye given dry AMD in the first eye. The Beaver Dam study allows the estimation of annual incidence rates of 0.059 and 0.009 for second eye dry AMD and wet AMD, respectively, given dry AMD in the first eye.

These rates are similar, and as first eye incidence was informed by the Rotterdam study, which is also deemed to have the most similar demographics to a UK population, the Rotterdam study informs the aggregate incidence rate for fellow eyes given dry AMD in the first eye.

Age-specific incidence rates for central dry AMD and wet AMD in patients with uniocular AMD were estimated using data presented by AREDS.²⁷ Only 74 of 788 (9.4%) patients had first eye dry AMD, although these data remain the most relevant agespecific data, and the ratios of incidence rates in the different age groups were used to disaggregate the aggregate Rotterdam study incidence rates.

To represent the uncertainty in the parameter estimates, the above parameters were specified as beta distributions; the sampled values from each were combined to estimate age-specific fellow eye incidence rates.

Fellow eye incidence given wet AMD in the first eye

Three studies presented data describing fellow eye incidence given first eye wet AMD and no ARM (where no ARM includes small drusen) at baseline in the fellow eye, two of which were published by the MPS group. An aggregate annual incidence rate of 0.061 for the development of fellow eye wet AMD in second eyes with no or small drusen is estimated.¹⁹ The rates were similar in the no drusen (0.051) and the <63- μ m drusen (0.064) categories. Age-specific rates of fellow eye wet AMD are also presented, although not by drusen characteristics.³⁰

Data presented by the Beaver Dam study²⁴ allow the estimation of annual incidence rates of 0.022 for both fellow eye dry and wet AMD given wet AMD in the first eye. No cases of dry or wet AMD were observed in the fellow eye given no or small drusen at baseline, although an upper 95% CI of 0.11 is estimated from the data presented for both outcomes. Baun and colleagues³⁵ observed no cases of CNV in fellow eyes with no changes (based on five observations), but five out of eight cases in eyes with small clusters of drusen. The CIs do overlap.

Data from the Rotterdam study estimate annual rates of 8.8% and 0% for fellow eye wet and dry AMD, respectively, given first eye wet AMD.² These data inform beta probability distributions describing the incidence of fellow eye dry or wet AMD, given first eye wet AMD. As above, the agespecific incidence rates of central dry AMD and wet AMD in patients with uniocular advanced AMD presented by AREDS are used to define agespecific incidence rates.

Natural history

The natural history of ARM refers to the progression of disease between different stages of the disease.

Progression from early ARM and dry AMD

Six studies are identified that present data describing the progression of early ARM or dry AMD to wet AMD. The data extracted from these studies and the re-analyses of these data are presented in this section.

The Rotterdam study presents age-specific 5-year incidence of dry and wet AMD in eyes with early ARM at baseline.² The estimation of the age-specific probabilities of AMD required some reanalysis of the presented data as separate mean
Age (years)	Person-years	Cases ^a	Mean ^b	95% CI	
				Lower	Upper
60–69	359	3.6	0.0101	0.003	0.0242
70–79	724	16.7	0.0230	0.0137	0.0356
80+	335	11.7	0.0350	0.0186	0.058

TABLE 25 Annual incidence of AMD in persons with early ARM: the Rotterdam study²

^{*a*} The numbers of cases are estimated from the reported mean rate and implied person-years in order to inform the Cls. ^{*b*} The mean rates are estimated as weighted proportions of the separately presented rates for ARM categories 2 and 3, based on the distribution of patients in categories 2 and 3 at follow-up.

TABLE 26 Incidence of AMD in persons with alternative forms of early AMD at baseline: the Copenhagen study²²

	At risk	Cases ^a	14-year incidence		Annual incide		dence	
			Mean	95%	6 CI	Mean	95 %	6 CI
				Lower	Upper		Lower	Upper
Early ARM								
Wet AMD	37	18	0.486	0.319	0.631	0.046	0.027	0.069
Dry AMD	37	12	0.324	0.180	0.470	0.028	0.014	0.044
Hard distinct drusen								
Wet AMD	105	3	0.029	0.0059	0.0812	0.002	0.000	0.006
Dry AMD	105	0	0	0	0.0281	0.000	0.000	0.002
Soft distinct drusen								
Wet AMD	47	9	0.192	0.0915	0.3326	0.015	0.007	0.028
Dry AMD	47	4	0.083	0.0237	0.2038	0.006	0.002	0.016
Soft indistinct drusen								
Wet AMD	32	17	0.531	0.3474	0.7091	0.053	0.030	0.084
Dry AMD	32	11	0.344	0.1857	0.5319	0.030	0.015	0.053

^a Early AMD is defined as the absence of signs of advanced AMD and the presence of (1) soft, indistinct or reticular drusen or (2) hard, distinct or soft, distinct drusen with pigmentary abnormalities (RPE depigmentation or increased retinal pigment).

rates are presented for stage 2 and stage 3 ARM (the component parts of early ARM). A combined AMD incidence rate is estimated as a weighted proportion of the separately presented rates for ARM categories 2 and 3, based on the mean distribution of patients between categories 2 and 3 at follow-up (the distribution of ARM stages at baseline is not presented).

CIs for AMD incidence in persons with early ARM were informed by estimating the age-specific numbers of cases of AMD developing in persons previously in ARM stages '0 + 1' and '2 + 3' from the combined incidence rates and the implied person-years at risk. The same age-specific distribution of dry and wet AMD, as described for incident cases of AMD in persons without early ARM (see the section 'First eye incidence', p. 12) is applied to the aggregate AMD rate to estimate the individual rates for dry and wet AMD. The resulting estimates are presented in *Table 25*.

The Copenhagen study²² describes 14-year incidence of dry AMD and wet AMD by drusen category and presence of early ARM at baseline, although not by age. *Table 26* presents 14-year and annual incidence rates of dry and wet AMD for different subgroups of early ARM at baseline.

AREDS report no. 19^{27} describes AMD incidence in patients with "early or intermediate AMD at baseline", which is defined as including AMD categories 2 and 3a. Category 2 participants had mild or borderline age-related macular lesions [multiple small drusen, non-extensive (<20), intermediate drusen (63–124 µm in diameter), pigment abnormalities or any combination of these] in their most advanced eye and VA of 20/32 or better in both eyes. In category 3a, both eyes had at least one large druse ($\geq 125 \ \mu m$ in diameter), extensive (as measured by drusen area) intermediate drusen, dry AMD that did not involve the centre of the macula or any combination of these. The mean follow-up time was 6.3 years and the data presented combine patients in all four treatment groups (placebo, antioxidants, zinc, antioxidants + zinc), although the only significant effects of treatment were for zinc and antioxidants + zinc for the prevention of CNV in patients with unilateral AMD.

Data describing the number at risk and the number of cases of dry AMD (involving the centre of the macula) and wet AMD are presented. In addition, ORs for the development of dry or wet AMD for a range of variables compared with the base case are presented, including age 65–69 versus <65 years and age >70 versus <65 years. As the paper does not present the age-specific event rates, the number of cases of dry and wet AMD occurring in each age group are estimated by solving equations for the presented ORs simultaneously so that the total number of cases equals the observed cases. The outputs from these estimates are presented in *Table 27*.

Data from the Beaver Dam study²⁴ describe the 10-year incidence of dry AMD (central and noncentral) and wet AMD in the right eye of patients with different categories of early ARM or no ARM in their right eye at baseline. The estimated annual incidence rates and CIs for dry and wet AMD are presented in *Tables 28* and *29*, respectively. **TABLE 27** Annual incidence rates of central dry AMD and wetAMD from early and intermediate AMD: AREDS²⁷

	Mean ^a	95%	6 CI
		Lower	Upper
Wet AMD			
<65	0.010	0.007	0.013
65–69	0.016	0.013	0.019
>70	0.022	0.018	0.026
Dry AMD			
<65	0.005	0.003	0.008
65–69	0.009	0.007	0.012
>70	0.011	0.009	0.014

^a Age-specific rates are estimated by solving equations for the presented ORs simultaneously so that the total number of cases equals the observed cases.

Wang and colleagues³⁶ describe the 5-year risk of AMD in eyes with macular drusen and hyperpigmentation from the BMES, using a similar categorisation to that used by AREDS. However, the study presents separate results for right and left eyes, does not report any agespecific progression rates and does not separate out eyes with and without fellow eyes with AMD. These data are of limited relevance given other data sources.

Holz and colleagues³⁷ report data from a prospective study undertaken in the UK at Moorfields Eye Hospital on 126 patients with bilateral macular drusen (BMD). Patients

	At	Cases	10-year	95 %	95% CI Annual 95%		6 CI			
	risk		incidence	Lower	Upper	incidence	Lower	Upper		
Maximum drusen diameter (μ m)										
0	381	0	0	0	0.008	0.000	0.000	0.001		
<63	2191	0	0	0	0.001	0.000	0.000	0.000		
63–125	527	2	0.003	0.001	0.014	0.000	0.000	0.001		
125–250	229	10	0.044	0.021	0.079	0.004	0.002	0.008		
>250	59	7	0.117	0.049	0.229	0.013	0.005	0.026		
>125	288	17	0.059	0.035	0.093	0.006	0.004	0.010		
>63	815	19	0.023	0.014	0.036	0.002	0.001	0.004		
Drusen type										
None or hard indistinct	378	0	0	0	0.008	0.000	0.000	0.001		
Hard distinct	2465	0	0	0	0.001	0.000	0.000	0.000		
Soft distinct	246	3	0.012	0.003	0.035	0.001	0.000	0.004		
Soft indistinct	241	16	0.065	0.038	0.106	0.007	0.004	0.011		
Soft distinct or indistinct	487	19	0.039	0.024	0.060	0.004	0.002	0.006		
Pigmentary irregularities										
Any	224	18	0.082	0.0483	0.124	0.009	0.005	0.013		

TABLE 28 Incidence of dry AMD in the right eye of patients by maximum drusen diameter and drusen type: the Beaver Dam study²⁴

	At	Cases	10-year	95%	% CI	Annual	95%	6 CI
	risk		incidence	Lower	Upper	incidence	Lower	Upper
Maximum drusen diameter	(μm)							
0	355	0	0	0	0.008	0.000	0.000	0.001
<63	2158	0	0	0	0.001	0.000	0.000	0.000
63–125	517	5	0.010	0.003	0.022	0.001	0.000	0.002
125–250	229	10	0.045	0.021	0.079	0.005	0.002	0.008
>250	64	16	0.244	0.150	0.374	0.028	0.016	0.046
>125	293	26	0.089	0.059	0.127	0.009	0.006	0.014
>63	810	31	0.038	0.026	0.054	0.004	0.003	0.006
Drusen type								
None or hard indistinct	362	0	0	0	0.008	0.000	0.000	0.001
Hard distinct	2447	0	0	0	0.001	0.000	0.000	0.000
Soft distinct	239	5	0.020	0.007	0.048	0.002	0.001	0.005
Soft indistinct	248	24	0.097	0.063	0.141	0.010	0.006	0.015
Soft distinct or indistinct	487	29	0.060	0.040	0.084	0.006	0.004	0.009
Pigmentary irregularities								
Any	236	15	0.065	0.036	0.1027	0.007	0.004	0.011

TABLE 29 Incidence of wet AMD in the right eye of patients by maximum drusen diameter and drusen type: the Beaver Dam study²⁴

meeting the defined eligibility criteria were recruited over a 3-year period. The study reports separate 1-year incidence rates (with 95% CIs) for dry and wet AMD for all patients with BMD, where dry AMD is defined as involving the fovea (*Table 30*).

Combined incidence rates for AMD are presented by drusen characteristics in the worst affected eye (*Table 31*).

Sunness and colleagues³³ aimed to report the rate of developing wet AMD in eyes with dry AMD and the characteristics of the wet AMD in these eyes. They report separate rates of developing wet AMD in 91 patients with bilateral dry AMD and 13 patients with dry AMD in the study eye and drusen and/or pigmentary abnormalities of the RPE without evidence of dry or wet AMD in the fellow eye. The criteria for dry AMD included at least one area of discrete RPE atrophy of at least

TABLE 30	Incidence of dry	and wet AMD	in þ	atients	with
bilateral dru	sen: the Moorfie	elds study ³⁷			

Age	No.	l-year	95%	6 CI
(years)	cases	incidence	Lower	Upper
Wet AMD >65 All ages	2 3	0.064 0.047	0.0335 0.025	0.1088 0.0782
Dry AMD >65 All ages	5 5	0.026 0.018	0.0086 0.0058	0.0602 0.0407

© Queen's Printer and Controller of HMSO 2008. All rights reserved.

500 mm in diameter within one disc diameter of the fovea.

None of the 13 patients with drusen in the fellow eye developed wet AMD over the follow-up period, though the numbers at risk in each year allow the estimation of upper CIs. Wet AMD incidence rates for persons with bilateral dry AMD, with accompanying CIs, are presented directly. Both sets of data are presented in *Table 32*.

Sunness and colleagues³³ hypothesise that higher rates of wet AMD may be expected in eyes with smaller areas of dry AMD where there is still some intact macular choriocapillaris to support the growth of wet AMD. Their study found that central atrophy less than two disk areas was a risk factor for developing wet AMD in eyes with dry AMD in the fellow eye CNV group. The AMD natural history model does not categorise dry AMD, so the implicit assumption is that the size distribution of dry AMD in the model is equal to that in the studies reporting rates of wet AMD progression from dry AMD.

Two other studies from the late 1980s report the development of wet AMD in eyes with dry AMD, although neither explicitly describes the status of the fellow eyes at either baseline, or within the population who developed wet AMD at follow-up.^{38,39} As the presence of wet AMD in the fellow eye is known to be a risk factor for the development of wet AMD, the data presented by these studies are of limited use in the context of the AMD natural history model.

Age (years)	Cases	Person-years	I-year incidence	9 5%	6 CI
		at risk"		Lower	Upper
Drusen no. central					
<10	0	27	0	0	0.105
10 to 20	I	13	0.0769	0.0019	0.3603
>20	15	138	0.1087	0.0621	0.1729
Drusen no. peripheral					
<10	I	38	0.0263	0.0007	0.1381
10 to 20	5	37	0.1351	0.454	0.2877
>20	10	103	0.0971	0.0475	0.1713
Drusen size (µm) central					
<50	0	13	0	0	0.2058
50 to 500	14	143	0.0979	0.0546	0.1588
>500	2	22	0.0909	0.0112	0.2916
Drusen size (µm) peripheral					
<50	0	21	0	0	0.1329
50 to 500	16	157	0.1019	0.0594	0.1602
Focal hyperpigmentation cent	ral				
Absent	5	118	0.0424	0.0139	0.0961
Present	11	60	0.1833	0.0952	0.3044
Non-foveal dry AMD					
Absent	11	154	0.0714	0.0362	0.1242
Present	5	24	0.2083	0.0713	0.4215

TABLE 31 Combined incidence rates for AMD by drusen characteristics in the worst affected eye: the Moorfields study³⁷

TABLE 32 Incidence of wet AMD in patients with dry AMD: the Baltimore study ³³

Follow-up (years)	Incidence	95 %	6 CI	Annual	95% CI	
		Lower	Upper	incidence	Lower	Upper
Bilateral dry AMD						
1	0	0	0.032	0	0	0.032
2	0.020	0	0.050	0.010	0	0.025
3	0.040	0	0.090	0.014	0	0.031
4	0.110	0	0.210	0.029	0	0.057
Fellow eye drusen						
1	0	0	0.238	0	0	0.238
2	0	0	0.283	0	0	0.153
3	0	0	0.369	0	0	0.142
4	0	0	0.527	0	0	0.171

Commentary on progression from early ARM and dry AMD

Eye one progression from early ARM given no ARM in the fellow eye

A problem with many of the data presented in the literature is that papers do not explicitly describe the status of the fellow eye of eyes with early ARM. It is assumed that unless a population is stated to have bilateral ARM, only one eye is affected.

There is a marked difference between the alternative studies, even after considering differences in definitions. Based on follow-up at 2 and 6.5 years, the Rotterdam study² reports aggregate annual progression rates to dry and wet AMD from early ARM of between 0.01 (60–69 years) and 0.035 (80+ years). The Copenhagen study²² estimates an aggregate annual progression rate of 0.074 for right eyes across all ages, where over 80% of people with early ARM at baseline had progressed to either dry or wet AMD at the 14-year follow-up.

The Beaver Dam study²⁴ presents 10-year incidence data from which annual incidence rates of progression to AMD are estimated as 0.007 for

right eyes with soft indistinct drusen, and 0.016 for right eyes with any pigmentary irregularities. AREDS²⁷ presents aggregate progression rates from early and intermediate ARM to AMD of between 0.015 (<65 years) and 0.033 (>70 years).

The Beaver Dam study²⁴ progression rates appear slightly lower than those is the other studies, although comparison is hindered by the fact that progression rates relating to drusen size, drusen type and pigmentary irregularities are all presented separately, and not as a singly defined early ARM category. The highest rate in the Beaver Dam study is observed for the pigmentary irregularities category, although eyes with pigmentary irregularities alone are not defined as early ARM in the Rotterdam scale. Hence one may expect an equivalent 'early ARM' progression rate (if available) to be in line with the Rotterdam and AREDS rates. The age-specific rates estimated from the Rotterdam study and AREDS are in close agreement.

The obvious outlier is the Copenhagen study, from which a much higher annual progression rate is estimated, although the CIs overlap with all of the other studies.²² As noted above, the higher follow-up rate achieved in the Copenhagen study may be a significant factor in these observed results. As in the previous section, the effect of the lower follow-up rate in the Rotterdam study is a concern, particularly as 38.5% of incident AMD cases were found through home examinations in the Copenhagen study, which may be assumed to be the cases most likely to be missed by the other studies.

A similar approach to that used to specify first eye incidence rates was used to define input parameter estimates for the progression of first eye early ARM, in the absence of ARM in the fellow eye (*Table 33*). The values were based on the Rotterdam data, with the same age-specific uprate factors applied to the low follow-up rates.

Progression from early ARM given early ARM in both eyes

AREDS report no. 19²⁷ states that the paper presents AMD incidence in patients with bilateral AMD, although the detailed description of the AREDS categories included in the paper implies that unilateral drusen were also included. As such, these data were included in the previous section that described unilateral early ARM progression.

The only other identified paper that usefully describes bilateral drusen progression was based at Moorfields Eye Hospital in England.³⁷ Data from this study were used to estimate the incidence of the first case of AMD in patients with BMD. The aggregate over 65 years progression probability is disaggregated into three probabilities representing progression in persons aged 60-69, 70-79 and over 80 years based on the age-specific progression probabilities observed for the progression from early ARM given no ARM in the fellow eye. It is assumed that the observed aggregate probability of progression from early ARM given BMD for persons aged over 65 years applies to persons aged over 60 years. Age-specific progression probabilities were estimated to fit the aggregate progression probability and the observed ratios of the probabilities between the three age groups. The type of incident AMD (dry or wet AMD) in patients aged over and under 65 years is based on the aggregate split between dry and wet AMD in patients aged over and under 65 years, respectively.

Progression from early ARM given dry AMD in the fellow eye

No data were identified describing progression from early ARM given dry AMD in the fellow eye. It was assumed that progression rates are similar to those estimated for progression from early ARM given early ARM in the fellow eye.

Progression from early ARM given wet AMD in the fellow eye

The CNVPT group presented data that allowed the estimation of annual progression rates to dry

TABLE 33 Estimation of uprate factors for first eye progression rates given no ARM in the fellow eye

Age (years)	Mean proportion population at follow-up	Mean late ARM rate	Uprate factor	Revised progression rate
60–69	0.46	0.010	1.20	0.012
70–79	0.38	0.023	1.40	0.032
80+	0.16	0.035	1.60	0.056
Weighted mean (all ages)		0.019		0.027
Weighted mean increase			1.41	

AMD and wet AMD for eyes with early ARM whose fellow eyes have wet AMD.³² Separate data are presented for years 1–4, although only an aggregate rate is derived as the differences between years were not significant. The estimated aggregate annual rates of 0.057 and 0.112 for dry and wet AMD, respectively, were combined with the distribution of second eye wet AMD observed across three age groups of patients experiencing second eye wet AMD given first eye wet AMD (second eyes could have no ARM, early ARM or dry AMD) reported by the MPS group.³¹

Progression from dry AMD given no ARM in the fellow eye

No data were identified describing progression rates for dry to wet AMD in patients with no evidence of ARM in the fellow eye. As the observed data for dry AMD where the fellow eye has drusen show a zero progression rate (as described in the next section), it is assumed that the probability of progression of dry AMD given no ARM in the fellow eye is the same as that observed for the progression of dry AMD given early ARM in the fellow eye.

Progression from dry AMD given early ARM in the fellow eye

Only one study was identified that reported progression rates to wet AMD for patients with dry AMD. Sunness and colleagues³³ report that none of 13 patients with dry AMD and fellow eye drusen developed wet AMD over the 4-year follow-up period. The aggregate number of person-years at risk is estimated by multiplying the estimated follow-up percentages at years 1, 2, 3 and 4 by the numbers of patients dropping out after each follow-up point. The follow-up percentages at years 1, 2, 3 and 4 are based on the reported follow-up percentages at years 2 and 4 and assuming a constant rate of decrease between baseline, year 2 and year 4.

Assuming a binomial distribution, the aggregate upper 95% CI is 0.206. The positive relationship between the probability of progression of dry AMD and age is the same as that observed for the progression of early ARM given no ARM in the fellow eye. Age-specific upper CIs for the dry AMD progression parameter were estimated to fit the aggregate upper 95% CI and the observed ratios of the probabilities between the three age groups for the early ARM (no ARM in the fellow eye) progression parameter. The median age of all participants at baseline (including 91 patients with bilateral dry AMD and 46 with dry AMD in one eye and fellow eye CNV) was 78 years, with a range of 63–94 years. As the specific age distribution of the 13 persons is not reported, the population is assumed to be equally distributed between the three age categories (60–69, 70–79 and over 80 years).

Based on the mean value of zero and the estimated upper 95% CIs for each age category, standard deviations informing normal distributions with mean zero are fitted so that the upper 95% CIs equal the estimated intervals for each of the three age categories. The model can then sample a probability of progression such that if a value between 0 and 0.5 is sampled, the probability of progression is zero, whereas values above 0.5 result in a positive probability of progression.

Progression from dry AMD given bilateral dry AMD

Sunness and colleagues³³ observed progression to wet AMD in four of 91 patients with bilateral dry AMD over a 4-year follow-up period (the follow-up rate at 4 years was 32%). Progression rates, with 95% CIs, are presented for each year of follow-up, from which equivalent annual rates are estimated.

As above, it is assumed that progression probabilities increase with age, and age-specific mean values for the dry AMD progression parameter were estimated to fit the aggregate progression probability (0.029) and the observed ratios of the probabilities between the three age groups for the early ARM (no ARM in the fellow eye) progression parameter. The ages of the four patients at the time of progression are presented (74, 76, 80 and 87 years), although it is only stated that the median age of all participants at baseline was 78 years, with a range of 63-94 years. As the baseline age distribution is not reported, the population is assumed to be equally distributed between the three age categories (60-69, 70-79 and over 80 years).

Given the estimated mean values describing the probability of progressing, and the assumed equal distribution of patients between the three age categories, separate beta distributions are estimated for each age category.

Progression from dry AMD given wet AMD in the fellow eye

Three studies report numbers of cases of progression from dry AMD to wet AMD, given wet AMD in the fellow eye, together with person-years at risk.^{29,31,33} These data are combined to inform mean annual rate of progression of 11.2%

(24 cases in 214.4 person-years), and also a beta distribution for the aggregate rate of progression.

Age-specific rates of progression were estimated based on the ratio of events reported in an MPS study of progression to wet AMD in the fellow eye of eyes with wet AMD.³¹

Progression to subfoveal wet AMD

Eyes with EF wet AMD may progress to JF or SF wet AMD, and eyes with JF wet AMD may progress to SF wet AMD. Occult lesions may progress to MC or PC lesions, and MC lesions may progress to PC lesions.

Patient-level data from patients diagnosed with wet AMD from the study reported by Ali and colleagues were analysed to inform rates of progression.⁴⁰ The dataset contained morphometric analyses of 98 sets of angiograms separated by an interval of at least 3 weeks, with no treatment delivered in the intervening period between angiograms. Fully corrected distance VA measured on logMAR Early Treatment of Diabetic Retinopathy Study charts was available at baseline and at a subsequent visit in 78 subjects.

Initial analyses of combined progression of lesion by type and location were undertaken, but the results show insignificant associations between type and location. It was assumed, therefore, that movement between locations and types of lesion are independent. Separate survival regression analyses were undertaken to inform progression rates from:

- EF to JF
- EF to SF
- JF to SF
- occult to MC
- occult to PC
- MC to PC lesions.

Progression rates were found to be time dependent, and a Weibull survival curve was found to be the best fit to the available data. *Figures 4* and 5 show the estimated mean survival curves for progression between lesion type and location, respectively. The estimated standard errors around each of the survival analysis parameters were used to inform probability distributions around the survival curves that were sampled from multivariate normal distributions.

Given incidence of a non-end state wet AMD lesion type (i.e. occult or MC) and/or lesion location (i.e. EF or JF), the model randomly samples a value between 0 and 1 for sequential months following incidence. If the sampled value is below the sampled probability of progression for the given lesion type and/or location within that month, the lesion is described as progressing. This process of sampling progression of wet AMD continued until either:

- the next sampled event occurred after the individual reached the age of 100 years, or
- the lesion location progressed to SF.



FIGURE 4 Rate of progression between lesion types for occult and MC lesions



FIGURE 5 Rate of progression between lesion locations for extrafoveal and juxtafoveal lesions

Visual acuity

As described in Chapter 2, the AMD screening model estimates QALYs by assigning utility values to the VA level in each individual's best seeing eye in each year of life. The following sections describe the data, assumptions and analyses undertaken to inform the following VA parameters:

- progression of VA in eyes with dry AMD
- progression of VA in eyes with wet AMD.

Visual acuity in dry AMD

Dry AMD does not usually lead to an abrupt loss of vision - a complaint of abrupt vision loss leads one to suspect wet AMD; however, dry AMD can be associated with a loss in VA.⁴¹ Sunness and colleagues report data describing the loss of VA in 139 eyes with dry AMD (defined here as GA), which did not develop wet AMD over the followup period.⁴² The rate of losing three or more lines of VA due to worsening of dry AMD alone in the fellow eye wet AMD group was 31% (95% CI 8 to 47%) at 2 years and 45% (95% CI 15 to 65%) at 4 years. The corresponding rates of a three-line or greater loss in patients with bilateral dry AMD was 29% (95% CI 16 to 39%) at 2 years and 47% (95% CI 29 to 60%) at 4 years. As the rates of decline were not statistically significantly different between the two groups, these data were combined to estimate rates of VA progression due to dry AMD. Table 34 presents the mean parameter values and the 95% CIs that were used to inform uniform probability distributions representing the uncertainty in these parameters.

TABLE 34	Input parameter values describing progression from
drv AMD	

Year since incidence	Cumulat ≩	Cumulative proportion losing ≥3 lines vision				
		95% CI				
	Mean	Lower	Upper			
I	0.15	0.06	0.23			
2	0.41	0.27	0.52			
3	0.60	0.44 0.72				
4	0.70	0.49	0.83			

Rates of loss of six lines or greater for all study eyes were 13% (95% CI 6 to 18%) by 2 years and 29% (95% CI 17 to 39%) by 4 years. Although these data included an additional 13 patients whose dry AMD progressed to wet AMD, they are used to inform the magnitude of VA loss. These data inform the estimate that two-thirds of patients experiencing a loss of at least three lines of vision lose three lines, whereas one-third lose six lines of vision.

Visual acuity in wet AMD

At the point of incidence with wet AMD, lesions are assigned a type (occult, MC or PC) and location (EF, JF or SF). Given a lesion type and location, a VA score at incidence of wet AMD is sampled. The probability distributions of VA scores at incidence were based on subjectively defined ranges of VA that were informed by discussion with clinical expert members of the research team and reflect the assumption that lesions further away from the fovea have fewer

AMD type	Min	Max
Occult EF	0	0.5
Occult JF	0.1	0.7
Occult SF	0.3	1.2
MC EF	0	0.5
MC JF	0.1	0.7
MC SF	0.3	1.2
PC EF	0	0.5
PC JF	0.1	0.7
PC SF	0.3	1.2
Dry AMD	0	0.3

TABLE 35 Estimated ranges of visual acuity (in logMAR units) at incidence of alternative forms of AMD

negative effects on VA. The specified ranges of VA at AMD incidence are described in *Table 35*.

Upon progression, a new VA score is sampled from additional regression analyses of the data from Ali and colleagues' study⁴⁰ (see the section 'Progression to subfoveal wet AMD', p. 31). VA progression was modelled as a function of previous VA, the time in previous vision state and the current lesion type. Other covariates were tested, including current and previous lesion location, but such covariates were found to be highly insignificant. The results of the final regression analysis informing VA progression up to the incidence of SF wet AMD are presented in *Table 36*.

From the point at which a lesion progresses to an SF location, a separate model function describes VA progression over the remainder of individuals' lifetimes. Patient-level data from the placebo arms of the TAP and the VIP trials of PDT (which were restricted to SF lesions) were combined with patient-level data from the SubFoveal Radiotherapy Study (SFRADS).⁴³

The SFRADS compared 12 Gy of external beam radiation therapy (EBRT) delivered as 6×2 Gy fractions to the macula of an affected eye versus observation. Patients were UK-based and had SF CNV and a VA equal to or better than 6/60

(logMAR 1.0). The trial found that there were no statistically significant differences between treatment and control subjects in any measures of VA. Hence the data from both the control and treatment groups are combined to describe VA progression in non-treated SF wet AMD on the advice of a clinical member of the research team (UC).

The maximum duration of follow-up for placebo patients in the TAP and VIP trials was 2 years (placebo patients in the TAP trial crossed over to PDT at 2 years), whereas the maximum follow-up in the SFRADS was 2.67 years.

The combined dataset was adapted to describe the time to failure for each included eye, where failure was defined as a drop in VA of at least five letters (equivalent to a 0.1 drop on the logMAR scale), or the time to censoring if VA did not decline by five letters over the follow-up period. After a VA level drop, any subsequent follow-up data were used to create a new observation, that is, the baseline VA for the new observation is the VA level to which the index eye dropped at the last drop.

The combined and adapted dataset contained 1238 observations. Survival analyses on the combined dataset were undertaken to describe the survival curves for individuals maintaining VA at initial levels. Alternative survival models were tested, but the implemented model was a Weibull model containing the following covariates: age at incidence of SF wet AMD, VA at incidence of SF wet AMD and dummy variables for presence of occult and MC lesions. Figure 6 describes the mean survival curves for individuals maintaining their initial VA by lesion type. The estimated standard errors around each of the survival analysis parameters were used to inform probability distributions around the survival curves that were sampled from multivariate normal distributions.

The curves show that the profiles do not differ greatly by lesion type, although further analysis suggested that the magnitude of the decrease for

TABLE 36 Results of the regression analysis predicting VA progression in lesions up to progression to subfoveal wet AMD

Variable	Regression coefficient	Standard error	t	p > t	95% CI
Baseline VA	0.919	0.067	13.657	0.000	0.786 to 1.052
Log(days) Lesion type ^a	0.048 	0.038	1.258 -1.630	0.211 0.106	-0.027 to 0.123 -0.109 to 0.011
Constant	0.070	0.188	0.373	0.710	-0.303 to 0.443

^a Lesion type was modelled as an ordered categorical variable, increasing severity from occult to MC to PC.



FIGURE 6 Rate of loss of VA in individuals with subfoveal wet AMD by lesion type, for a 71-year-old with 0.3 logMAR VA at incidence

those losing VA does differ more significantly between the lesion types. Additional regression analyses were undertaken to inform the extent of the drop in VA (i.e. number of letters lost) at each drop point. This was a multi-logit regression analysis that described the probabilities that the number of letters lost at each drop point was either 5, 10, 15 or 20. The best fitting model included the following covariates: age at SF incidence or last drop, time since SF incidence or last drop, VA score at SF incidence or last drop and dummy variables for presence of occult and MC lesions. The results of this regression analysis are presented in *Table 37*. Given a set of input parameters (i.e. age, follow-up time, etc.) the regression coefficients are used to estimate probabilities describing the likelihood that a drop in VA is of 5, 10, 15 or 20 letters. *Table 38* provides an example of the conversion process.

As noted above, there is a more noticeable difference in the magnitude of visual loss across the lesion types than was observed for individuals losing any VA (as shown in *Figure 6*). PC lesions are more likely to be associated with larger drops

Letters lost	Parameter	Regression coefficient	Standard error	Z	p > z	95% CI
10	Age (years)	0.007	0.013	0.534	0.593	-0.019 to 0.033
	Follow-up time	-0.00 I	0.001	_I.478	0.139	-0.002 to 0.000
	Baseline VA	0.013	0.007	1.985	0.047	0.000 to 0.027
	Occult	0.201	0.203	0.989	0.323	-0.197 to 0.598
	MC	-0.454	0.217	-2.094	0.036	-0.880 to -0.029
	Constant	-1.655	1.075	-I.539	0.124	-3.762 to 0.453
15	Age	-0.024	0.015	-1.599	0.110	-0.053 to 0.005
	Follow-up time	0.000	0.001	-0.561	0.575	–0.002 to 0.001
	Baseline VA	0.006	0.008	0.774	0.439	–0.009 to 0.021
	Occult	-0.137	0.233	-0.587	0.557	–0.595 to 0.321
	MC	-0.522	0.241	-2.170	0.030	-0.993 to -0.051
	Constant	0.716	1.190	0.602	0.547	–1.617 to 3.048
20	Age	0.002	0.014	0.123	0.902	-0.026 to 0.029
	Follow-up time	-0.001	0.001	-1.506	0.132	-0.002 to 0.000
	Baseline VA	0.050	0.007	6.824	0.000	0.036 to 0.065
	Occult	-0.673	0.219	-3.071	0.002	-1.103 to -0.244
	MC	-0.958	0.229	-4.176	0.000	-1.407 to -0.508
	Constant	-2.832	1.150	-2.463	0.014	-5.087 to -0.578

TABLE 37 Results of a multi-logit regression analysis to predict the magnitude of decrease of VA in subfoveal lesions

Regre	ssion coefficient	values		Probabilities		
2 levels	3 levels	4 levels	l level	2 levels	3 levels	4 levels
-1.4495	-1.5769	-1.9291	0.6302	0.1479	0.1302	0.0917

TABLE 38 An example of the estimation of the probability distribution describing the magnitude of drop in VA for patients with subfoveal wet AMD

in VA. The estimated standard errors around each of the survival analysis parameters were used to inform probability distributions around the survival curves that were sampled from multivariate normal distributions.

This process of sampling progression of VA in subfoveal lesions continues until either:

- the next sampled VA drop occurred after the individual reached the age of 100 years, or
- VA decreased to under 20 letters.

Further survival regression analyses describing the progression of lesion type in SF wet AMD were undertaken, which used data from the Ali and colleagues' dataset⁴⁰ to describe the annual probability that occult or MC lesions progress. Due to the small number of observations (two occult lesions and three MC lesions were observed to progress), the analyses were not linked to VA progression.

Chapter 4

Screening and treatment parameters

This chapter presents the available evidence regarding the effectiveness of screening tools for detecting AMD and on interventions for treating different stages of AMD. The screening model parameters to be informed include the probability of individuals detecting an abnormality and presenting at their opticians, to be defined by severity of disease, that is, early ARM, dry AMD and wet AMD. The screening process also includes the probabilities that opticians would refer on patients with the different severities of disease to the hospital eye service.

An effective screening programme must be able to offer an appropriate treatment for the condition in the cases detected. At clinical presentation, it was also necessary to define the treatment options available for individuals with different severities of disease. The ARM screening model facilitates two broad forms of intervention for ARM. First, interventions may be specified that reduce the risk of disease progression, for example, of early ARM progressing to dry or wet AMD or dry AMD progressing to wet AMD, or that reduce the risk of the incidence of ARM in the fellow eve of the first affected eye. Second, interventions may be implemented that reduce the loss of vision in eyes due to either dry AMD or different forms of wet AMD.

The first sections of this chapter describe the limited evidence on the screening parameters, which led to the implementation of an expert elicitation exercise to inform these parameters further. Subsequent sections describe the data and assumptions informing the impact of interventions on the natural history of ARM.

Screening parameters

There was little evidence to suggest that widespread use of the Amsler grid would prove to be a valuable screening tool.⁴⁴ *Table 39* shows the studies which have undertaken work looking at screening tools, and illustrates the problem that the present study had in identifying a simple and effective tool which would be used in the community. It became apparent that this would be a major area that would need to be referred to our panel of experts. **Screening parameter expert elicitation** Once the searches were completed, we were able

to see where the literature lacked data. We began the process of contacting experts in the field of this branch of ophthalmology. A short presentation describing the aims and objectives of the study was delivered at a number of key scientific meetings in the UK. The presentation explained about the study, why we needed the input of a number of experts and what kinds of information, including any unpublished data in which the team were interested. This was followed up by written and telephone communications to a number of individuals who had expressed an interest in collaborating with the team.

A number of visits were undertaken by team members to meet with the experts on an individual basis. We convened a panel of six experts (UC, AB, IC, YCY, SH, CB). Members of the core study team met with these individuals to develop the elicitation study.

A questionnaire was developed out of these meetings which addressed a number of issues, including screening tests, referrals from optometrists, treatments and future possibilities. We posted this out to the panel members with a letter explaining that they could complete a paper version or an electronic version or talk directly to a member of the study team (see Appendix 3). The experts were contacted again if any clarification was required.

In addition, as part of the collaborative process of this work, a number of the clinicians made available existing datasets and additional unpublished data from previous studies. We were able to reanalyse these and incorporate them in the model.

We also undertook a small survey of ophthalmologists (n = 30) who referred patients to either the Sheffield or Liverpool PDT clinics in order to inform estimates of the number of patients presenting with AMD, and the distribution of AMD states at presentation. These data inform the model calibration process and are described in more detail in Chapter 8. This survey document is presented in Appendix 4.

Study	Study design	Inclusion/exclusion	Numbers	AMD type	Statistical results	Comments
AREDSRG (2001) USA ²⁵	Standardised photographs Cumulative sample	Not described	1230 eyes	Advanced AMD	Agreement 96% (kappa = 0.88) Satisfactory reliability for detecting onset of advanced AMD	Suitable for longitudinal multi-centre studies
Barthes (2001) France ⁴⁵	Computerised image processing Compared with results of 3 specialists' counts of drusen	Not described	58 retinal images	Drusen only	Manual counting inter-observer correlation coefficients lie in the range 0.71–0.78 A correlation coefficient of 0.89 is obtained when the average of the three expert countings is compared with the drusen number given by the computerised method	Analysis is quicker but manual validation is still required
Fine (1986) USA ⁴⁶	Case series of 103 patients with neovascular maculopathy and relatively recent vision loss	Consecutive patients – interview (first visual symptom) asked about Amsler grid use	103 patients, 68 with AMD	AMD	Of 89 who previously had the Amsler grid, only 49 were actually using it to monitor their central vision on a regular basis. Although all but one of these 49 patients had a demonstrable grid defect, only five stated that the first visual symptom was noticed while observing the Amsler grid	Targeted use of the grid does not seem to be effective
van Leeuwen (2003) Europe ⁷¹	EUREYE study data Comparison of stereo digital images with stereo 35 mm transparencies	Validation study in Belfast and Rotterdam, 65+-year-olds invited for eye examination	137 eyes (91 subjects)	ARM according to International Classification System	Our study showed there were no important differences between both photographic techniques in the grading results in an epidemiologic setting	Practical advantage of digital imaging
Schmitt (2003) USA ⁴⁷	Evaluation trial	92 patients with vision 20/80 or better aged 50+ years	92 eyes tested	AMD, 30 people	We found EMS RT distribution did not differ between AMD, cataract, diabetic retinopathy and glaucoma groups	Not a sensitive screening tool
Woods (1998) Australia ⁸⁸	Retrospective analysis of clinical cross- sectional survey	Randomly selected residents aged over 50 years from an Australian health district	3283 subjects, diagnosis confirmed in 2522	AMD and other pathologies	A person older than 50 years of age with reduced contrast sensitivity, as determined by Arden plate 7, requires extra care in subsequent examinations because that person is likely to have an ophthalmic disease	Arden 7 had a PPV of 0.96 and an NPV of 0.97 in the population studied
EMS, Eger mac	ular stressometer; NPV, ne	sgative predictive value; PP	V; positive predictive	e value; RT, recover)	/ time.	

TABLE 39 Summary of screening studies

38

The survey included both qualitative and quantitative responses. The qualitative responses are summarised, highlighting areas of general agreement and any specific detail that was relevant to the model. The quantitative responses are presented as ranges that were used to inform the relevant model parameters.

Regarding the choice of screening tools and their fitness for purpose, the panel indicated that there was some use for the Amsler chart in clinical practice. The experts were clear that the Amsler grid was of some use with patients, but that it was not an appropriate test to be used by optometrists to screen patients and refer on to an ophthalmologist. The problem highlighted with any test was that they cannot detect recent onset or progression.

It was generally agreed that it was appropriate to give patients instructions to check for visual distortion by covering one eye and observing an object with fixed horizontal lines, for example door or window frames. Patients would be asked to repeat this for the second eye. The abnormality would be described as disturbances, blurring or distortion of the central vision.

Quantitatively, the experts were asked to identify the proportion of people using the cover test who would notice an abnormality when self-testing. *Table 40* shows the results as the range of the percentages of people who the experts predicted would detect AMD using the self-completing cover test.

It was questioned whether macular haemorrhage was an appropriate indicator of wet AMD and, if not, what other indicators could be used by optometrists to identify suspected wet AMD. It was agreed that macular haemorrhage was easily detected if present, but other indicators such as subretinal fluid, exudate and hyperpigmentation

TABLE 40 Percentage of people detecting visual distortion with cover test by AMD type

AMD type	Proportion noticing visual distortion after self-completing a cover test (%)	
No AMD	<1-5	
Drusen	2–5	
Dry AMD	5–8	
EFCNV	10–30%	
JF CNV	30–80	
SF CNV	60–100	

was a high referral rate with any signs of wet AMD, but that some indicators could be overlooked by optometrists even if patients had early symptoms described as photo stress or dazzle.

were also important and these could be found by

optometrists using a slit lamp. It was felt that there

The second quantitative question concerned the proportion of patients presenting at opticians who would be referred on to a hospital eye service, by disease severity. *Table 41* shows the ranges of percentages estimated by the expert panel of ophthalmologists, accounting for the use of macular haemorrhage as a marker for wet AMD, and for other indicators of wet AMD.

Specific questions were asked about the stage of disease at which NHS consultants would expect patients to be referred to the hospital eye service. Although there was some small variation in the comments, there was generally more agreement that suspected wet AMD should be the main referral criterion. There was more willingness to see other patients in the academic unit, which perhaps reflects individual research interests.

Model screening parameters

The above responses informed the definition of screening process in the AMD screening model that was based on self-testing within the eligible population, driven by a reminder letter at specified intervals. Self-defined positive screens are advised to present at an optometrist's for examination and possible referral on to a hospital eye service.

The screening parameters included in the model describe the likelihood that individuals who receive a reminder to self-test for AMD actually undertake the advised self-test, and also the likelihood that individuals with different forms of ARM in one or both eyes present at an

TABLE 41	Proportion referred to consultant ophthalmologist by
optometrist	s by AMD type

AMD type	Macular haemorrhage	Other indicator of CNV
No AMD		I <i>—</i> 5
Drusen		5–15
Dry AMD		3–25
EF CNV	10-100	30–70
JF CNV	30-100	30–75
SF CNV	25–100	50–75

optometrist following a self-test. No data were available to inform the probability that individuals who receive a reminder to self-test for AMD undertake the advised self-test. Such parameters are not specific to the disease process and so were not able to be estimated as part of the expert elicitation study. Therefore, the rate of uptake of screening was described as a uniform distribution ranging from 0.3 to 0.7. *Table 42* presents the applied ranges for the probability of screen detection according to the ARM state of the worst affected eye, which were represented by uniform distributions in the model.

As described in Chapter 2, the costs and benefits of individuals without ARM presenting at an optometrist as a result of an AMD screening programme are assumed to be neutral, that is, the benefits of the interventions received equal the costs. Therefore, the AMD model does not incorporate a false positive rate.

Treatment parameters

The systematic review did not include individual trial papers on treatments for AMD, although high-quality reviews of treatment options were assessed. However, treatment options for AMD are currently in a state of flux, with multiple intervention options at various stages of the evaluation and regulatory process. The first stage of the analysis incorporated expert elicitation to inform the appropriateness of current treatment options, and also to inform a process of horizon scanning to identify future possible treatments. The second stage involved the analysis of any available data to populate the model's treatment parameters.

Treatment parameter expert elicitation

The survey used to inform the screening parameters also included questions relating to treatment options for AMD. We wanted to obtain

TABLE 42 Screening parameters given the stage of AMD in theworst eye

ARM state	Minimum	Maximum
Probability of presentation at	optometrist	
Early ARM	0.02	0.08
Dry AMD	0.02	0.1
Probability presentation and I	referral to hospit	tal eye service
EF wet AMD	0.1	0.3
JF wet AMD	0.5	0.8
SF wet AMD	0.6	0.95

a spread of opinion about the treatment options for various subgroups of patients. We asked first about treatment options other than PDT that would be considered for subfoveal lesions. All agreed that pegaptanib would be a choice of treatment. In addition, one suggested that intravitreal steroid injections (IVSI), argon laser, transpapillary thermo-therapy (TTT) and inhibitors of vascular endothelial-derived growth factor (anti-VEGFs) could all be considered.

For JF lesions, pegaptanib, PDT and steroid use were also considered along with argon laser. They were asked about treatments for EF lesions and whether they agreed that laser photocoagulation was the only treatment. The experts were divided in that two agreed that laser would be the first line of treatment, whereas another concurred but only for classic lesions, as the evidence did not support any intervention for occult EF lesions. The other two experts agreed that there was some scope for clinical judgement and that PDT may be useful along with IVSI or anti-VEGFs.

To inform sensitivity analyses of the model that assessed the potential impact of future interventions for AMD, the experts were asked to engage in some horizon scanning in the area of treatments for dry AMD. They were not hopeful; only one suggested RPE cell transplantation as a possible option. The same exercise was set for prevention of progression to wet AMD and how many cases could be prevented. The experts were more divided, as three did not think that there would be major changes in prevention interventions and did not give any numbers for cases prevented. Two thought that between 5 and 8% of cases may be prevented. Lifestyle changes, mainly stopping smoking, were considered useful generally.

The experts were also asked about what advice, if any, patients could expect to receive if they were not going to be seen in hospital. Lifestyle advice about stopping smoking would be the main advice offered. There was less agreement about diet, although most thought it reasonable, but vitamin supplements were more controversial. Those patients with visual loss due to dry AMD were likely to be referred to the low vision service, but this seems to be dependent on provision within the hospital. Any further monitoring was variable as some did not wish to monitor and others recommended reporting any changes in fellow eyes.

Current treatment model parameters

The base case analysis of the model follows current guidelines in the UK NHS, which do not include the use of interventions aimed at reducing progression from early ARM or dry AMD, or second eye ARM incidence. The main purpose of detecting individuals with early ARM is to inform them about the increased likelihood of developing AMD and so encourage more frequent self-testing. In the screening model, it is assumed that individuals with detected early ARM present, and receive appropriate intervention, at the point at which wet AMD develops. This is recognised to be an optimistic assumption, which would be a key aspect to be assessed in any pilot study of a screening programme for AMD.

PDT with verteporfin is the principal current treatment option for wet AMD in the UK, although the likelihood of treatment is dependent on the type and location of the lesion. *Table 43* describes the assumed treatment options for wet AMD by type and location. Once an eye has been treated with either PDT or triamcinolone, it is assumed that the same eye cannot receive additional intervention if and when the lesion progresses; for example, if an eye with an occult lesion is treated with PDT, and later progresses to PC, the eye does not receive a new course of PDT.

The effectiveness parameters for PDT were defined to describe the additional proportion of patients receiving PDT who maintained their VA at their initial level compared with patients receiving placebo. To estimate effectiveness parameters for PDT, we had access to the patient-level data from the TAP and VIP trials. Analyses of the TAP and VIP data require weak assumptions about the longterm effects of treatment, as only 2-year follow-up data are available. In line with the above conservative assumptions regarding clinical presentation with AMD, optimistic estimates of the effectiveness of PDT treatment are defined. The main assumption is that patients in the TAP and VIP trials who lost fewer than 15 letters at the 2-year follow-up point maintain the same level of VA for the remainder of their lifetime. The data

were analysed to estimate the absolute risk reduction (ARR) in the proportion of patients losing 15 letters or more, which was then defined in the screening model as the probability that patients receiving PDT maintain their vision from the point of treatment.

Table 44 presents the results of the analysis of the TAP and VIP trial data, by wet AMD lesion type. Analyses of MC lesions, and for 15 letters lost and also 10 letters lost, are presented for comparative purposes. The data show that the baseline VAs between the PDT and placebo groups of each lesion type were very similar, so the ARR is simply estimated as the difference in the proportions of patients losing the defined number of letters at 24 months. PDT is shown to have the greatest effect on patients with PC lesions, with the ARR for occult lesions being approximately half that for PC lesions. The ARR for MC lesions is significantly lower, as expected, as PDT is not a current treatment option for these types of lesion. The proportions losing >10 letters at 24 months are described as beta distributions in the model, with the ARRs for PC and occult lesions being estimated for each iteration of the model based on the sampled values from the relevant distributions.

Estimates of the effectiveness of laser photocoagulation for the treatment of EF lesions are based on the reported results of a trial that randomised 119 patients to argon blue-green laser treatment and 117 to no treatment (Table 45). Inclusion criteria included EF lesions with VA of 20/100 or better. The trial had a 5-year follow-up period.49 Results are only presented for the proportion of eyes with >30 letters lost at 5 years. To inform estimates of effectiveness, it was optimistically assumed that the additional proportion of eyes in the treatment group without >30 letters lost at 5 years maintained their vision at their initial VA over the remainder of their lifetime. Beta distributions were used to describe the uncertainty in the proportions.

TABLE 43 Treatment options for wet AMD with VA better than 6/60^a

Lesion location		Lesion type				
	Occult	МС	РС			
EF JF/SF	Laser photocoagulation Not treated	Laser photocoagulation Not treated	Laser photocoagulation Photodynamic therapy			
^a Eyes with VA worse than or	equal to 6/60 do not receive treat	ment intervention, but are re	ferred to the low vision			

service.

TABLE 44 Effectiveness data from the TAP and VIP trials forPDT

Variable	N	Mean
Proportion with > 15 letters lost at 24 r	months	
PC on PDT	129	0.38
PC on placebo	73	0.67
MC on PDT	172	0.53
MC on placebo	84	0.55
Occult on PDT	174	0.51
Occult on placebo	93	0.69
Absolute risk reduction		
PC on PDT		0.29
MC on PDT		0.02
Occult on PDT		0.18
Proportion with > 10 letters lost at 24 r	months	
PC on PDT	129	0.54
PC on placebo	73	0.79
MC on PDT	172	0.60
MC on placebo	84	0.65
Occult on PDT	174	0.61
Occult on placebo	93	0.74
Absolute risk reduction		
PC on PDT		0.25
MC on PDT		0.05
Occult on PDT		0.13

Potential treatment model parameters

A range of potential treatment options for ARM are described in the following sections. Estimates of the potential effectiveness of alternative interventions for different stages of the disease are specified and tested in sensitivity analyses of the AMD screening model.

Interventions aimed at preventing progression from early ARM or second eye ARM incidence

The potential impacts of preventative interventions are explored in additional analyses. Two systematic reviews of antioxidant vitamin and mineral supplements for ARM were identified,^{10,50} including a total of eight trials, six of which were common to both studies. Excluding the ABTA study of β -carotene and vitamin A,⁵¹ as it did not include a baseline measurement, the AREDS study comprised over 88% of the total numbers of individuals randomised.⁵⁰ As a meta-analysis was not deemed possible by the authors of the reviews

TABLE 45 Proportion of EF lesions with >30 letters lost at 5 years: MPS group Senile Macular Degeneration Study (SMDS)

Treatment group	N	Proportion
Laser photocoagulation	119	0.48
No treatment	117	0.62

(due to the trials using different formulations), and given the dominance of the AREDS study, the estimates of the potential effectiveness of preventative measures were based on the results of the AREDS study.

Unpublished data describing progression of individuals with drusen up to size 63 μ m, or a maximum diameter of 125 μ m (about 5–15 small drusen) and no pigment abnormalities, from the AREDS trial were kindly made available to the current study. One out of 451 individuals at risk and three out of 490 at risk in the antioxidant and placebo arms, respectively, experienced AMD (dry or wet) over a 5-year follow-up period. These data inform an estimated relative risk (RR) of AMD in individuals receiving antioxidant supplements of 0.36 (with very wide CIs). These data were considered too unstable for use in the model.

Published data include ORs describing the likelihood of individuals with bilateral drusen (early ARM) or unilateral AMD (dry or wet) experiencing AMD in eyes without AMD at baseline. Based on these data, in the sensitivity analyses it was assumed that individuals presenting with early ARM or unilateral AMD receive zinc supplements that reduce their risk of developing wet AMD and dry AMD by 15 and 20% in individuals with early ARM, respectively, and by 47 and 24% in individuals with unilateral AMD, respectively. Log-normal distributions were estimated for each of the four defined RRs based on the reported CIs, as presented in *Table 46*.

Interventions aimed at treating dry AMD

At present, ophthalmologists do not intervene for patients presenting with dry AMD, and the reference case analysis assumes that individuals with detected dry AMD present, and receive appropriate intervention, at the point at which wet AMD develops.

TABLE 46	Relative risks associated with zinc supplements:
AREDS	

	Mean	95% CI	
		Lower	Upper
Early ARM to Dry AMD Wet AMD	0.8 0.85	0.49 0.57	1.34 1.28
2nd eye incidence of Dry AMD Wet AMD	0.53 0.76	0.35 0.35	0.81 1.65

A clinical trial evaluating the safety and effectiveness of treatment with the investigational drug anecortave acetate to a sham procedure with the aim of preventing progression of dry AMD to wet AMD is currently ongoing. Eligible patients must currently have (or have previously had) wet AMD in one eye and dry AMD in the other eye, and be at risk of having their dry AMD progress to wet AMD.⁵² Although no effectiveness data are available, this is a potentially important area of development of treatment. Due to the absolute uncertainty around the potential effectiveness of this intervention, a uniform probability distribution of RRs of between 0.25 and 0.75 is specified.

Another intervention, RHEOVision Treatment, is currently being evaluated as part of a small clinical trial for the prevention of vision loss in patients with dry AMD. This intervention is a specialised form of blood cleansing technique that aims to remove excess levels of substances known to thicken the blood, decrease blood flow and cause damage to capillary vessels. A recently published paper of interim results from the trial for the first 43 treated patients showed that at 12 months, 28% of the rheopheresis-treated eyes versus 18% of the placebo-treated eyes showed an increase of at least two lines of best corrected visual acuity (BCVA), and 12 versus 0%, respectively, showed an increase of at least three lines.⁵³ Pulido⁵⁴ also present data showing that, in all patients, 8% of the rheopheresis-treated eyes versus 18.2% of placebotreated eyes showed at least a two-line loss in BCVA to 12 months. These effectiveness data informed estimates of the additional proportion of treated eyes that gain three lines of VA (0.12), and the additional proportion who did not lose vision (0.102). Log-normal distributions were used to describe the uncertainty in the 'difference in proportion' estimates, based on the standard errors presented in Table 47.

A surgical option for the treatment of dry AMD involves the transplantation of sheets of fetal neural retina together with its adjacent retinal

TABLE 47	Effectiveness	estimates	for	RHEOvision
----------	---------------	-----------	-----	------------

	Improve VA by 3 lines	Maintain VA
RHEOvision probability	0.12	0.92
No. at risk	43	43
No treatment probability	0	0.82
No. at risk	43	43
Mean ARR	0.12	0.102
Standard error	0.05	0.07

pigment epithelium into the space under the fovea. The most recent report on this technique describes its effect on only one patient with dry AMD (and four with retinitis pigmentosa). The dry AMD patient had a pre-surgery VA of 20/640, which improved to 20/240 at 6 months postoperation, but declined to 20/400 at 12 months. Based on such preliminary data, the model does not assess the potential impact of this surgical procedure for dry AMD.

Interventions aimed at treating wet AMD

In addition to the currently used interventions for treating wet AMD, two additional analyses explored the potential impact of other intervention options that have recently been discussed in the literature. These options are summarised in *Table 48* and discussed below.

In the USA, the debate at the time of writing appears to be around the use of combination therapy (PDT + triamcinolone) for all lesions, that is, PDT is provided to patients with MC wet AMD. A poll taken among attendees of the annual meeting of the American Society of Retina Specialists in 2004 showed that 80% were using combination therapy.⁵⁵ Little evidence of the effectiveness of intravitreal injection of triamcinolone acetonide was available, and some of the data identified indicate that triamcinolone is least effective in eyes with MC wet AMD. Jonas and colleagues reported a mean loss in VA for 18 eyes with MC wet AMD, while reporting mean

TABLE 48	Alternative	treatment	option	sets	for wet	AMD ^a
----------	-------------	-----------	--------	------	---------	------------------

Treatment	Lesion	Lesion type		
option set	location	Occult	МС	PC
 2	JF/SF IF/SF	PDT + triamcinolone Anti-VEGF therapy	PDT + triamcinolone Anti-VEGF therapy	PDT + triamcinolone PDT + Anti-VEGF therapy
-	j .,		· ····································	· - · · · · · · · · · · · · · · · · · ·

^a Eyes with VA worse than or equal to 6/60 do not receive treatment intervention, but are referred to the low vision service.

gains for eyes with occult or 100% classic wet AMD. 56

Bhavsar⁵⁷ reported a retrospective review of 26 eyes of 23 patients with MC SF wet AMD, who were treated with PDT and 4 mg of triamcinolone. At a median follow-up of 6 months, 13 eyes experienced an increase in VA; 23 eyes (88%) gained acuity or lost less than three lines.

The largest case series to date has recently been published, which provided combination PDT + triamcinolone to 184 patients with wet AMD and a mean baseline VA of 20/125.⁵⁸ Aggregate results are presented, which show that the average increase in VA was between 1.22 and 1.43 lines, although no data on the proportions maintaining and improving vision are reported. The paper does state that there was no difference in response regarding lesion type.

The data reported by Bhavsar⁵⁷ were used to inform the potential effectiveness of the PDT + triamcinolone combination. An ARR of 0.15 is estimated by comparing the 6-month rates of visual loss reported by Bhavsar and the MC patients receiving placebo in the TAP trial (0.12 vs 0.27). After adjusting for differences in the ARRs for >15 letters and >10 letters lost for PDT (as reported in Table 44), a final ARR for PDT + triamcinolone of 0.11 was assumed. It is recognised that the assumptions informing the potential effectiveness of triamcinolone for MC CNV lesions are weak and based on questionable indirect comparisons, but they are in line with the optimistic assumptions made about the effectiveness of PDT.

The other potential intervention that is tested in the sensitivity analysis is a generically specified form of anti-VEGF endothelial growth factor therapy. Vascular endothelial growth factor (VEGF) is a chemical that is critical in causing abnormal blood vessels to grow under the retina. Recently reporting clinical trials have reported promising results around the use of anti-VEGF therapies that can block VEGF and hence reduce the growth of abnormal blood vessels, slow their leakage and help to slow vision loss.⁵⁹

Pegaptanib sodium (pegaptanib) blocks VEGF, which is thought to be one of the underlying mechanisms promoting the occurrence of wet AMD. The VEGF Inhibition Study in Ocular Neovascularisation (VISION) found that at 2 years 59 and 45% of patients receiving pegaptanib 0.3 mg and the sham intervention, respectively, had lost less than 15 letters, whereas 27 and 10% had gained at least two lines of vision, respectively. The proposed advantage of pegaptanib over PDT is that it is equally effective for all types of wet AMD, that is, pegaptanib could be preferred in MC and occult lesions. In the VISION, in all the treatment groups, an average of 8.5 injections were administered per patient out of a possible total of nine injections.⁶⁰

The other principal potential intervention that has been evaluated as part of a randomised clinical trial is ranubizimab, which binds to and inactivates VEGF. The MARINA trial compared ranubizimab with a sham injection in eyes with MC or occult wet AMD. The 1-year MARINA data showed that 95% of ranubizimab patients had lost fewer than 15 letters compared with 62% in the control group, and between 25 and 34% (depending on dose) gained 15 or more letters compared with 5% in the control group, with 40% of ranubizimab patients achieving VA of 20/40.61 The ANCHOR trial compared ranubizimab with PDT in PC lesions. The 1-year ANCHOR data showed that 95% of ranubizimab patients had lost fewer than 15 letters compared with 64% in the PDT group, and between 36 and 40% (depending on dose) gained 15 or more letters compared with 6% in the PDT group.⁶¹

A Phase I/II trial, FOCUS, compared ranubizimab in combination with PDT with PDT alone in eyes with SF wet AMD. At 1 year, 90% of the ranubizimab + PDT group had stable or improved VA compared with 68% of the PDT alone group. Ranubizimab is provided in four weekly cycles, although no data are presented on the number of ranubizimab treatments received by patients. A new trial, the Safety Assessment Intravitreal Ranubizimab for AMD (SAILOR) trial, that evaluates safety, states that ranubizimab will be administered once per month for 3 months and thereafter as needed based on retreatment criteria (which are unstated).

An interesting development in the treatment of wet AMD is the increasing usage of off-label bevacizumab (which Genentech developed for cancer treatment) as treatment for MC and occult forms of wet AMD. In the USA, off-label use of bevacizumab is quickly replacing pegaptanib, due to its more impressive success rate and lower cost.⁶² At the 2005 American Academy of Ophthalmology annual meeting, Genentech presented a range of concerns about the drug's potentially adverse effects when used intravitreally for wet AMD. However, Dr Rosenfeld (who is leading the study of bevacizumab for retinal

44

treatment at the Bascom Palmer Eye Institute in Miami) stated that his laboratory has had no safety issues so far, and that the majority of concerns raised by Genentech are theoretical and have not been identified in actual practice.

Gragoudas and colleagues⁶⁰ identify other anti-VEGF interventions that are being developed, including VEGF-Trap that "competitively binds" VEGF, and which is in early phase trials. Studies of RNA molecules that target specific aspects in the VEGF receptor mRNA are also ongoing.

The impact of a generic anti-VEGF therapy is tested in the sensitivity analysis, where the most optimistic estimates of the potential effectiveness of such an intervention are specified, based on the observed results reported by clinical trials to date. The main eligibility criterion for these treatments appears to be that an eye's RPE remains intact, which provides the potential for visual improvement with such biological interventions. RPE status can be approximated as a function of lesion size with lesions smaller than 7 mm^2 usually having an intact RPE, whereas in lesions of size 10 mm² or greater the RPE is usually removed [information provided by one of the authors (AB)]. However, as only aggregate results have been published, and as the screening model does not describe lesion status by RPE status or lesion size, the effectiveness of anti-VEGF monotherapy is assumed to be equal across all MC or occult JF and SF wet AMD. Assumed rates of VA maintenance and improvement in patients with MC or occult CNV are as follows:

- 30% improved their VA by three lines compared with 5% in the control group.
- 75% of treated patients maintained their vision at the same level as at diagnosis compared with 29% in the control group.

In patients with PC JF and SF wet AMD receiving PDT + anti-VEGF therapy:

- 38% improved their VA by three lines compared with 6% in the control group.
- 96% of treated patients lost fewer than 15 letters compared with 64% in the control group.

Uncertainty is represented by normal distributions describing the uncertainty in the difference in proportions between each of the outcomes in the two treatment groups, as presented in *Table 49*.

Other potential treatment options include submacular surgery, which has been tested in patients with new and recurrent SF CNV. Thomas and Ibanez⁶³ report, however, that no difference in the proportion of patients losing two or more lines was observed between a surgery and observation group in a total of 454 eyes, and that surgery is not recommended for any subgroup. An alternative form of surgery for AMD patients with SF CNV involves the translocation of a retinal choroidal patch. van Meurs and van den Biesen⁶⁴ report a case series of 41 patients (29 MC, nine occult, three classic) with baseline VA ranging from 20/400 to 20/80. At 12 months' follow-up, 67% of patients had lost two lines or less, and eight patients had VA of 20/80 or better (only one had 20/80 vision at baseline). Submacular surgery is not included in the additional analyses of potential interventions. Retinal choroidal patch grafting appears to be more promising, especially as the procedure is described as being relatively simple (involving a one-step, 1.5 hour process) that is applicable to a wide range of neovascular membranes (e.g. MC, very large). Moreover, the authors imply that more effective procedures based on "more sophisticated upgraded cultivated RPE cells on a suitable artificial substratum" may soon be available. The clinical impact of such interventions may be that CNV becomes a curable disease such that intervention can be delayed until vision declines beyond an acceptable level (e.g. 20/100), which may reduce the need for screening as all clinically presenting lesions will be treatable.

TABLE 49 Estimated effectiveness of anti-VEGF therapy for juxtafoveal and subfoveal wet AMD

	Occult or MC		PC		
	Improve VA by 3 lines	Maintain	Improve VA by 3 lines	Maintain	
Probability (anti-VEGF)	0.30	0.75	0.38	0.96	
No. at risk	480	480	282	141	
Probability (no anti-VEGF)	0.05	0.29	0.06	0.64	
No. at risk	240	240	282	141	
Mean ARR	0.25	0.46	0.32	0.32	
Standard error	0.03	0.04	0.03	0.04	

© Queen's Printer and Controller of HMSO 2008. All rights reserved.

Chapter 5 Costs and utility parameters

The screening model incorporates three broad categories of costs: screening, diagnosis and surveillance costs, treatment costs and costs associated with blindness and rapidly deteriorating vision. The broad set of assumptions for the base case analysis and the additional sensitivity analyses are presented in *Table 50*. The following sections describe the data and assumptions informing the estimated cost values used in the screening model in more detail.

Screening, diagnosis and surveillance costs

The actual process of screening is assumed to be a costless activity, as it is assumed to comprise a simple self-test; for example, individuals are advised to cover each eye in turn and look at the window frame, and if the window frame appears not to be straight the individuals are advised to attend an optometrist's. Costs are incurred in the organisation of the screening programme, which is assumed to involve the distribution of personalised reminders to all individuals aged over 50 years on a 3-yearly cycle.

Other economic studies of mass screening programmes have estimated the cost of administering screening invitations, although no empirical evidence of the administration costs of screening were identified. A pilot evaluation of liquid-based cytology for screening for cervical cancer stated that there will be economies of scale benefits in having letter administration coordinated in larger centres, and assumed that the cost of administration of letters to the women was £3 per smear.⁶⁵ Roderick and colleagues⁶⁶ describe the costs of inviting and administering the general population sample of around 4000 to attend a screening for infection with Helicobacter *pylori* to be £2, although it is not clear if this is the cost per attendee or per invitation.

TABLE 50 Costing assumptions for the base case and additional sensitivity analyses

Base case analysis	Additional sensitivity analyses					
Individuals present at an optometrist with identified early ARM, dry AMD or wet AMD						
The costs and benefits of individuals presenting at an optometr AMD, are neutral from the perspective of the NHS and are no	ist without ARM, as a result of a screening programme for t included in the AMD screening model					
Individuals with early ARM are managed in the community and deterioration	told to re-attend at the optometrist's upon vision					
Individuals with dry AMD or wet AMD are referred on to the	hospital eye service for diagnosis					
Individuals with JF or SF PC wet AMD receive PDT. Individuals with EF wet AMD receive laser photocoagulation (PDT)	Individuals with early ARM receive treatment advice from an optometrist. A range of treatment options is specified for different forms of AMD					
Individuals with early ARM, dry AMD, EF wet AMD, JF or SF occult or MC wet AMD are advised to self-test their vision at regular intervals and to re-attend if and when their vision deteriorates	Individuals with early ARM, dry AMD or EF wet AMD are advised to self-test their vision at regular intervals and to re-attend if and when their vision deteriorates					
Individuals re-attend upon vision deterioration, a maximum of once per year until they develop either JF or SF PC wet AMD	Individuals re-attend upon vision deterioration, a maximum of once per year until they develop either JF or SF wet AMD					
Patients treated for EF wet AMD may receive subsequent PDT. No further treatment costs are assumed for individuals with JF or SF PC wet AMD	Patients treated for early ARM, dry AMD or EF wet AMD may receive subsequent treatments. No further treatment costs are assumed for individuals with JF or SF wet AMD					
Individuals whose vision deteriorates to 6/60 or worse incur es	timated costs associated with blindness and rapidly					

It was assumed that an AMD screening programme would be administered at a national level as there is no need to coordinate screening invitations and screening test resources (as the test is a self-test). Drawing on the economies of scale argument, the cost of administration of a screen reminder was assumed to be between £0.50 and £2 per invitation sent.

Regardless of whether a case is screen detected, all individuals who seek medical care owing to a defect in vision that is due to either early ARM, dry AMD or wet AMD are assumed to present initially at an optometrist's. The Royal College of Ophthalmologists' guidelines recommend that mild, early AMD requires no special management and can be managed in the community.⁶⁷ Optometrists are assumed to continue to carry out routine examinations and refraction in this group of patients upon vision deterioration until dry AMD or wet AMD is suspected.

All individuals with dry AMD or wet AMD are referred on to a hospital eye service to confirm diagnosis via a fluorescein angiography and an outpatient appointment with an ophthalmologist. It is recognised that some individuals may present via their GP, but the cost impact of the different pathways to the hospital eye service is likely to be small and so a single pathway is described.

The unit costs informing the diagnosis and surveillance of ARM are presented in *Table 51*. The cost of the initial presentation at an optometrist's is estimated from the perspective of the NHS, and is informed by the published value of vouchers available to help with the costs of optical care. The proportion of individuals who are eligible for such help is unknown, and so it is assumed that 50% of the individuals presenting with vision defects due to ARM will be eligible. The costs of a first outpatient visit plus an angiography are applied at the point of diagnosis, whereas the costs of a follow-up outpatient visit plus angiography are applied at subsequent points of surveillance.

Individuals with bilateral ARM follow the pathway described for the eye with the most intensive intervention (e.g. a pathway leading to treatment). Individuals with no lesion(s) eligible for treatment are assumed to receive no further care until vision in the affected eye(s) worsens, which may or may not be due to the lesion type or location varying. It is recognised that vision may worsen without the lesion type or location varying. At the point at which vision declines, the individual is assumed to attend the hospital eye service and receive a further fluorescein angiography to identify the current lesion type and location.

Individuals are assumed to continue to attend the hospital eye service for surveillance upon worsening VA until either treatment for JF or SF wet AMD is instigated or the individual's better eye VA declines to 6/60, at which point treatment is assumed to be ineffective and costs associated with blindness and rapidly deteriorating vision are applied to individuals (see the section 'Costs associated with blindness and rapidly deteriorating vision', p. 50). In the absence of treatment initiation, it is assumed that individuals present only once per annum at a hospital eye service.

Treatment costs

Costs for treatments included in the base case analysis of the model are described in the subsequent section, followed by a section describing cost estimates for the potential interventions that are included in the sensitivity analyses of the AMD screening model.

Current treatment options

In the base case model analysis, only two treatment options are included: laser photocoagulation and PDT. The estimated costs of

irveillance costs	and	Diagnosis	TABLE 51
irveillance cost	and	Diagnosis	TABLE 51

Resource	Mean cost (£)	Interquartile range (£)	Data source
Optometrist visit	18.39		Department of Health ^a
Angiography	142.64	82.12–166.91	2002 Reference Costs ^b
First outpatient visit	92.11	71.39–108.85	2004 Reference Costs ^b
Follow-up outpatient visit	56.92	46.10-63.77	2004 Reference Costs ^a

^a HC12 Charges and optical voucher values, I April 2005: sight tests and NHS vouchers for glasses and contact lenses. 50% of attendees are assumed to be eligible for help with health costs.

^b All NHS reference cost sources are uprated to January 2006 values using the NHS Pay and Prices Index.

PDT are based on the reported frequency of therapy observed in the pivotal trials of PDT: the TAP and VIP trials. A mean cost per patient on PDT of £8052 (discounted at 3.5%) is estimated based on the data reported in *Tables 52* and *53*, which describe the proportion of patients receiving alternative numbers of PDT sessions. For each PDT session, the costs comprise the costs of verteporfin, laser treatment of the retina, an angiography and a follow-up outpatient visit. Each patient is assumed to attend for one additional angiography and outpatient visit after finishing PDT. The aggregate cost is applied at the point at which PDT commences and is assumed to cover all subsequent healthcare received by an individual with respect to the treated eye until best eye VA declines to 6/60. Individuals may receive subsequent treatment for the fellow eye.

Unit costs for resources associated with laser photocoagulation included the costs of angiography, outpatient visits and laser treatment of the retina. The frequency of intervention was informed by two studies. Busbee and colleagues estimated the cost–utility of laser photocoagulation for EF wet AMD.⁶⁸ They describe the process of undertaking additional angiographies 2–4 and 4–6 weeks after initial treatment, in addition to

TABLE 52	PDT	treatment	frequency	and costs
----------	-----	-----------	-----------	-----------

No. of treatments within 2 years	Proportion	Cost (£)	Discounted (£)
I	0.045	1349	1349
2	0.072	2497	2497
3	0.102	3646	3646
4	0.085	4795	4795
5	0.144	5944	5743
6	0.127	7093	6853
7	0.137	8242	7963
8	0.288	9391	9074
Mean no. of treatments beyond 2 years	1.52	1746	1630
Total cost			8052

TABLE 53 Unit costs associated with the provision of PDT

Resource	Mean cost (£)	Interquartile range (£)	Data source
Verteporfin Laser treatment of retina Angiography (fluorescein or indocyanine)	850.00 99.39 142.64	73.06–130.65 82 12–166 91	BNF 50 2002 Reference Costs ^a 2002 Reference Costs ^a
Follow-up outpatient visit	56.92	46.10–63.77	2004 Reference Costs ^a

^a All NHS reference cost sources are uprated to January 2006 values using the NHS Pay and Prices Index.

assuming that 25% of patients require re-treatment, incorporating three additional angiographies and outpatient visits (in addition to the extra laser treatment). Bonastre and colleagues present a burden of illness analysis for AMD in France, and present observed data on 105 patients, 76 of whom received laser photocoagulation.⁶⁹ The mean number of sessions for treated patients is stated to be 2.2. The data from these two studies are combined to estimate an aggregate mean cost of laser photocoagulation of £1536, as presented in *Table 54*.

Potential treatment options

A range of potential treatment options were included in the sensitivity analyses of the screening model. Treatment costs for interventions included in the additional model analyses are listed in *Table 55*. The data and assumptions informing the cost estimates are described in this section. For early ARM, vitamin supplements are assumed to be bought by individuals upon advice from the optometrist and do not impose a cost on the NHS.

The cost of anecortave acetate (Retaane depot) is unknown, although Sharma and colleagues⁷⁰ analysed the same intervention in the context of a new treatment for wet AMD in order to determine the cost at which this treatment might offer economic value to society, using incremental costeffectiveness ratios (ICERs). They found that an

TABLE 54	Costs of	laser	photocoagulation
----------	----------	-------	------------------

Costs per session	Cost (£)						
3 imes angiography	428						
3 imes follow-up outpatient visit	171						
I imes laser treatment	99						
Total	698						
Mean number of sessions	2.2						
Total cost per patient	1536						
^{<i>a</i>} It is assumed all treatment sessions occur within 1 year of initial treatment.							

Treatment	Aggregate treatment cost (£) ^a	Data source
Vitamin supplements for early ARM	0	Individuals' self-supply
Retaane depot (for dry AMD)	5,100	Threshold analysis (Sharma et al. ⁷⁰)
RHEOVision treatment (for dry AMD)	5,100	Assumed to be same as Retaane depot
PDT + triamcinolone	1,579	Augustin and Schmidt-Erfurth ⁵⁸
Generic anti-VEGF therapy	10,148	Pegaptanib US cost
Bevacizumab anti-VEGF therapy	1,358	Pollack ⁷²

TABLE 55 Medication costs for potential interventions for AMD evaluated in the additional sensitivity analyses

ICER of US\$100,000 per QALY would be associated with an anecortave cost of US\$3022 per vial, and an ICER of US\$50,000 per QALY with an anecortave cost of US\$2986 per vial. A cost of US\$3000 (£1700) per 15 mg is assumed in the screening model.⁷⁰ No costs for RHEOVision treatment are presented, so a similar cost profile to that estimated for anecortave acetate is used (£1700 per 6 months). No data were identified that could inform the likely number of treatment sessions, so a range of 2–4 sessions was assumed for both treatment options for dry AMD (with a mean of three sessions).

Triamcinolone is given in 4-mg doses as injections (aqueous suspension) for which the medication cost is £1.02 (Adcortyl, triamcinolone acetonide 10 mg/ml; net price of a 1-ml ampoule = \pounds 1.02). The cost per session of providing PDT + triamcinolone is estimated as the cost of a PDT session, as described above, plus the cost of the additional medication. In a case series of 199 patients treated with verteporfin PDT combined with intravitreal triamcinolone for CNV, patients required a mean of 1.25 treatments to achieve persistent inactivation of the neovascular membrane.⁷¹ All treated patients are also assumed to receive an additional angiography and outpatient visit following their last treatment session.

Regarding the costs of anti-VEGF therapies, no costs for the provision of ranubizimab have been published; whereas some have predicted a cost of US\$4000 per dose,⁷³ others suggest that the cost will be similar to that of pegaptanib, which costs around US\$1000 per dose (but is given every 6 weeks, rather than every 4 weeks for ranubizimab).⁷² The cost of a generic anti-VEGF therapy, based on the reported costs of pegaptanib, assumes a drug cost of £600 per injection, plus the cost of a follow-up outpatient visit per 6-weekly treatment application.

Only very limited information describing the number of ranubizimab injections that patients

may receive is included in the criteria for the SAILOR trial, which state that ranubizimab will be administered once per month for 3 months and thereafter as needed based on re-treatment criteria (which are unstated). Therefore, treatment frequency was also based on data from the pegaptanib studies, which reported that a mean number of treatments for the 2 years for all patients re-randomised to continue pegaptanib therapy was 16 out of 17 possible treatments.^{60,74}

Given the reported use of bevacizumab, and ongoing clinical studies evaluating bevacizumab as a potentially cheaper form of anti-VEGF therapy, which may have similar levels of effectiveness and frequency of intervention (although some doctors think that bevacizumab will require less frequent injections), but provided at a much lower cost, bevacizumab is not an inexpensive drug, costing US\$4400 per month when used as an intervention for colorectal cancer, but these doses are being divided into tiny portions to be injected into the eye. The amount needed for each injection costs only about US\$30–100.⁷²

Costs associated with blindness and rapidly deteriorating vision

Costs to the NHS and local and central government associated with blindness and rapidly deteriorating vision were informed by the estimates presented in an HTA of PDT.⁷⁵ Uprating the presented aggregate costs to January 2006 using the NHS Pay and Prices Index, the estimated initial costs associated with blindness range from £52 to £295, and the annual costs range from £1325 to £16,804 (*Table 56*).

Utility values

The QoL literature review presented identified two separate groups that had assessed utility related to diminished vision. Colleagues at ScHARR undertook a study to obtain general

	Base case (£)		Low range (£)		High range (£)	
	Proportion	Annual	Proportion	Annual	Proportion	Annual
Blind registration	0.945	97	0.5	40	0.945	170
Low vision aids	0.33	136	0.33	56	0.74	136
Low vision rehabilitation	0.11	205	0.11	125	0.11	309
Housing/council tax benefit	0.45	2,714	0.21	2,413	0.73	3,588
Social security	0.63	1,924	0.17	0	0.63	2,876
Tax allowance	0.05	319	0.05	145	0.18	319
Depression	0.386	392	0.06	392	0.5	392
Hip replacement	0.05	3,669	0.005	1,177	0.247	3,933
Community care	0.06	2,849	0.06	1,138	0.4	4,759
Residential care ^a	0.3	11,133	0.13	5,490	0.56	16,509
Initial cost ^b	160 (170)	52 (56)		295 (315)	
Annual cost ^b	6,295 (5,7Í9)	I,325 (14 ⁽ 15)	16,804 (I	17,937)

TABLE 56 Costs associated with blindness and rapidly deteriorating vision⁷⁵

^a It is assumed that 30% of individuals requiring residential care fund themselves and do not contribute costs from the perspective of the government.

^b Figures in parentheses are uprated to January 2006.

population utility values for VA and contrast sensitivity (CS) states.⁷⁶ A total of 209 patients with unilateral or bilateral AMD were recruited and interviewed. The study involved the completion of a number of visual tests to measure near and distant VA and CS. To estimate utility values, the respondents completed three preference-based measures, the Health Utility Index 3 (HUI-3), EuroQol 5 Dimensions (EQ-5D) and Short Form 6 Dimensions (SF-6D), in addition to estimating utility values via a time trade-off (TTO) process.

This study actually showed that CS had a slightly better relationship and explained more variation in health status than VA. However, the other data used in the model did not measure CS, hence it was not possible to develop a model based on CS progression and so the model is restricted to using VA-related utility data. *Table 57* presents the summary data describing utility values as a function of alternative levels of VA for each of the valuation approaches.

In the Sheffield study, a marked correlation with age was identified, which led to the estimation of an algorithm to predict utility values as a function of both age and VA. The algorithm was based on the HUI-3 results as general population values were considered most relevant [in line with recommendations by the National Institute for Health and Clinical Excellence (NICE)]. Regression analyses of the 209 patient responses tested a range of alternative models. Restricting the observations to the range 0–1.6 logMAR units (as used in the model), a linear model had the highest R^2 and produced the best predictions. The algorithm parameters are presented in *Table 58*.

TABLE 57	Mean	(standard	deviation)	scores o	f bre	ference-	based	measures	bv VA	in the	e hest	seeino	eve: Sl	heffie	ld stud	v ⁷⁶
IADLE J/	meun	Scunduru	deviation	300163 0	י אין	leience-	Duseu	meusures	רא ער		Dest	Secilia	Cyc. Ji	lellie	iu stuu	y

Distant logMAR	N	тто	HUI-3	SF-6D	EQ-5D
>2.00	16	0.47 (0.31)	0.10 (0.18)	0.63 (0.10)	0.63 (0.22)
1.31-2.00	60	0.60 (0.33)	0.27 (0.24)	0.65 (0.11)	0.71 (0.21)
0.61-1.30	59	0.64 (0.30)	0.36 (0.25)	0.66 (0.14)	0.75 (0.20)
0.31-0.60	41	0.67 (0.31)	0.38 (0.25)	0.67 (0.14)	0.70 (0.20)
≤0.30	33	0.73 (0.30)	0.50 (0.35)	0.70 (0.18)	0.75 (0.27)
η ²		0.04 ^b	0.13 ^{<i>a</i>,<i>b</i>}	0.02	0.02

 η^2 : variability in health status score explained by VA, and calculated as the sum of squares between groups divided by the total sum of squares from the analysis of variance results.

^{*a*} *p* < 0.05 between groups.

b p < 0.05 linear trend.

Parameter	Coefficient
Age	-0.00792
LogMAR units (0–1.6)	-0.10872
Constant	1.078315

An example estimate of utility is presented below for an 85-year-old with a better eye logMAR score of 0.2:

utility value = $1.0783 - (0.0079 \times 85) - (0.2 \times 0.10872) = 0.384$

Brown and colleagues have published a range of AMD utility studies.^{77,78} The most relevant study that was identified used the TTO approach to elicit utility values for AMD patients' current health state.⁷⁷ The results are presented in *Table 59*.

Although the Brown study used a slightly different grouping for VA compared with that used by the Sheffield group, the Brown estimates indicate a slightly stronger relationship between VA levels and utility than the Sheffield study, that is, the difference between the best and worst VA categories is greater. However, the differences between the two sets of TTO results are unlikely to be statistically significant, even though there were differences in the applied TTO methodology. Brown and colleagues used an open-ended variant of TTO where respondents are asked how long they are expected to live and then asked how much of the remaining time of life they would be willing to trade for a treatment that would restore TABLE 59 Non-age-specific AMD utility values by VA⁷⁸

Vision	Mean utility	95% CI
6/6–6/7.5	0.89	0.82 to 0.96
6/10-6/15	0.81	0.73 to 0.89
6/20-6/30	0.57	0.47 to 0.67
6/60–3/60	0.52	0.38 to 0.66
Count fingers to light perception	0.4	0.29 to 0.5

vision. The Sheffield study used the interactive variant developed at York University, and which was subsequently used to value the EQ-5D.⁷⁶

The model was analysed using two separate sets of utility values. The first set was based on the reported Sheffield algorithm, which adjusted the aggregate utility values for co-morbidities and other effects related to ageing. This utility algorithm assigns a relatively small utility effect to changes in VA; for example, a person with a better eye logMAR score of 0 (equivalent to 6/6 vision) has only 0.109 better utility than a person with a logMAR score of 1 (equivalent to 6/60 vision, and able to register as blind). The equivalent utility difference in the Brown study is around 0.3 utility points.

The model was also analysed using a set of utility values that were not adjusted for age effects, and so the inputted utility values are constant across all age groups. To provide the greatest contrast with the age-specific utility values, the values presented by Brown and colleagues are used. These values have also previously been used to inform the NICE appraisal of PDT for AMD.⁷⁹

Chapter 6

Model implementation and analysis

The evaluative framework for the evaluation of L screening programmes for AMD is a hybrid individual sampling/cohort Markov decision model, which is designed to reduce the number of iterations that would be required if the model was built as a pure individual sampling model. Given the complexity of the pathways described, implementation of a pure cohort model was infeasible. Individual sampling models analyse multiple first-order iterations of the model to estimate mean values for each of the model's outputs for a given set of input parameters. Each first-order iteration of the model describes the progression of disease in an individual person from age 50 to 100 years. For every model iteration, and every defined natural history, the model also describes the progression of disease assuming that individuals are clinically diagnosed in each year subsequent to the age at which ARM is the first incident; for example, if an individual first develops early ARM at age 70 years, the model describes separate 32 lifetime profiles for that individual to represent the effects of that individual being diagnosed with ARM at ages 70, 71, 72, ..., 98, 99, 100 years, or ARM never being diagnosed. The costs and effects estimated for each of the separate lifetime profiles are multiplied by the probability that an individual presents at each of the ages (or that the individual never presents). The probabilities of clinical diagnosis are dependent on the lesion(s) state(s), whether one or both eyes are affected and whether a screening programme is in place.

Multiple second-order analyses of the model are required to estimate the mean cost-effectiveness of alternative screening programmes for AMD, and also to represent the uncertainty around the mean cost-effectiveness estimates. Mean values for the model's input parameters were not specified because there was great uncertainty around the value of many of the parameters, and especially concerning the correlation between alternative input parameters. It is particularly important to minimise the likelihood of sampling infeasible combinations of values for alternative input parameters. The impact of infeasible correlations is likely to be greater in screening models than treatment models, because the range of input parameters is greater. An example involves the

sampling of high values for both disease incidence and disease progression, which can lead to extremely unlikely model outputs. Therefore, the second-order analysis of the model was informed by a process of calibration, which is described in the following section.

Calibration methods

Almost all natural history-based screening models include unobservable parameters, such as the probability of clinical presentation from a given stage in the natural history of a disease. Many screening models use calibration methods to populate baseline (mean) values for unobservable parameters, although very few use calibration to represent parameter uncertainty. The AMD screening model is calibrated to inform sets of correlated input parameters that are used to estimate mean cost-effectiveness values, in addition to informing probabilistic sensitivity analyses that represent the sum of the uncertainty in the model.

The general calibration process involves analysing the model for a large number of second-order iterations, from which the model predicts a set of outputs for alternative sets of input parameter values. The outputs include parameters that can be compared with data that have been empirically observed, such that the accuracy of each set of input parameter values in predicting alternative model outputs can be assessed quantitatively. The accuracy of each iteration's predictions for the defined outputs is represented as a weight. If multiple outputs are predicted, the sum of the weights across all the outputs is used to describe the likelihood that each input parameter set is the most accurate set of input parameters. For each iteration of the second-order analysis of the model, a set of input parameters is sampled according to the weights attached to each input parameter set as part of the calibration process.

Calibration outputs

Three types of observed data were used to calibrate the AMD screening model:

- age- and state-specific clinical diagnosis rates of ARM
- VA at presentation by ARM state
- age-specific rates of bilateral 6/60 vision or worse due to ARM.

Age- and state-specific clinical diagnosis rates of ARM

Age- and state-specific clinical diagnosis rates of ARM incorporate the incidence and the natural history of ARM, in addition to clinical presentation rates. No routine data or published studies of clinical diagnosis rates of ARM were identified, so it was necessary to pursue new primary sources. To inform the current analysis, data were made available from the ophthalmology department at the Sheffield Teaching Hospitals NHS Foundation Trust that described the number of patients receiving treatment for wet AMD with PDT in the years 2002–5, by age.

The analysis of the treated patients was restricted to patients living at postcodes defined as being within the city of Sheffield, so excluding patients who were referred to the Sheffield PDT clinic from surrounding towns and cities, including Doncaster, Barnsley and Chesterfield. The analysis was so restricted in order to be able to define accurately the size of the relevant age-specific population from which the patents were drawn, but also because one of the authors (CB) advised that the referred patients from the Sheffield area were more likely to represent the full set of referred patients (e.g. patients from Chesterfield may be referred elsewhere).

The estimation of the total number of patients presenting clinically with ARM and the distribution of ARM states (in both eyes) at presentation required back-extrapolation of the PDT data. The first stage involved estimating the total number of patients who presented at the PDT clinic, because the data only described those patients who received PDT. The consultant ophthalmologist author (CB) was the sole clinician seeing attending patients at the Sheffield PDT clinic, and so his estimate of the percentage of presenting patients who received treatment was used to extrapolate to the total number of presenting patients. He estimated that between 33 and 40% of referred patients were treated, so a mean estimate of 36.5% was used to estimate the total number of presenting patients at the PDT clinic.

The data from 2004 and 2005 were used, as these were the most current estimates, and the number

of treated patients increased year on year. The 2005 figures are uprated by 12/11 as the data describe numbers treated to November 2005.

The second step in the back-extrapolation required estimation of the proportions of patients presenting with ARM at general hospital eye services who were referred on to the Sheffield PDT clinic. This stage of the analysis was informed by a survey of all ophthalmologists who referred into the Sheffield PDT clinic. The ophthalmologists were asked the following questions:

- Of 100 patients receiving their first diagnosis of AMD or ARM, please estimate in your opinion how many are in each of the following AMD categories.
- Of 100 patients that you diagnose in the different categories of ARM, how many do you refer on to a PDT clinic? (Appendix 4).

Nine responses were received (a response rate of 31%), although three ophthalmologists stated that they could not provide responses without collecting data. The six sets of quantitative responses are presented in *Table 60*, which shows a very large range in the answers to both parts of the survey. As question 2 informs the total number of presented cases, it was decided to use the nonextreme responses to question 2 (respondents 2, 5 and 6) to inform the predicted number of individuals clinically presenting with ARM, as the extreme responses (both high and low) were unlikely to be representative of the full set of referring ophthalmologists. As the distribution of ARM states is likely to be correlated with the proportion of cases referred on to the PDT clinic, and a range of distributions were described by respondents 2, 5 and 6, the combined responses to questions 1 and 2 from these ophthalmologists were used to back-extrapolate the total number of presenting ARM patients and the corresponding distribution of lesion types.

The data presented in *Table 60* in answer to question 2 (for respondents 2, 5 and 6) were combined with the estimated numbers of patients presenting at the PDT clinic (*Table 61*) to estimate the age-specific numbers of patients presenting with some form of ARM in Sheffield. *Table 61* presents the resulting numbers showing that the largest group of patients is expected in the 75–84 years age group.

The responses to question 1 in *Table 60* were used to estimate ranges for the proportion of individuals within different age categories that

Question Respondent							
		I	2	3	4	5	6
I	Of 100 patients receiving their first diagnosis of AMD o many are in each of the following categories:	r ARM, ple	ase est	imate in	your o	pinion l	now
	Unilateral ARM	30	10	10	5	2	25
	Unilateral dry AMD	27	20	9	10	2	12
	Unilateral EF AMD	4	5	4	I	0	2
	Unilateral JF AMD	6	10	5	I	I	5
	Unilateral SF AMD	3	20	10	10	I	5
	Bilateral ARM	70	10	90	50	98	75
	Bilateral ARM/AMD, worst eye has dry AMD	50	10	10	5	82	40
	Bilateral ARM/AMD, worst eye has EF AMD	10	5	10	5	2	I
	Bilateral ARM/AMD, worst eye has JF AMD	5	5	70	I	4	5
	Bilateral ARM/AMD, worst eye has SF AMD	5	5	85	2	94	30
2	Of 100 patients that you diagnose in the following cates	gories, how	/ many	do you i	refer or	n to a P	DT
	clinic?:						
	Unilateral JF AMD	3	25	100	1	50	50
	Unilateral SF AMD	3	25	100	2	80	50
	Bilateral ARM/AMD, worst eye has JF AMD	5	30	100	2	60	50
	Bilateral ARM/AMD, worst eye has SF AMD	5	20	100	5	80	25

TABLE 60 Survey of ophthalmologists: responses to questions about patients referred on to PDT clinic in Sheffield

TABLE 61 Estimated age-specific clinical incidence of ARM in Sheffield

Age category (years)	Population at risk	Estimated cases presenting at PDT clinic	Cases present hospital e	ting at general eye service
			Minimum	Maximum
50–59	120,553	11	19	133
60–64	50,362	12	20	142
65–74	85,868	42	72	515
75–84	60,966	138	235	I,684
85–89	14,412	32	54	388
90+	6,866	18	30	215
Total	339,027	253	429	3,077

present with different combinations of ARM states in both eyes (including no ARM in one eye). The resulting range of presenting patients for each age and state combination are presented in *Table 62*.

Visual acuity at presentation by ARM state

The second output parameter used to calibrate the AMD screening model describes the VA of presenting eyes at the point of clinical presentation. A dataset containing the morphometric status of eyes and the corresponding VA of 99 eyes with wet AMD at the time of initial presentation was made available. The dataset and the methods used to examine the eyes are described in detail by Ali and colleagues.⁴⁰ *Table 63* describes the frequencies of the alternative lesion types and locations. Three separate regression analyses were undertaken to predict VA scores at presentation by lesion location. Covariates for occult and MC lesions were tested, although they were not statistically significant in any of the analyses. Based on the simple analysis, *Table 64* presents the 95% CIs for three lesion locations that were used in the calibration analysis.

Age-specific rates of bilateral 6/60 vision or worse due to ARM

The final output parameters against which the AMD screening model was calibrated describe the age-specific rates of visual impairment due to AMD in the UK. Evans and Wormald in 1996⁸⁰ looked at the incidence of those registered blind or partially sighted due to AMD over a period of 50 years. Over 15,000 people are registered

ARM state	Age category (years)					
	50–59	60–64	65–74	75–84	85–89	90+
Unilateral ARM	0–11	0-12	0–42	0-139	0–32	0–18
Unilateral dry AMD	0.4–22	0.4–23	1.6–85	5.2–277	1.2–64	0.7–35
Unilateral EF AMD	0–5.5	0–6	0–22	0–69	0–16	0–9
Unilateral JF AMD	0.2–11	0.2-12	0.8–42	2.6-139	0.6–32	0.3–18
Unilateral SF AMD	0.2–22	0.2–23	0.8–85	2.6–277	0.6–64	0.3–35
Bilateral ARM	0–11	0-12	0–42	0-139	0–32	0-18
Bilateral ARM/AMD, worst eye has dry AMD	11–20	12-21	42–75	139–246	32–57	19–31
Bilateral ARM/AMD, worst eye has EF AMD	0.4–5.5	0.5–6	1.7–21	5.5–69	1.3–16	0.7–9
Bilateral ARM/AMD, worst eye has JF AMD	0.9–5.5	0.9–6	3.4–21	. _69	2.6–16	1.4–9
Bilateral ARM/AMD, worst eye has SF AMD	5.5–21	6–22	21-80	69–261	16–60	9–33
Population at risk	120,553	50,362	85,868	60,966	14,412	6,866

TABLE 62 Estimated numbers of individuals presenting with AMD by bilateral ARM status in Sheffield

TABLE 63 Summary of ARM lesions by type and location in Ali and colleagues' dataset 40

Location	Occult	Ту	ре	Total
		мс	PC	
EF	2	2	4	8
JF	5	7	17	29
SF	14	13	35	62
Total	21	22	56	99

TABLE 64 95% CIs for wet AMD lesions at presentation (logMAR units)

Lesion	95%	95% CI			
	Lower	Upper			
EF	0.27	0.79			
JF	0.36	0.61			
SF	0.58	0.76			

annually due to this condition.⁸⁰ Owen and colleagues⁴ undertook a systematic review of AMD prevalence and visual loss caused by AMD, and noted that there was a dearth of population-based prevalence studies in the UK and that the numbers were too small to provide accurate estimates. Based on a meta-analysis of six studies (including one UK-based study), three outcome measures were estimated: partial sightedness due to AMD, prevalence of dry AMD and prevalence of wet AMD.

The model is used to predict age-specific rates of partial sightedness, defined as 6/60 vision or worse, which is assumed to be equivalent to the rates of 6/60 vision as presented by Owen and colleagues.⁴ *Table 65* presents the summary data

from Owen and colleagues⁴ regarding the percentage prevalence of individuals with best VA of between 6/60 and 3/60 and worse than 3/60. The reported data are combined to estimate rates of bilateral vision of 6/60 or worse in the age ranges specified in the AMD screening model.

Calibration analyses

To calibrate the AMD screening model, 2500 sets of the model's input parameters were randomly sampled from the probability distributions described for each input parameter. For each input parameter set, values for each of the calibration outputs described in the previous section were predicted, based on 200 first-order iterations of the model for each input parameter set. The number of first-order iterations describes the number of individuals flowing through the model for each second-order iteration, although the scale is multiplied up by the hybrid nature of the model, which incorporates multiple points of clinical presentation and subsequent treatment effect (as described in Chapter 2).

The predicted values of the calibration parameters were used to attach probabilities to each input parameter set, to reflect the probability that each set was the optimal set for predicting the calibration outputs. This process involved the following steps:

1. For each input parameter set, compare the predicted aggregate rate of clinical presentation with ARM with the (estimated) observed rate of clinical presentation. If the predicted value lies outside the observed range, the input parameter set is excluded from the main analysis. Otherwise, progress to step 2.

Age range (years)	N	Reported		Age range	Estimated	
		6/60–3/60	<3/60	(years)	<6/60	
50–54	1823	0 (0 to 0.2)	0 (0 to 0.2)	50–59	0.02 (0 to 0.12)	
55–59	2943	0 (0 to 0.13)	0.03 (0 to 0.19)	60–64	0.03 (0 to 0.19)	
60–64	3528	0 (0 to 0.1)	0 (0 to 0.1)	65–74	0.06 (0.02 to 0.14)	
65–69	3787	0 (0 to 0.1)	0 (0 to 0.1)	75–84	0.38 (0.21 to 0.63)	
70–74	3288	0.06 (0.01 to 0.22)	0.06 (0.01 to 0.22)	85–89	4.21 (2.72 to 6.2)	
75–79	2527	0.24 (0.09 to 0.52)	0.36 (0.16 to 0.68)	90+	8.67 (5.13 to 13.52)	
80–84	1422	0.84 (0.44 to 1.47)	0.84 (0.44 to 1.47)		, , , , , , , , , , , , , , , , , , ,	
85–89	570	0.88 (0.29 to 2.04)	3.33 (2.02 to 5.16)			
90+	196	2.55 (0.83 to 5.85)	6.12 (3.2 to 10.45)			

TABLE 65 Percentage of population with alternative levels of partial sight/blindness with 95% Cls in parentheses⁴

- 2. For each input parameter set, compare the predicted proportions of individuals presenting with ARM by age and bilateral ARM status to the observed ranges for each of the 60 output parameters described in Table 62. For each of the 60 age and state combinations, estimate the difference between the predicted value and the nearest boundary of the observed range. If the predicted value is within the range a difference of zero is estimated; if the predicted value is above the range, the difference is the predicted value minus the upper limit of the range; and if the predicted value is below the range, the difference is the lower limit of the range minus the predicted value. Sum the value of the differences across the 60 output parameters.
- 3. For each input parameter set, compare the predicted VA scores at presentation with the observed ranges for each of the three defined wet AMD states. For each parameter, estimate the difference between the predicted value and the nearest boundary of the observed range. Sum the value of the differences across the three output parameters.
- 4. For each input parameter set, compare the rates of bilateral visual acuity of 6/60 or worse with the observed ranges for each of the six defined age ranges reported in *Table 65*. For each parameter, estimate the difference between the predicted value and the nearest boundary of the observed range. Sum the value of the differences across the six output parameters.
- 5. For each input parameter set, calculate a separate weight for each of the three categories of calibration output parameters. The weights are estimated as the reciprocal of the sum of the differences for each of the three calibration outputs. For clinical incidence, if the sum of the differences across all parameters describing age- and state-specific clinical presentation rates is 0.025, a weight of 40 is estimated. If the

sum of the differences is zero, a subjective weight of 1000 is applied.

- 6. For each of the three categories of calibration output parameters, sum the weights across all included input parameter sets.
- 7. For each input parameter set, calculate the probability that each set is the optimal set by dividing the weight for each input parameter set by the sum of the weights for each of the three output parameter categories.
- 8. For each input parameter set, calculate a weighted aggregate probability that each set is the optimal set across all calibration output parameters by combining the estimated probabilities according to the following weights:
 - (a) age- and state-specific clinical diagnosis rates of ARM: 0.6
 - (b) VA at presentation by ARM state: 0.15
 - (c) age-specific rates of bilateral 6/60 vision or worse due to ARM: 0.25.

The weights applied to the three output parameter categories are subjectively defined, and reflect the relative importance of model predictions of clinical presentation rates.

9. Order the input parameter sets and estimate the cumulative probability that the sets prior to and including each set are the optimal sets. In the main analysis of the model, input parameter sets are selected by randomly sampling a value between 0 and 1 and selecting the input parameter set with the cumulative probability that is nearest to the sampled value.

Calibration results

The results of the calibration analysis showed that of the 2500 second-order iterations, 716 iterations estimated aggregate incidence rates outside the observed range: these parameter sets were excluded from further analyses. The following paragraphs and figures provide an indication of the success of the calibration process in predicting the observed data.

Of the remaining 1784 iterations, *Figure* 7 displays the distribution of the sum of the differences between the predicted and observed logMAR scores for three locations of wet AMD: EF, JF and SF lesions. The distribution is more skewed than that for the sum of differences of incidence rates, with over 40% of iterations reporting a maximum of sum of differences of 0.2 logMAR unit. Over 90% of iterations reported a sum of differences across the three lesion locations of less than 0.6 logMAR unit.

The other calibration parameter describes the sum of the age-specific incidence rates of individuals whose best eye VA decreases to at least 6/60 (equivalent to a logMAR score of 1). These data also show a skewed distribution. The differences appear fairly large, although it is noted that *Figure 8* presents sums of the differences across six independent age categories, and the aggregate difference is smaller than the sum. The data still show that over 90% of the iterations result in a sum of differences of less than 0.06.

Model analysis

Two main sets of analyses of the AMD screening model were undertaken, each of which included 2000 second-order iterations, which were informed by 200 first-order iterations. For each set of analyses, 400,000 individuals passed through the model. For each second-order iteration, a set of input parameters was sampled based on the weights estimated as part of the calibration. Separately sampled input parameter values for the screening, cost and utility values were also included. Based on the mean values for each of the 200 first-order iterations, the following model outputs were collected for each second-order iteration:

- treatment costs
- costs associated with blindness
- aggregate costs
- QALYs
- cases of blindness.

Sixteen screening options (including no screening) were evaluated, with the outputs collected for the following range of screening options:

- no screening
- screening every 1, 2, 3, 4 and 5 years
- initiating screening at ages 50, 60 and 70 years.

The outputs from these analyses inform mean estimates of costs, QALYs and numbers of cases of legal blindness for each of the defined treatment options, which were used to estimate relevant ICERs for both outcomes. The estimated ratios identified dominated and extendedly dominated screening options, which were excluded and the



FIGURE 7 Distribution of the sum of differences between the predicted and observed ranges for logMAR VA scores at clinical presentation



FIGURE 8 Distribution of the sum of differences between the predicted and observed ranges for age-specific rates of legal blindness

ratios re-estimated to provide an accurate assessment of the cost-effectiveness of screening.

Probabilistic sensitivity analyses were based on the observed costs and effects associated with each of the 2000 second-order iterations for each of the two analyses. The net benefits of each screening option in each iteration (including no screening), for a range of monetary values for the principal outcome measure (the QALY), are estimated to inform cost-effectiveness acceptability curves (CEACs). The CEACs inform the probability that each screening option is the most cost-effective at different monetary values of a QALY.

Future treatment options sensitivity analyses

Additional analyses of the 16 screening options were also undertaken for two additional treatment option scenarios. The chosen treatment option scenarios were based on the interventions described in Chapter 4. The first potential treatment option tested the impact of vitamin supplements that reduce the progression of early ARM to AMD, in addition to reducing the incidence of AMD in the fellow eye. This scenario also included the use of anecortave (Retaane) to reduce progression from dry AMD to wet AMD. The second potential treatment option tested the impact of the new anti-VEGF therapy for all JF and SF wet AMD lesions (plus PDT for predominantly classic lesions).

The hypothesis was that the availability of more preventative interventions would improve the costeffectiveness of screening (treatment option 1), whereas more effective interventions available at the latter stages of disease would reduce the costeffectiveness of screening.

Similar sets of analyses to those described for the main analyses were undertaken for the future treatment options, including 2000 second-order iterations, each of which was also informed by 200 first-order iterations. The same model outputs were collected and analysed to estimate mean costs and effects, in addition to a probabilistic sensitivity analysis.
Chapter 7 Results

s described in Chapter 6, the analysis of the ${f A}$ model comprises two broad sets of analysis. The reference case analysis is based on current treatment options offered to AMD patients in the NHS, which were informed by expert consultation with ophthalmologists. Two sets of utility values were analysed in conjunction with the reference case clinical and cost parameter values. The first reference case analysis used the non-age-specific values that have been presented by Brown and colleagues,⁷⁷ which have informed most previous economic evaluations of interventions for AMD. The second set of utility values was based on an algorithm estimated by the Sheffield AMD patient utility study,⁷⁶ based on responses to the Health Utilities Index (version 3). The regression analyses found age to be an important predictor of utility, and so the algorithm estimates utility as a function of age and VA.

A series of alternative analyses, based on a range of potential treatment option combinations, were also undertaken. The first section below describes the range of results derived from the reference case analyses. The subsequent section describes the results from the horizon scanning analyses that illustrate how the cost-effectiveness of screening for AMD may change with the advent of new treatment options.

Reference case results

The reference case analysis comprised 2000 iterations of the AMD screening model, each informed by a separate set of input parameter values. Two outcome measures were defined: QALYs and cases of blindness. The model outputs were analysed to estimate the incremental cost per QALY and per case of blindness prevented between alternative screening options arranged in increasing order of effectiveness, after excluding dominated (more costly and less effective) options and extendedly dominated options (the subsequent screening option has a lower incremental cost-effectiveness).⁸¹

The results of the reference case analysis using the non-age-specific utility values are presented in *Tables 66–69*. Only the screening options that were

non-dominated based on the mean estimates of cost-effectiveness are shown. Separate results are presented for the outcome measures, QALYs gained and cases of blindness prevented, and also for alternative costs of sending screening reminders to the eligible population at the relevant screening intervals, of £2 and of £3 per reminder.

The results presented in *Tables 66* and 67 show that the assumed cost of the screening programme has little impact on the incremental cost per QALY gained. The set of non-dominated screening options is similar, and based on the mean results an annual screening programme beginning at age 60 years is clearly the most costeffective option based on recognised thresholds for QALY gains.⁸²

Analyses based on the outcome 'number of cases of blindness prevented' do not vary greatly according to the cost of screening, although an additional non-dominated screening option (annual screening from age 70 years) is added to the core group of five non-dominated screening options when the cost per screen is ± 3 . Based solely on the mean results, there is an obvious step up in the cost-effectiveness ratios when annual screening from age 60 years is compared with annual screening from age 50 years, where the ICERs increase by an order of magnitude to around ± 1 million per additional case of blindness avoided.

The QALY results from the reference case analyses in which the age-specific utility values were used are presented in *Tables 70* and 71. These show that screening is predicted to be less cost-effective. The cost-effectiveness of the least effective (nondominated) screening option compared to no screening is between £14,878 and £19,606 depending on the assumed cost of screening. Using a threshold of £20,000 per QALY, screening from age 60 years with a 5-year interval is the most cost-effective option (based on the mean results).

In the probabilistic sensitivity analysis, ideally, credible intervals would be estimated for every possible screening option. However, alternative screening options may be dominated within different iterations of the model, and the ordering of the screening options may change by iteration. There were also comparisons that included some iterations in which costs and effects were higher for one screening programme, and other iterations in which costs and effects were higher for the other screening option (observations in the north-east and north-west quadrants). These factors make the analysis of results massively complex, if not infeasible. Therefore, 95% credible intervals describing the uncertainty around the mean cost-effectiveness results were estimated for each of the non-dominated screening options for which iterations were not observed in both the north-east and north-west quadrants.

The results for the non-age specific utility analyses, presented in *Tables 66–69*, show large levels of uncertainty across all analyses and all comparisons. The majority of comparisons demonstrate ranges of uncertainty that start with the more effective screening options (based on the mean results) dominating the less effective option, and end with the less effective option dominating the more effective option. Of the non-dominated upper intervals, the screening option with the lowest ratio is 4-yearly screening from age 60 years compared with no screening, although the incremental cost per QALY is still above £300,000.

Similar results are derived from the analyses informed by the age-specific utility values presented in *Tables 70* and 71. The majority of comparisons are accompanied by 95% credible intervals that stretch from the less effective option dominating to the more effective option dominating. The only option that is unlikely to be cost-effective is annual screening from age 50 years, for which the lower interval is at least £195,000.

The next stage in the uncertainty analysis involved the estimation of cost-effectiveness acceptability frontiers. Cost-effectiveness acceptability frontiers are derivatives of the more usually presented CEACs. Frontiers are required in cases in which the distribution of net benefits is skewed and the median and mean net benefits diverge. The frontier describes the probability that the screening option with the highest mean net benefits (for a given value of output, e.g. a QALY) is the most cost-effective option based on the proportion of iterations of the model in which that option displayed the highest net benefits.

Figures 9–12 present the cost-effectiveness acceptability frontiers for the four QALY-based

analyses (at screening costs of £2 and £3, for the age- and non-age-specific utility analyses). The frontiers do not vary greatly, and show that the no screening option has the highest net benefits at very low values of a QALY, and also has a high probability of being the most cost-effective screening option at these values. As the value attached to a QALY increases, and alternative screening options report the highest mean net benefits, the probability of the option with the highest mean net benefits being the most costeffective option decreases dramatically. In *Figure 9*, for example, at a value of £10,000 per QALY gained, 2-yearly screening from age 60 years is the option reporting the highest net benefits, but this option has the highest net benefits in only 2% of the 2000 model iterations. As the value of a QALY increases further, the probability of the screening option with the highest net benefits - annual screening from age 60 years - being the most cost-effective option increases to 32% at a value of a QALY of £100,000.

These results illustrate the high level of uncertainty around the mean estimates of costeffectiveness. The results also represent the fact that the differences in the number of QALYs gained and cases of blindness prevented between the screening options is very small, such that a relatively few iterations in which larger differences are observed have very significant effects on the mean cost-effectiveness. This effect is most noticeable with respect to the no screening option, which maintains the highest probability of being the most cost-effective option to a value of a QALY of £24,000, yet it only has the highest mean net benefits to a value of £2000.

The frontiers based on the age-specific utility values show that the no screening option produces the highest net benefits to a OALY value of between £15,000 and £19,000, depending on the cost of screening. There is also a relatively high probability that no screening is the most cost-effective option. Subsequently, up to a QALY value of between £70,000 and £90,000, the probability of the screening option with the highest net benefits being cost-effective is very low. After these thresholds, annual screening from age 60 years produces the highest net benefits. However, the no screening option remains the option with the highest probability of being cost-effective over all values of a QALY. At a value of £20,000, the probability of no screening being the most cost-effective option is over 50%.

Age at	Screening	Screening	Treatment	Blind	Aggregate	QALYs	Diffe	rences	Increm	iental cost per	QALY (£)
iirst screen (years)	Interval	costs (f)	costs (£)	costs (£)	costs (f)		Costs (£)	QALYs	Mean	2.5th percentile	97.5th percentile
No screening		0.00	70	710	780	16.49096					
60	5-yearly	4.55	124	656	784	16.49300	4.0	0.0020	1,970	Dominates	303,539
60	3-yearly	6.20	145	637	788	16.49369	3.7	0.0007	5,349	Dominates	Dominated
60	2-yearly	9.64	165	619	794	16.49428	5.7	0.0006	9,710	Dominates	Dominated
60	Annual	19.96	198	591	809	16.49530	15.5	0.0010	15,169	Dominates	757,786
50	Annual	36.84	661	590	826	16.49535	16.3	0.0000	345,252	68,700	Dominated

valu
tility
fic u
speci
age-
non-
£3 (
ð
screen
per s
cost
D
Вu
Ē
, assi
ð
¥
õ
р Б
nir
ree
SCI
q
lts
resu
Ę
ò
per
cost
ital
nen
'en
lnci
67
ш
ВЦ

Age at	Screening	Screening	Treatment	Blind	Aggregate	QALYs	Diffe	rences	Increm	iental cost per	QALY (£)
lirst screen (years)	Interval	costs (£)	costs (£)	costs (£)	costs (£)		Costs (£)	QALYs	Mean	2.5th percentile	97.5th percentile
Vo screening		0.00	70	710	780	16.49096					
0	5-yearly	6.83	124	656	787	16.49300	6.3	0.0020	3,085	Dominates	317,837
0	4-yearly	7.22	133	648	788	16.49330	4.1	0.0003	4,807	Dominates	Dominated
09	3-yearly	9.30	145	637	162	16.49369	3.1	0.0004	7,861	Dominates	Dominated
09	2-yearly	14.46	165	619	799	16.49428	7.4	0.0006	12,641	Dominates	Dominated
00	Annual	29.94	198	591	819	16.49530	20.6	0.0010	20,227	Dominates	836,724
00	Annual	55.26	661	590	844	16.49535	24.7	0.0000	524.511	115.326	Dominated

63

 $\textcircled{\sc c}$ Queen's Printer and Controller of HMSO 2008. All rights reserved.

(non-age-specific utility values)
, assuming a cost per screen of £2
ost per case of blindness prevented.
TABLE 68 Incremental co

64

of blindness	97.5th percentile		Dominated		ostly and less	ions ^a		Dominated	
al cost per case prevented (£)	2.5th percentile		844		Includes less c	effective iterat		Dominates	
Increment	Mean		1,155	2,489]	3,272	5,322	8,827]	819,112	
rences	Blind		0.0035	0.0005	0.0007	0.0011	0.0018	0.0000	
Diffe	Costs (£)		4.0	1.3	2.4	5.7	15.5	16.3	
Blind		0.04523	0.04175	0.04122	0.04049	0.03942	0.03767	0.03765	
Aggregate costs	(ד)	780	784	786	788	794	809	826	rrvals infeasible.
Blind costs	(F)	710	656	648	637	619	591	590	credible inte
Treatment costs	(F)	70	124	133	145	165	198	661	ne estimation of
Screening costs	(ד)	0.00	4.55	4.81	6.20	9.64	19.96	36.84	ations renders tl
Screening interval			5-yearly	4-yearly	3-yearly	2-yearly	Annual	Annual	s of such observ
Age at first screen	(years)	No screening	60	60	60	60	60	50	^a The presence

TABLE 69 Incremental cost per case of blindness prevented, assuming a cost per screen of £3 (non-age-specific utility values)

s	a)		τ			0			Β	
ie of blindnes ()	97.5th percentil		Dominated			costiy and less	ations		Dominated	
l cost per cas prevented (i	2.5th percentile		1,011				errective itera		Dominates	
Incrementa	Mean		1,809	ر 2,735 م	4,223	6,928	9,069	19,437 ا	1,244,407	
ences	Blind		0.0035	0.0005	0.0007	0.0011	0.0013	0.0005	0.0000	
Differ	Costs (£)		6.3	4 .	3.1	7.4	8.II	8.9	24.7	
Blind		0.04523	0.04175	0.04122	0.04049	0.03942	0.03812	0.03767	0.03765	
Aggregate costs	(E)	780	787	788	162	799	810	819	844	rvals infeasible.
Blind costs	(F)	710	656	648	637	619	608	591	590	credible inte
Treatment costs	(1)	70	124	133	145	165	189	198	661	ie estimation of
Screening costs	(F)	0.00	6.83	7.22	9.30	14.46	13.42	29.94	55.26	tions renders th
Screening interval			5-yearly	4-yearly	3-yearly	2-yearly	Annual	Annual	Annual	of such observa
Age at first screen	(years)	No screening	60	60	60	60	70	60	50	^d The presence

Age at	Screening	Screening	Treatment	Blind	Aggregate	QALYs	Differ	ences	Increme	ental cost per	QALY (£)
tirst screen (years)	Interval	costs (£)	costs (£)	costs (£)	costs (£)		Costs (£)	QALYs	Mean	2.5th percentile	97.5th percentile
No screening		0.00	73	698	171	10.21505					
60	5-yearly	4.55	129	644	778	10.21553	7.2	0.0005	14,878	Dominates	I,484,478
60	3-yearly	6.20	151	625	782	10.21569	4.I	0.0002	25,116	Dominates	Dominated
60	2-yearly	9.64	171	608	789	10.21583	7.0	0.0001	48,883	Dominates	Dominated
60	Annual	19.96	204	582	805	10.21607	16.5	0.0002	68,854	Dominates	2,898,130
50	Annual	36.84	204	580	82 I	10.21609	16.1	0.0000	I,260,676	195,119	Dominated

valu
ility
ìc ut
ecifi
ge-st
3 (ag
of £
een c
r scr
t þe
1 COS
ing c
mns
, as
É
Ł
g fo
enin
scre
s of 5
sults
Y re
QAL
þer
cost
ntal
ame
Incre
-
ш
3L
TA

Age at	Screening	Screening	Treatment	Blind	Aggregate	QALYs	Diffe	rences	Increm	ental cost per	QALY (£)
tirst screen (years)	Interval	costs (£)	costs (£)	costs (£)	costs (£)		Costs (£)	QALYs	Mean	2.5th percentile	97.5th percentile
No screening		0.00	73	698	177	10.21505					
60	5-yearly	6.83	129	644	780	10.21553	9.4	0.0005	19,606	Dominates	1,551,097
60	3-yearly	9.30	151	625	785	10.21569	4.9	0.0002	30,212	Dominates	Dominated
60	2-yearly	14.46	171	608	794	10.21583	8.7	0.0001	60,914	Dominates	Dominated
60	Annual	29.94	204	582	815	10.21607	21.6	0.0002	90,439	Dominates	3,440,834
50	Annual	55.26	204	580	840	10.21609	24.5	0.0000	1,922,621	285,602	Dominated







FIGURE 10 Cost-effectiveness acceptability frontier for the cost per QALY of screening for AMD, assuming a cost per screen of $\pounds 3$ (non-age-specific utility values)



FIGURE 11 Cost-effectiveness acceptability frontier for the cost per QALY of screening for AMD, assuming a cost per screen of £2 (age-specific utility values)



FIGURE 12 Cost-effectiveness acceptability frontier for the cost per QALY of screening for AMD, assuming a cost per screen of $\pounds 3$ (age-specific utility values)

Sensitivity analyses results

Two separate analyses of the AMD screening model were undertaken to estimate the impact of potential future interventions for different stages of ARM:

- Vitamin supplements that reduce the progression of early ARM to AMD, in addition to reducing the incidence of AMD in the fellow eye. This scenario also included the use of anecortare (Retaane) to reduce progression from dry AMD to wet AMD.
- The new anti-VEGF therapy for all JF and SF wet AMD lesions (plus PDT for PC lesions).

The following sections describe the incremental cost per QALY results for the two sensitivity analyses. The analyses incorporate only one set of utility values due to the running time requirements for the model. These analyses are largely illustrative, so the non-age specific utility values were chosen owing to the larger differences between the VA states, which in turn allows for more noticeable differences between the screening options.

Vitamin supplements for patients without dual eye wet AMD and anecortare (Retaane) for dry AMD

The sensitivity analysis of the AMD screening options assuming effective preventive

interventions that reduce progression to set AMD shows that these interventions do increase the cost-effectiveness of screening. *Tables* 72 and 73 present the results. Comparing these results with the reference case results presented in *Tables* 68 and 69, the set of non-dominated screening options is similar. The magnitude of the decrease in the ICERs is moderate. Assuming a screening cost of £2 per test, the cost-effectiveness of screening compared with no screening is £344 per QALY; in the reference case it was £1970. The ICER for annual screening from age 60 years, compared with 2-yearly screening from age 60 years, decreases from £15,169 in the reference case to £11,479.

The credible intervals for each comparison remain large, although the upper intervals for screening versus no screening approaches the $\pm 100,000$ level for a screening cost of ± 2 . Across the other comparisons, most stretch from dominates to dominated, although it appears unlikely that annual screening from age 50 years is a costeffective option.

The cost-effectiveness acceptability frontiers, presented in *Figures 13* and *14*, show similar patterns. No screening has the highest expected net benefits at QALY values approaching zero; 2-and 3-yearly screening from age 60 years is

TABLE 72 Incremental cost per QALY results of screening for AMD for the sensitivity analysis of preventative interventions, assuming a cost per screen of £2

Age at first	Screen	Aggregate	QALYs	Differ	ences	Incremer	ntal cost per	QALY (£)
screen (years)	interval	costs (£)		Costs (£)	QALYs	Mean	2.5th Cl	97.5th Cl
No screen		741.5	16.52080					
60	4-yearly	742.3	16.52321	0.8	0.0024	344	Dominates	122,584
60	3-yearly	745.2	16.52367	2.9	0.0005	6,384	Dominates	Dominated
60	2-yearly	750.8	16.52441	5.5	0.0007	7,471	Dominates	Dominated
60	Annual	764.6	16.52561	13.8	0.0012	11,479	Dominates	417,478
50	Annual	780.9	16.52562	16.4	0.0000	1,289,380	495,081	Dominated

TABLE 73 Incremental cost per QALY results of screening for AMD for the sensitivity analysis of preventative interventions, assuming a cost per screen of $\pounds 3$

Age at first	Screen	Aggregate	QALYs	Differ	ences	Incremen	ital cost per	QALY (£)
screen (years)	intervai	COSTS (L)		Costs (£)	QALYs	Mean	2.5th Cl	97.5th Cl
No screen		742	16.52080					
60	4-yearly	745	16.52321	3.2	0.0024	1,342	Dominates	130,905
60	3-yearly	748	16.52367	3.6	0.0005	7,910	Dominates	Dominated
60	2-yearly	756	16.52441	7.2	0.0007	9,797	Dominates	Dominated
60	Annual	775	16.52561	19.0	0.0012	15,775	Dominates	565,479
50	Annual	799	16.52562	24.8	0.0000	1,953,219	752,016	Dominated







FIGURE 14 Cost-effectiveness acceptability frontier for the cost per QALY for the sensitivity analysis of preventative interventions for wet AMD, assuming a cost per screen of $\pounds 3$

preferred at low QALY values, whereas annual screening from age 60 years remains the option with the highest net benefits at all other QALY values. The probability that annual screening is the most cost-effective option is relatively low, at only 20% at a QALY value of £30,000.

Anti-VEGF therapy for JF or SF wet AMD (plus PDT for predominantly classic lesions)

The sensitivity analysis of AMD screening given the use of an anti-VEGF therapy for JF or SF wet AMD lesions shows that screening becomes slightly less cost-effective than when analysing current treatment options (based on non-age-specific utility values). The mean results presented in *Tables 74* and 75 show that the same screening options are included in the final analysis and that the mean ICERs are slightly higher than in the reference case analysis. The credible intervals remain significantly wide for most comparisons, although the upper interval for annual screening from age 60 years is now less than £100,000, and the lower interval for annual screening from age 50 years is within acceptable thresholds.

The cost-effectiveness acceptability frontiers, presented in *Figures 15* and *16*, show that all six screening options are included. In both cases, annual screening from age 60 years has an

increasing probability of being cost-effective from a low value of a QALY gained, reaching around 45% at a value of a QALY of £50,000.

Summary

This chapter has presented the mean results, and the results of probabilistic sensitivity analyses, for a reference case specification of current treatment options, and also a range of potential treatment options that may be provided at some point in the future. The results show that, given the assumptions incorporated in the analyses, some form of screening may well be cost-effective, although there remains very large uncertainty around the presented estimates of costeffectiveness. Of the screening options analyses, annual screening from age 60 years appears to consistently be the most cost-effective option.

The assumptions informing the input values for these parameters were described in previous chapters, and the relationship between these assumptions and the observed results are discussed in Chapter 8, which interprets the results in terms of identifying the key drivers of the costeffectiveness results and the strength of the data and assumptions informing the key parameters.

TABLE 74 Incremental cost per QALY results of screening for AMD for the analysis of anti-VEGF therapy for wet AMD, assuming a cost per screen of £2

Age at first	Screen	Aggregate	QALYs	Differ	ences	Increme	ntal cost per	QALY (£)
screen (years)	interval	costs (£)		Costs (£)	QALYs	Mean	2.5th Cl	97.5th Cl
None		858	0	858	16.4765			
60	5-yearly	897	4.55	902	16.4841	5,773	Dominates	38,544
60	4-yearly	910	4.81	915	16.4852	12,306	Dominates	Dominated
60	3-yearly	931	6.20	938	16.4866	15,545	Dominates	Dominated
60	2-yearly	971	9.64	981	16.4889	19,410	Dominates	Dominated
60	Annual	1,053	19.96	1,073	16.4928	23,358	285	86,875
50	Annual	I,054	36.84	1,090	16.4930	141,476	11,254	Dominated

TABLE 75 Incremental cost per QALY results of screening for AMD for the analysis of anti-VEGF therapy for wet AMD, assuming a cost per screen of £3

Age at first	Screen	Aggregate	QALYs	Differ	ences	Increme	ntal cost per	QALY (£)
screen (years)	Interval	COSTS (L)		Costs (£)	QALYs	Mean	2.5th Cl	97.5th Cl
None		858	0.00	858	16.4765			
60	5-yearly	897	6.83	904	16.4841	6,070	Dominates	39,123
60	4-yearly	910	7.22	917	16.4852	12,429	Dominates	Dominated
60	3-yearly	931	9.30	941	16.4866	16,014	Dominates	Dominated
60	2-yearly	971	14.46	986	16.4889	20,183	Dominates	Dominated
60	Annual	1,053	29.94	1,083	16.4928	24,658	1,214	91,984
50	Annual	I,054	55.26	1,109	16.4930	212,113	16,235	Dominated







FIGURE 16 Cost-effectiveness acceptability frontier for the cost per QALY for the analysis of anti-VEGF therapy for wet AMD, assuming a cost per screen of $\pounds 3$

Chapter 8 Discussion

The results of the evaluation of screening for AMD presented in the previous chapter underline the *a priori* beliefs about the level of uncertainty around the mean cost-effectiveness results. The probabilistic sensitivity analyses that were undertaken for four alternative scenarios produced very wide credible intervals for the ICERs; most of the intervals ranged from one option costing less and being more effective to the other option costing less and being more effective.

The uncertainty around the utility values used in the model was well defined. Two relevant utility studies were identified, which reported very different values. In one study, 209 patients completed health status questionnaires including the Index of Visual Function (VF)-14 and three preference-based measures (HUI-3, EQ-5D and SF-6D) and the TTO.⁷⁶ Brown and colleagues⁷⁷ estimated utility values using the TTO method for a population of patients with actual visual loss due to AMD. The Sheffield study estimated lower utility values in the better VA categories (0.81 versus 0.89) and higher values in the worse VA groups (0.58 versus 0.40).

Rather than combine the studies, the reference case model was analysed separately using the two sets of utility values. Using the set that was more favourable to screening (i.e. that had a larger differential between levels of VA) suggested that annual screening from age 60 years provides the highest mean net benefits at commonly accepted ranges of the value of a QALY;⁸² the mean cost-effectiveness results indicate that this screening programme should be adopted. However, the upper 95% credible interval for these analyses showed incremental costs per QALY of over £0.75 million.

The analysis based on the less favourable (agespecific) utility values predicted that screening from 60 years was cost-effective, although the screening interval should be at least every 3 years, and more likely every 5 years. The upper 95% credible intervals for these analyses were above £1 million per QALY gained.

The presented cost-effectiveness acceptability frontiers show that the screening options with the

highest expected net benefits at different QALY values tend to have relatively low probabilities of being the most cost-effective option. In one of the reference case analyses, for example, annual screening from age 60 years had the highest net benefits from a QALY value of £20,000, but the probability of it being the most cost-effective option was less than 20% at a QALY value of £30,000. This result is to be expected given the large uncertainties in the analysis, combined with the fact that 15 alternative screening options are considered.

The two other scenario analyses incorporated the potential effect of treatments that may become available at some point in the future. These included treatments to prevent progression to wet AMD and the incidence of wet AMD in the fellow eye, and the availability of therapies that inhibit the VEGF (the anti-VEGFs) for the treatment of wet AMD. In the case of the anti-VEGF treatment option, the future is potentially very close as this class of treatments is currently being assessed by NICE.

The addition of a preventive treatment option for individuals detected with early-stage disease was found to have only a moderate effect on the cost-effectiveness of screening. The moderate impact on the cost-effectiveness of screening of potential preventive interventions that reduce progression rates to wet AMD is initially surprising because one might expect significant returns from interventions that are effective in preventing disease progression. However, the assumption in the reference case that first eyes are treated upon detection reduces the scope for further large benefits from preventive interventions. Screening is assumed to increase the detection of first eye wet AMD, such that a significant proportion of patients will receive treatment for wet AMD in both eyes. As utility is based solely on VA in the worst seeing eye, this means that screening provides two chances to maintain VA in at least one eye and thus maintain QoL. If treatment of the first eye was not considered appropriate (perhaps on the grounds of cost-effectiveness), then the impact of the preventive interventions would be expected to increase.

It is also worth noting that the preventive interventions are not 100% effective; the mean RR for vitamin supplements in preventing fellow eye incidence was 0.76.

The introduction of the new, more effective treatment options for wet AMD (the anti-VEGFs) had a smaller predicted effect on the mean estimates of the cost-effectiveness of screening. With the anti-VEGFs, the same non-dominated screening options were identified as in the reference case, although each mean ICER was slightly higher than in the reference case. These results were not surprising as the anti-VEGFs are significantly more expensive therapies, but also because this new therapy option improves vision in a proportion of patients, thus reducing the effects of late presentation in the absence of screening.

However, compared with the reference case, the probabilistic sensitivity analysis for the anti-VEGF analyses showed reduced credible intervals for screening compared with no screening, and for annual screening from age 60 years. This is most likely due to the increased certainty around the effectiveness of the anti-VEGFs relative to the certainty around the effectiveness of PDT. As the difference in QALYs between the alternative screening options is small, a minor change in the effectiveness of treatment leading to an even more minor change in QALYs can cause a relatively large change in the estimated ICER.

Unlike the other non-dominated screening options, the mean estimate of cost-effectiveness for annual screening from age 50 years compared with annual screening from age 60 years decreases (becomes more cost-effective) with an anti-VEGF treatment option. The credible interval also appears to widen as the lower bound shows screening from age 50 years to be potentially costeffective. Most of the few individuals detected and treated in their 50s would remain alive and receive treatment in their 60s if screening commences at age 60 years. They therefore incur similar costs. However, it seems that the relative effect of treatment with the anti-VEGFs in individuals aged 50–60 years compared with individuals aged over 60 years is greater than the relative effect of PDT.

It should also be noted that the assumptions made about the long-term effectiveness of all of the interventions included in the analysis were favourable; in particular, it was assumed that the observed effect during the treatment period (a maximum of 2 years for PDT and the anti-VEGFs) was maintained over the course of a patient's lifetime.

The remainder of this discussion chapter focuses on the interpretation of the results from the perspective of defining the major areas of uncertainty. At the outset of the evaluation, it was intended to undertake a full expected value of information (EVI) analysis of the AMD screening model to inform quantitatively priorities for further research in this disease area. Two main factors persuaded us against undertaking an EVI analysis. First, as will be detailed in subsequent sections, there were such levels of uncertainty about certain aspects of the AMD screening model structure and population that the derivation of quantitative estimates to inform further research may provide a misleadingly definitive representation of the value of additional research. Second, although there are significant shortcomings in the structure and population of the model, and an innovative approach was adopted to reduce the analytic burden (i.e. the hybrid cohort/individual sampling approach), it remains a complex model that requires significant computer running time to analyse. The sensitivity analyses of two potential treatment scenarios required eight Pentium 4 processor PCs running from Friday afternoon to Monday morning to generate 2000 second-order iterations, each comprising only 200 first-order iterations. EVI analyses, even using short-cut Gaussian process techniques,⁸³ require significantly larger numbers of model iterations that were not feasible within the time frame of the evaluation.

It was decided that a qualitative appraisal of the major areas of uncertainty around the potential cost-effectiveness of a screening programme for AMD would be a more useful approach. Although no specific analyses of the contribution of the effects of individual parameters were undertaken, the understanding of the AMD screening model that has been developed over the course of the evaluation is an important output from the project that should be disseminated to inform future work in this area.

Areas of uncertainty

The following sections discuss the impact of particular aspects of the model structure and the data and assumptions used to populate key parameters, on the results of the AMD screening model, and how the screening model may be improved with respect to these issues:

- model structure
- rates of clinical presentation
- screening test and optometrist effectiveness
- treatment effectiveness
- costs of blindness.

Model structure

Recent guidelines on the use of decision analytic modelling suggest that "[health] states should not be omitted because of lack of data",⁸⁴ a statement with which we agree. It is generally preferable to define a model structure based on the best available evidence regarding the disease process, disaggregating the disease process into states that best represent the differential effects of the alternative stages of the disease on costs and outcomes (i.e. survival and QoL).

The initial stages of the model development process for the AMD screening model were based on discussions with expert ophthalmologists, reading the literature and previous research undertaken by the research team. This process indicated that VA was not the preferred marker of visual deterioration, as utility studies had shown that changes in CS provided a more accurate reflection of the impact of visual deterioration on QoL.⁷⁶ McClure and colleagues developed a visual functioning index [daily living tasks dependent on vision (DLTV)] that demonstrated that specific levels of vision as measured by VA, reading index and CS corresponded with different perceived amounts of difficulty in the performance of daily living tasks.⁸⁵ Thus, VA is unlikely to represent all of the potential QoL effects of AMD.

Ideally, therefore, the model would describe the progression of AMD in terms of progression of a range of measures of vision, but if only one were to be picked it would be CS. However, very few of the data sources used to populate the model (either primary or secondary sources) presented data describing the CS of eyes with AMD. Although expert elicitation was an integral part of this study, it was not considered feasible to ask experts to estimate levels of CS as a function of AMD lesion and time since incidence, which would have been required. Therefore, the model was built around progression of VA, despite a preference for CS.

Recent evidence shows that intact autofluorescence at the macula in wet AMD correlates with VA, lesion size and symptom length, and that in such cases VA might be rescued if treatment could suppress neovascular growth without damaging the RPE/retina complex.⁸⁶ Further developments to the model could include the representation of lesion size and treatment effect by symptom length, and also incorporating autofluorescence imaging as a diagnostic test that influences treatment strategy.

Rates of clinical presentation and reasons for non-presentation

The estimated rates of clinical presentation of ARM, by stage, were possibly the most important set of input parameters for the AMD screening model as these rates informed the main part of the model calibration process. These data were informed by local data from the Sheffield PDT clinic, which were extrapolated backwards to estimate the aggregate numbers of individuals referred to general hospital eye services by AMD lesion type, using responses from a survey of general ophthalmologists who referred to the PDT clinic. The main problems with these data included the fact that the sample of patients referred to the PDT clinic was small, the PDT database was not validated and the survey of ophthalmologists and inconsistencies was limited in the responses received.

The 'observed' rates of clinical presentation were much lower than the estimated population incidence rates that were primarily based on data reported by the Rotterdam study.² This imbalance meant that the calibration process estimated relatively low values for the probability of clinical presentation by AMD lesion; for example, the mean probability of clinical presentation for patients with the worst combination of AMD states (SF wet AMD in both eyes) was 0.10. The data from the Rotterdam study used to inform population incidence rates are not ideal; for example, UK data with more frequent, shorter and more complete follow-up intervals would be preferred. However, these data are of reasonable quality relative to the data used to estimate clinical presentation rates.

Given the role of the clinical presentation data in calibrating the model, we believe that serious consideration should be given to setting up a pilot study to estimate rates of clinical presentation. It may be possible to collate routinely collected data at relatively low expense.

The estimated difference between population prevalence and clinical presentation led the model to predict significant potential for screening to detect and successfully treat individuals who otherwise would not have presented. The model did not differentiate between the utility effects of preventing VA loss in patients who would have presented clinically and those who would not have presented clinically. However, the accuracy of the model's results depends to a large degree on the explanation for why individuals do not present clinically (if that is the case). It may be that nonpresenters are accepting of fate, in which case they would be unlikely to present following screening, which would affect the estimated screening detection rates. Alternatively, they may have a higher threshold for presentation due to experiencing lower utility effects. In this case, less utility gain should be assigned to an improved VA profile. Finally, non-presenters may perceive a lack of treatment options for AMD, such that the utility gain may be assumed to be constant across all individuals with AMD.

Screening and optometrist effectiveness

The effectiveness of the screening test (a reminder letter to undertake a cover test) is linked to the effectiveness of optometrists in dealing with individuals who present as a result of receiving a reminder letter. No data were identified that described the likelihood of individuals with particular combinations of ARM in both eyes (including no ARM) presenting as a result of receiving a letter advising them to undertake a cover test. Data were elicited from experts that described the probability that persons with different forms of ARM, who undertook the test, would identify abnormalities that might lead to them presenting at an optometrist's. These estimates cover only one of three of the required probabilities (that individuals undertake the test, notice an abnormality and present at an optometrist's) but the other two could not be feasibly informed by ophthalmologists.

No data were identified that described how effective optometrists are in diagnosing and referring on individuals who present to them with AMD. The AMD screening model incorporates the assumption that patients who present as a result of screening and do not have ARM do not impose costs on the screening programme. The model also assumes that optometrists accurately refer all cases of dry and wet AMD on to hospital ophthalmologists, while not referring any cases of early ARM.

Data to inform these parameters would require the setting up of pilot sites in which screening for AMD might be administered to a defined population. The study would need to be carefully designed to minimise bias, as it would be necessary to follow up all individuals who attend optometrists as a result of the screening programme. Ideally, diagnostic tests would be undertaken on a sample of the population in order to estimate the test characteristics, although useful information could be obtained from an ageweighted comparison of presentation rates with literature-based prevalence estimates.

Treatment effectiveness

The reference case analysis included only two forms of treatment: laser photocoagulation for EF wet AMD and PDT for PC JF or SF wet AMD. Although good quality trials inform both interventions, the laser photocoagulation trial for EF lesions describes effectiveness out to 5 years, but similar data are only available to 2 years for PDT. The lack of long-term follow-up data inevitably requires the use of weak assumptions to extrapolate the observed effectiveness data.

A similar approach was used to inform the effectiveness of both interventions. It was assumed that the additional proportion of patients who lost fewer than six lines at 5 years (laser photocoagulation) or two lines at 2 years (PDT) in the intervention group over the placebo group maintained their VA at that level over the remainder of their lifetime in the treated eye. The choice of threshold for laser photocoagulation was enforced by the limited publication of further results, while the tighter PDT threshold was defined as the patient-level data from the pivotal trial were made available.

With respect to the assumed intervention for early ARM, a key assumption concerns the role of optometrists in promoting the use of vitamin supplements and the likelihood that individuals comply. The assumption made in the model was that all individuals presenting with early ARM selftreat with the recommended supplement. The effectiveness of zinc in preventing progression was informed by published ORs describing the likelihood of individuals with bilateral drusen (early ARM) or unilateral AMD (dry or wet) experiencing AMD in eyes without AMD at baseline. The issue around the representation of effectiveness of vitamin supplements is not particularly around the stated levels of effectiveness in patients taking the supplement, as appropriate CIs were defined. Rather, the model assumes that no-one takes such supplements until they are advised to do so by an optometrist, whereas a proportion of the population at risk of AMD is likely to be taking such supplements.

Data informing these parameters could be collected as part of a screening pilot study, as described in the previous section. The scope of the screening pilot study may be extended to incorporate rates of uptake of vitamin supplements and compliance as a result of initiation by an optometrist. Existing rates of vitamin usage aimed at preventing AMD could be estimated by a representative sample survey.

The main issue around the effectiveness of the potential interventions for dry and wet AMD is similar to that described for the current treatment options of laser photocoagulation and PDT – the need for long-term follow-up data. As and when these interventions are approved and implemented in the UK, longitudinal studies should be set up to monitor their longer term effectiveness. It is recognised that such longitudinal studies are not ideally suited to establishing long-term effectiveness, but extended follow-up in the relevant pharmaceutical companysponsored clinical trials is unlikely. Longitudinal studies should provide some indication of the continued effectiveness of an intervention.

Costs of blindness

The described costs associated with blindness and deteriorating vision were a significant factor in the AMD screening model, especially in the anti-VEGF sensitivity analyses, where the no-screening option incurred higher lifetime costs due to the costs of blindness. The AMD screening model incorporates a binary threshold for costs associated with blindness: no such costs are incurred when the VA in the best-seeing eye is better than 6/60 or worse; significant costs are incurred once an individual crosses the threshold and VA in the best-seeing eye is 6/60 or worse. This is unlikely to be a true representation, as at least some of the costs associated with blindness and deteriorating vision would be more appropriately described on a continuum.

Costs to the NHS and local and central government associated with blindness and rapidly deteriorating vision were informed by the estimates presented in an HTA of PDT.⁷⁵ The mean annual cost of £6295 comprises a wide range of resources that are not specific to individuals who are registered as being blind. The cost estimates do not describe the **additional** proportion of individuals who experience events such as hip fractures, depression or the provision of community care. The main cost driver is admission to residential care, which is based on the proportion of residents with low vision, the total number of individuals in residential care and the estimated prevalence of AMD in the elderly, leading to the estimation that 30% of individuals with low vision are in residential care. This analysis does not appear to describe the additional proportion of individuals in residential care due to low vision.

Therefore, another potential area for further research is a costing study of the impact of low vision. Ideally, the analysis should describe costs as a function of age and vision, including the full range of visual ability (e.g. from 6/6 to 3/60). The cost analysis presented by Meads and Moore⁷⁵ provides a starting point for the range of resource items to be included in the study, although additional reviews of the literature and primary research may be necessary to increase the validity of the findings.

Conclusions

Well-defined criteria informing the provision of screening programmes have long been defined,⁸ which have been adapted and adopted by the NSC. These criteria are interpreted in the light of the evidence identified as part of this study in Appendix 5. The conclusion is that there is probably not enough knowledge to justify the implementation of a screening programme for AMD at present.

It has been proposed that allocative decisions should be based on the mean results of economic evaluations of healthcare technologies.⁸⁷ However, the extent of the uncertainty around the mean results, and the additional resources and possible reorganisation of services required to implement screening, indicate that it may be preferable to reduce the level of uncertainty before implementing a *de novo* screening programme for AMD.

The evaluation of screening for AMD has identified an issue around the use of EVI analysis, which concerns a threshold of suitability of costeffectiveness models for informing research priorities, let alone healthcare resource priorities. The AMD screening model has extreme levels of uncertainty around key input parameters, some of which cannot be feasibly estimated using expert elicitation techniques. We contend that the next stage in the evaluation of screening for AMD should identify, and cost, methods for improving the accuracy of these key parameters.

Research recommendations

Table 76 provides examples of the kind of studies that may be required to inform different key

Parameter	Study type	Cost level	Model impact
Incidence of ARM	Longitudinal population-based	High	Medium
Clinical presentation of ARM	IT study integrating routine data	Medium	High
AMD progression by factors other than visual acuity	Longitudinal patient-based; change		-
	in trial reporting standards	High	Low
Screening and optometrist referral effectiveness	Pilot study	Medium	High
Long-term treatment effectiveness	Longitudinal patient-based	Medium	Medium
Costs of blindness	Primary costing study	Low	Medium

 TABLE 76
 Possible studies to inform key parameters in the AMD screening model

parameters. It also provides a subjective assessment of the relative cost levels of implementing such studies, and the impact of improved data in reducing model uncertainty and increasing the suitability of the model to inform resource allocation decisions. how routine data may be used to describe clinical presentation rates of ARM. Other potential studies include a pilot study of the effectiveness of screening and optometrists' referral patterns for AMD, and a costing study of blindness as a continuum of association with deterioration in vision.

From the preliminary list presented in *Table 76*, initial actions may be best targeted at assessing

Acknowledgements

This research was funded by the NIHR HTA Programme. The views in this report are those of the authors and not necessarily those of the NIHR HTA Programme. Any errors are the responsibility of the authors.

Contribution of authors

Jon Karnon (Senior Research Fellow) led the project; developed, implemented, populated and analysed the model; contributed to the literature reviews; input to survey design and drafted and compiled the final report. Carolyn Czoski-Murray (Research Fellow) contributed to the literature reviews and organised the surveys and drafted sections of the report. Kevin Smith (Honorary Senior Clinical Lecturer in Public Health Medicine, Medical Advisor to the Yorkshire and the Humber Regional Specialised Commissioning Group) contributed to the literature reviews; contributed to data analyses and drafted sections of the report. Chris Brand (Consultant Ophthalmologist) was the main clinical contact for advice on the natural history, treatment and screening options for AMD; provided primary data and had an input to the survey design. Usha Chakravarthy (Professor of Ophthalmology and Vision Sciences) supplied advice on the natural history, treatment and screening options for AMD and provided primary data. Sarah Davies (Senior Health Economist) contributed to the data analyses. Nick Bansback (Health Economist) contributed to the literature reviews. Catherine Beverley (Knowledge Manager) informed and undertook literature searches. Alan Bird (Emeritus Professor), Simon Harding (Consultant Ophthalmologist), Iain Chisholm (Consultant Ophthalmologist) and YC Yang (Consultant Ophthalmologist) all gave advice on treatment options and screening for AMD and provided primary data.



- 1. Bird AC, Bressler NM, Bressler SB, Chisholm IH, Coscas G, Davis MD, *et al.* An international classification and grading system for age-related maculopathy and age-related macular degeneration. The International ARM Epidemiological Study Group. *Surv Ophthalmol* 1995;**39**:367–74.
- van Leeuwen R, Klaver CC, Vingerling JR, Hofman A, de Jong PT. Epidemiology of agerelated maculopathy: a review. *Eur J Epidemiol* 2003;18:845–54.
- Leibowitz HM, Krueger DE, Maunder LR, Milton RC, Kini MM, Kahn HA, *et al.* The Framingham Eye Study monograph: an ophthalmological and epidemiological study of cataract, glaucoma, diabetic retinopathy, macular degeneration, and visual acuity in a general population of 2631 adults, 1973–1975. *Surv Ophthalmol* 1980;**24** (Suppl):335–610.
- Owen CG, Fletcher AE, Donoghue M, Rudnicka AR. How big is the burden of visual loss caused by age related macular degeneration in the United Kingdom? *Br J Ophthalmol* 2003; 87:312–17.
- Friedman DS, O'Colmain BJ, Munoz B, Tomany SC, McCarty C, de Jong PT, *et al.* Prevalence of age-related macular degeneration in the United States. *Arch Ophthalmol* 2004; 122:564–72.
- Chang TS, Hay D, Courtright P. Age-related macular degeneration in Chinese-Canadians. *Can J Ophthalmol* 1999;**34**:266–71.
- Bressler NM, Munoz B, Maguire MG, Vitale SE, Schein OD, Taylor HR, *et al.* Five-year incidence and disappearance of drusen and retinal pigment epithelial abnormalities. Waterman study. *Arch Ophthalmol* 1995;113:301–8.
- Wilson J, Junger G. Principles and practice of screening for disease. Public Health Paper 34. Geneva: World Health Organization; 1968.
- National Screening Committee. NSC criteria for appraising the viability, effectiveness and appropriateness of a screening programme. URL: http://www.nsc.nhs.uk/pdfs/Criteria.pdf. Accessed 7 July 2007.
- Bartlett H, Eperjesi F. Age-related macular degeneration and nutritional supplementation: a review of randomised controlled trials. *Ophthalmic Physiol Opt* 2003;23:383–99.

- 11. Smith W, Assink J, Klein R, Mitchell P, Klaver CC, Klein BE, *et al.* Risk factors for age-related macular degeneration: pooled findings from three continents. *Ophthalmology* 2001;**108**:697–704.
- Seddon JM, Willett WC, Speizer FE, Hankinson SE. A prospective study of cigarette smoking and age-related macular degeneration in women. *JAMA* 1996;**276**:1141–6.
- Christen WG, Glynn RJ, Manson JE, Ajani UA, Buring JE. A prospective study of cigarette smoking and risk of age-related macular degeneration in men. *JAMA* 1996;**276**:1147–51.
- Age-Related Eye Disease Research Study Group. Risk factors associated with age-related macular degeneration. A case-control study in the agerelated eye disease study: age-related eye disease study report number 3. *Ophthalmology* 2000;**107**: 2224–32.
- Delcourt C, Diaz JL, Ponton-Sanchez A, Papoz L. Smoking and age-related macular degeneration. The POLA Study. Pathologies Oculaires Liées à l'Age. Arch Ophthalmol 1998;116:1031–5.
- Vingerling JR, Hofman A, Grobbee DE, de Jong PT. Age-related macular degeneration and smoking. The Rotterdam Study. *Arch Ophthalmol* 1996;**114**:1193–6.
- Delcourt C, Michel F, Colvez A, Lacroux A, Delage M, Vernet MH, *et al.* Associations of cardiovascular disease and its risk factors with agerelated macular degeneration: the POLA study. *Ophthalmic Epidemiol* 2001;8:237–49.
- Klein R, Klein BE, Tomany SC, Cruickshanks KJ. The association of cardiovascular disease with the long-term incidence of age-related maculopathy: the Beaver Dam Eye Study. *Ophthalmology* 2003; 110:636–43.
- Klein R, Klein BE, Jensen SC. The relation of cardiovascular disease and its risk factors to the 5-year incidence of age-related maculopathy: the Beaver Dam Eye Study. *Ophthalmology* 1997; 104:1804–12.
- 20. Klein R, Klein BE, Wong TY, Tomany SC, Cruickshanks KJ. The association of cataract and cataract surgery with the long-term incidence of age-related maculopathy: the Beaver Dam Eye Study. *Arch Ophthalmol* 2002;**120**:1551–8.
- 21. Chaine G, Hullo A, Sahel J, Soubrane G, Espinasse-Berrod MA, Schutz D, *et al*. Case–control study of the risk factors for age related macular

degeneration. France–DMLA Study Group. BrJ Ophthalmol 1998;82:996–1002.

- 22. Buch H, Nielsen N, Vinding T, Jensen G, Prause JU, la Cour M. 14-Year incidence, progression and visual morbidity of age-related maculopathy. *Ophthalmology* 2005;**112**:787–98.
- 23. Mitchell P, Wang JJ, Foran S, Smith W. Five-year incidence of age-related maculopathy lesions: the Blue Mountains Eye Study [published erratum appears in *Ophthalmology* 2002;**109**:1588]. *Ophthalmology* 2002;**109**:1092–7.
- 24. Tomany SC, Klein R, Klein BE. The relation of coffee and caffeine to the 5-year incidence of early age-related maculopathy: the Beaver Dam Eye Study. *Am J Ophthalmol* 2001;**132**:271–3.
- 25. Age-Related Eye Disease Study Research Group. A randomized, placebo-controlled, clinical trial of high-dose supplementation with vitamins C and E, beta carotene, and zinc for age-related macular degeneration and vision loss: AREDS report no. 8. *Arch Ophthalmol* 2001;**119**:1417–36.
- 26. Sparrow JM, Dickinson AJ, Duke AM, Thompson JR, Gibson JM, Rosenthal AR. Seven year follow-up of age-related maculopathy in an elderly British population. *Eye* 1997;11(Pt 3):315–24.
- 27. Age-Related Eye Disease Study Research Group. Age-Related Macular Degeneration in the Age-Related Eye Disease Study (AREDS): AREDS report no. 19. *Ophthalmology* 2005;**112**:533–9.
- Choroidal Neovascularization Prevention Trial Research Group. Laser treatment in eyes with large drusen. Short-term effects seen in a pilot randomized clinical trial. Choroidal Neovascularization Prevention Trial Research Group. *Ophthalmology* 1998;105:11–23.
- Burgess DB, Hawkins BS, Jefferys JL, Bressler NM, Bressler SB, Hiner CJ, *et al.* 5-Year follow-up of fellow eyes of patients with age-related macular degeneration and unilateral extrafoveal choroidal neovascularization. *Arch Ophthalmol* 1993; 111:1189–99.
- Wilkinson CP, Hawkins BS, Javornik NB, Newhouse MM, Keller JA, Bressler NM, *et al.* Five-year follow-up of fellow eyes of individuals with ocular histoplasmosis and unilateral extrafoveal or juxtafoveal choroidal neovascularization. *Arch Ophthalmol* 1996; 114:677–88.
- Macular Photocoagulation Study Group. Risk factors for choroidal neovascularization in the second eye of patients with juxtafoveal or subfoveal choroidal neovascularization secondary to age-related macular degeneration. Macular Photocoagulation Study Group. *Arch Ophthalmol* 1997;115:741–7.

- 32. Bressler NM, Frost LA, Bressler SB, Murphy RP, Fine SL. Natural course of poorly defined choroidal neovascularization associated with macular degeneration. *Arch Ophthalmol* 1988; **106**:1537–42.
- 33. Sunness JS, Gonzalez-Baron J, Bressler NM, Hawkins B, Applegate CA. The development of choroidal neovascularization in eyes with the geographic atrophy form of age-related macular degeneration. *Ophthalmology* 1999;**106**:910–19.
- 34. Pauleikhoff D, Radermacher M, Spital G, Muller C, Brumm G, Lommatzsch A, et al. Visual prognosis of second eyes in patients with unilateral late exudative age-related macular degeneration. *Graefes Arch Clin Exp Ophthalmol* 2002;**240**:539–42.
- Baun O, Vinding T, Krogh E. Natural course in fellow eyes of patients with unilateral age-related exudative maculopathy. A fluorescein angiographic 4-year follow-up of 45 patients. *Acta Ophthalmol* 1993;**71**:398–401.
- Wang JJ, Foran S, Smith W, Mitchell P. Risk of agerelated macular degeneration in eyes with macular drusen or hyperpigmentation: the Blue Mountains Eye Study cohort. *Arch Ophthalmology* 2003; 121:658–63.
- Holz FG, Wolfensberger TJ, Piguet B, Gross-Jendroska M, Wells JA, Minassian DC, et al. Bilateral macular drusen in age-related macular degeneration. Prognosis and risk factors. *Ophthalmology* 1994;101:1522–8.
- Sarks JP, Sarks SH, Killingsworth MC. Evolution of geographic atrophy of the retinal pigment epithelium. *Eye* 1988;2:552–77.
- Schatz H, McDonald HR. Atrophic macular degeneration: rate of spread of geographic atrophy and visual loss. *Ophthalmology* 1989; 96:1551.
- Ali F, Chan WC, Stevenson MR, Muldrew KA, Chakravarthy U. Morphometric analysis of angiograms of exudative lesions in age-related macular degeneration. *Arch Ophthalmol* 2004; 122:710–15.
- 41. Sunness JS. The natural history of geographic atrophy, the advanced atrophic form of age-related macular degeneration. *Mol Vis* 1999;**5**:25.
- Sunness JS, Gonzalez-Baron J, Applegate CA, Bressler NM, Tian Y, Hawkins B, *et al.* Enlargement of atrophy and visual acuity loss in the geographic atrophy form of age-related macular degeneration. *Ophthalmology* 1999; 106:1768–79.
- Stevenson MR, Hart PM, Chakravarthy U, MacKenzie G, Bird AC, Owens SL, *et al.* Visual functioning and quality of life in the SubFoveal Radiotherapy Study (SFRADS): SFRADS report 2. *Br J Ophthalmol* 2005;89:1045–51.

- 44. Zaidi FH, Cheong-Leen R, Gair EJ, Weir R, Sharkawi E, Lee N, *et al.* The Amsler chart is of doubtful value in retinal screening for early laser therapy of subretinal membranes. The West London Survey. *Eye* 2004;**18**:503–8.
- Barthes A, Conrath J, Rasigni M, Adel M, Petrakian JP. Mathematical morphology in computerized analysis of angiograms in agerelated macular degeneration. *Med Phys* 2001; 28:2410–19.
- Fine AM, Elman MJ, Ebert JE, Prestia PA, Starr JS, Fine SL. Earliest symptoms caused by neovascular membranes in the macula. *Arch Ophthalmol* 1986; 104:513–14.
- Schmitt NJ, Grover DA, Feldon SE. The Eger Macular Stressometer: pilot study. *Am J Ophthalmol* 2003;**136**:314–17.
- 48. Lafaut BA, De Backer E, Kohno T, De Laey JJ, Meire FM. Indocyanine green angiography in Sorsby's fundus dystrophy. *Graefes Arch Clin Exp Ophthalmol* 2000;**238**:158–62.
- 49. Macular Photocoagulation Study Group. Argon laser photocoagulation for neovascular maculopathy: five year results from randomized clinical trials. *Arch Ophthalmol* 1991;**109**:1109–14.
- 50. Evans JR. Antioxidant vitamin and mineral supplements for age-related macular degeneration (Cochrane Review) In *The Cochrane Library*, Issue 1. Chichester: Wiley; 2004.
- Teikari JM, Laatikainen L, Virtamo J, Haukka J, Rautalahti M, Liesto K, *et al.* Six-year supplementation with alpha-tocopherol and betacarotene and age-related maculopathy. *Acta Ophthalmol Scand* 1998;**76**:224–9.
- 52. ClinicalTrials.gov. Open-label posterior juxtascleral injections of anecortave acetate 15 mg dose for long term use in patients with AMD. URL: http://www.clinicaltrials.gov/ct/show/NCT00065728; jsessionid=547BE5D966347E4DA5750BE230BB9 F66?order=1. Accessed 7 July 2007.
- 53. Pulido JS, Sanders DR, Klingel R. Rheopheresis for age-related macular degeneration: clinical results and putative mechanism of action. *Can J Ophthalmol* 2005;**40**:332–40.
- Pulido JS, Multicenter Investigation of Rheopheresis for AMD (MIRA). Multicenter prospective, randomized, double-masked, placebocontrolled study of Rheopheresis to treat nonexudative age-related macular degeneration: interim analysis. *Trans Am Ophthalmol Soc* 2002; 100:85–106.
- 55. Panel discussion. Evolving treatment strategies for exudative AMD. The combination of verteporfin PDT and triamcinolone is already changing practice patterns. *Retinal Physician* 2005. URL: http://www.retinalphysician.com/article.aspx?article =100108. Accessed 17 April 2008.

- 56. Jonas JB, Kreissig I, Degenring RF. Factors influencing visual acuity after intravitreal triamcinolone acetonide as treatment of exudative age related macular degeneration. *Br J Ophthalmol* 2004;**88**:1557–62.
- 57. Bhavsar A. Combined verteporfin photodynamic therapy and intravitreal triamcinolone in the treatment of minimally classic subfoveal CNV with or without RAP lesions. AAO poster presentation P0163. 2004.
- 58. Augustin AJ, Schmidt-Erfurth U. Verteporfin therapy combined with intravitreal triamcinolone in all types of choroidal neovascularisation due to age-related macular degeneration. *Ophthalmology* 2006;**113**:14–22.
- Eye facts. Anti-VEGF treatment for age-related macular degeneration. URL: https://secure3.aao.org/pdf/057052.pdf. Accessed 7 July 2007.
- 60. Gragoudas ES, Adamis AP, Cunningham ET Jr, Feinsod M, Guyer DR. Pegaptanib for neovascular age-related macular degeneration. *N Engl J Med* 2004;**351**:2805–16.
- 61. Genentech. Genentech granted six-month priority review for ranubizimab for the treatment of wet AMD. URL: http://www.mdsupport.org/. Accessed 7 July 2007.
- 62. Roberts D. *Avastin for wet AMD*? February 2007. URL: http://www.mdsupport.org/library/ avastin.html. Accessed 7 July 2007.
- 63. Thomas MA, Ibanez HE. Subretinal endophotocoagulation in the treatment of choroidal neovascularization. *Am J Ophthalmol* 1993;**116**:279–85.
- 64. van Meurs JC, van den Biesen PR. Autologous retinal pigment epithelium and choroid translocation in patients with exudative agerelated macular degeneration: short-term followup. *Am J Ophthalmol* 2003;**136**:688–95.
- Moss SM, Gray A, Legood R, Henstock E. Evaluation of HPV/LBC Cervical Screening Pilot Studies. First report to the Department of Health on evaluation of LBC. London: Department of Health; 2002.
- 66. Roderick P, Davies R, Raftery J, Crabbe D, Bhandari P, Patel P. The cost-effectiveness of screening for *Helicobacter pylori* to reduce mortality and morbidity from gastric cancer and peptic ulcer disease: a discrete event simulation model. *Health Technol Assess* 2003;7(6).
- 67. Goverdhan SV, Booth AJ, Luff AJ, Lotery AJ. Age-related macular degeneration. Where are we now? *Optometry Today* 2004; (17 December) 39–43.
- 68. Busbee B, Brown MM, Brown GC, Sharma S. CME review: a cost–utility analysis of laser photocoagulation for extrafoveal choroidal neovascularisation. *Retina* 2003;**23**:279–87.

- 69. Bonastre J, Le Pen C, Soubrane G, Quentel G. The burden of age-related macular degeneration: results of a cohort study in two French referral centres. *Pharmacoeconomics* 2003;**21**:181–90.
- Sharma S, Bakal J, Sharma SM, Covert D, Shah GK. Drug pricing for a novel treatment for wet macular degeneration: using incremental costeffectiveness ratios to ensure societal value. *Can J Ophthalmol* 2005;**40**:369–77.
- van Leeuwen R, Chakravarthy U, Vingerling J, Brussee C, Hooghart A, Mulder P, *et al.* Grading of age-related maculopathy for epidemiological studies – is digital imaging as good as 35-mm film? *Ophthalmology* 2003;**110**:1540–4.
- Pollack A. Genentech in competition with itself on eye drug. *New York Times* 27 September, 2005. URL: http://www.nytimes.com/2005/09/27/ business/27place.html. Accessed 17 April 2008.
- 73. Cooke R. I can see clearly now. At last, therapies improve vision lost to macular degeneration. *AARP Bull* 2005. URL: http://www.aarp.org/bulletin/yourhealth/macular.ht ml. Accessed 17 April 2008.
- 74. EyeTech. Press release: Safety profile of Pegaptanib[®] (pegaptanib sodium injection) in the treatment of neovascular age-related macular degeneration (AMD) maintained for two years. New York, 2 May, 2005. URL: http://www.mpmcapital.com/newsroom/news/ 050502ii.pdf. Accessed 7 July 2007.
- 75. Meads C, Moore D. The clinical effectiveness and cost utility of photodynamic therapy for wet agerelated macular degeneration: a systematic review and economic evaluation. *Health Technol Assess* 2001;7(9).
- Espallargues M, Czoski-Murray C, Bansback N, Carlton J, Lewis G, Hughes L, *et al.* The impact of age related macular degeneration on health state utility values. *Invest Ophthalmol Vis Sci* 2005; 46:4016–23.
- Brown GC, Sharma S, Brown MM, Kistler J. Utility values and age-related macular degeneration. *Arch Ophthalmol* 2000;118:47–51.
- Brown MM, Brown GC, Sharma S, Landy J, Bakal J. Quality of life with visual acuity loss from diabetic retinopathy and age-related macular degeneration. *Arch Ophthalmol* 2002;**120**:481–4.
- 79. Meads C, Hyde C. Photodynamic therapy with verteporfin is effective, but how big is its effect? Results of a systematic review. *Br J Ophthalmol* 2004;**88**:212–17.
- 80. Evans J, Wormald R. Is the incidence of registrable age-related macular degeneration increasing? *Br J Ophthalmol* 1996;**80**:9–14.
- Karlsson G, Johannesson M. The decision rules of cost-effectiveness analysis. *Pharmacoeconomics* 2006; 9:113–20.

- Towse A, Devlin N, Pritchard C, editors. Cost effectiveness thresholds: economic and ethical issues. London: Office for Health Economics/King's Fund; 2002.
- Stevenson M, Oakley J, Chilcott J. Gaussian process modelling in conjunction with individual patient simulation modelling. A case study describing the calculation of cost-effectiveness ratios for the treatment of osteoporosis. *Med Decis Making* 2004;24:89–100.
- 84. Weinstein MC, O'Brien B, Hornberger J, Jackson J, Johannesson M, McCabe C, *et al.* Principles of good practice for decision analytic modeling in health-care evaluation: report of the ISPOR Task Force on Good Research Practices – Modeling Studies. *Value Health* 2003;6:9–17.
- McClure ME, Hart PM, Jackson AJ, Stevenson MR, Chakravarthy U. Macular degeneration: do conventional measurements of impaired visual function equate with visual disability? *Br J Ophthalmol* 2000;84:244–50.
- Vaclavik V, Vujosevic S, Dandekar SS, Bunce C, Peto T, Bird AC. Autofluorescence imaging in agerelated macular degeneration complicated by choroidal neovascularization. A prospective study. *Ophthalmology* 2008;115:342–6.
- 87. Claxton K. The irrelevance of inference: a decision-making approach to the stochastic evaluation of health care technologies. *J Health Econ* 1999;**18**:341–64.
- Woods RL, Tregear SJ, Mitchell RA. Screening for ophthalmic disease in older subjects using visual acuity and contrast sensitivity. *Ophthalmology* 1998;105:2318–232.
- Vingerling JR, Dielemans I, Hofman A, Grobbee DE, Hijmering M, Kramer CF, *et al.* The prevalence of age-related maculopathy in the Rotterdam Study. *Ophthalmology* 1995;**102**:205–10.
- 90. Das BN, Thompson JR, Patel R, Rosenthal AR. The prevalence of eye disease in Leicester: a comparison of adults of Asian and European descent. J R Soc Med 1994;**87**:219–22.
- 91. Dickinson AJ, Sparrow JM, Duke AM, Thompson JR, Gibson JM, Rosenthal AR. Prevalence of age-related maculopathy at two points in time in an elderly British population. *Eye* 1997;11(Pt 3):301–14.
- Vinding T. Visual impairment of age-related macular degeneration. An epidemiological study of 1000 aged individuals. *Acta Ophthalmol* 1990; 68:162–7.
- 93. Klaver CCW, Wolfs RCW, Vingerling JR, Hofman A, De Jong PTVM. Age-specific prevalence and causes of blindness and visual impairment in an older population: the Rotterdam study. Arch Ophthalmol 1998;116:653–8.

- 94. Klaver CC, Assink JJ, van Leeuwen R, Wolfs RC, Vingerling JR, Stijnen T, et al. Incidence and progression rates of age-related maculopathy: the Rotterdam Study. *Invest Ophthalmol Vis Sci* 2001; 42:2237–41.
- Maguire MG. More pieces for the age-related macular degeneration puzzle, *Ophthalmology* 1997; 104:5–6.
- 96. Schachat AP, Hyman L, Leske MC, Connell AM, Wu SY. Features of age-related macular degeneration in a black population. The Barbados Eye Study Group. *Arch Ophthalmol* 1995; 113:728–35.
- 97. Weih LM, VanNewkirk MR, McCarty CA, Taylor HR. Age-specific causes of bilateral visual impairment. *Arch Ophthalmol* 2000;**118**:264–9.
- Reidy A, Minassian DC, Vafidis G, Joseph J, Farrow S, Wu J, *et al.* Prevalence of serious eye disease and visual impairment in a north London population: population based, cross sectional study. *BMJ* 1998;**316**:1643–6.
- 99. Mitchell RA. Prevalence of age related macular degeneration in persons aged 50 years and over resident in Australia. *J Epidemiol Commun Health* 1993;**47**:42–5.
- Munier A, Gunning T, Kenny D, O'Keefe M. Causes of blindness in the adult population of the Republic of Ireland. *Br J Ophthalmol* 1998; 82:630–3.
- Foran S, Wang JJ, Mitchell P. Causes of incident visual impairment: the Blue Mountains Eye Study. *Arch Ophthalmol* 2002;**120**:613–19.
- 102. Huang GH, Klein R, Klein BE, Tomany SC. Birth cohort effect on prevalence of age-related maculopathy in the Beaver Dam Eye Study. *Am J Epidemiol* 2003;**157**:721–9.
- 103. Gibson JM, Rosenthal AR, Lavery J. A study of the prevalence of eye disease in the elderly in an English community. *Trans Ophthalmol Soc UK* 1985; 104(Pt 2):196–203.
- 104. Cruickshanks KJ, Hamman RF, Klein R, Nondahl DM, Shetterly SM. The prevalence of age-related maculopathy by geographic region and ethnicity. The Colorado–Wisconsin Study of Age-Related Maculopathy. *Arch Ophthalmol* 1997; 115:242–50.

- 105. VanNewkirk MR, Nanjan MB, Wang JJ, Mitchell P, Taylor HR, McCarty CA. The prevalence of agerelated maculopathy: the visual impairment project. *Ophthalmology* 2000;**107**:1593–600.
- 106. Foran S, Wang JJ, Mitchell P. Causes of visual impairment in two older population cross-sections: the Blue Mountains Eye Study. *Ophthalmic Epidemiol* 2003;**10**:215–25.
- 107. Buch H, Vinding T, Nielsen NV. Prevalence and causes of visual impairment according to World Health Organization and United States criteria in an aged, urban Scandinavian population: the Copenhagen City Eye Study. *Ophthalmology* 2001; 108:2347–57.
- 108. Wang JJ, Foran S, Mitchell P. Age-specific prevalence and causes of bilateral and unilateral visual impairment in older Australians: the Blue Mountains Eye Study. *Clinical Exp Ophthalmol* 2000;**28**:268–73.
- Mitchell P, Smith W, Attebo K, Wang JJ. Prevalence of age-related maculopathy in Australia. The Blue Mountains Eye Study. *Ophthalmology* 1995; 102:1450–60.
- 110. Javitt JC, Zhou Z, Maguire MG, Fine SL, Wilke RJ. Incidence of exudative age-related macular degeneration among elderly Americans. *Ophthalmology* 2003;**110**:1534–9.
- 111. Klein R, Klein BE, Tomany SC, Moss SE. Ten-year incidence of age-related maculopathy and smoking and drinking: the Beaver Dam Eye Study. *Am J Epidemiol* 2002;**156**:589–98.
- 112. Rahmani B, Tielsch JM, Katz J, Gottsch J, Quigley H, Javitt J, *et al.* The cause-specific prevalence of visual impairment in an urban population: the Baltimore Eye Survey. *Ophthalmology* 1996;**103**:1721–6.
- 113. Farber MD. National Registry for the Blind in Israel: estimation of prevalence and incidence rates and causes of blindness. *Ophthalmic Epidemiol* 2003;10:267–77.
- 114. Frennesson CI. Photodynamic therapy with verteporfin in patients with age-related macular degeneration and juxtafoveal choroidal neovascularisation. *Acta Ophthalmol Scand* 2004;**82**:651–5.

Appendix I

Literature search plans and results

Search strategies

A comprehensive literature search was undertaken during March 2004 (updated in January 2005) to identify relevant literature pertaining to screening for AMD. Four major searches were conducted which were designed to retrieve:

- high level evidence (i.e. guidelines, systematic reviews and RCTs) concerning AMD
- papers describing the epidemiology of AMD
- papers describing the diagnosis and diagnostic tests associated with AMD
- cost-effectiveness and health utility literature in the field.

The following electronic bibliographic databases were searched:

- 1. Cochrane Database of Systematic Reviews (CDSR)
- 2. Cochrane Central Register of Controlled Trials (CENTRAL)
- 3. EMBASE
- 4. MEDLINE
- 5. NHS Database of Abstracts of Reviews of Effects (DARE)
- 6. NHS Health Technology Assessment (HTA) database
- 7. Science Citation Index (SCI)
- 8. Social Sciences Citation Index (SSCI)
- 9. Turning Research into Practice (TRIP) database.

Attempts were also made to identify 'grey' literature by searching appropriate databases (e.g. Health Management Information Consortium, Index to Theses, Dissertation Abstracts), current research registers (e.g. National Research Register, Research Findings Register, Current Controlled Trials) and relevant websites (e.g. Macular Disease Society, Moorfields Eye Hospital, Royal National Institute for the Blind).

The reference lists of included studies and relevant review articles were also checked.

The search strategy used in MEDLINE (Ovid) was as follows:

1 exp macular degeneration/

- 2 macula\$ degen\$.tw
- 3 maculopath\$.tw
- 4 AMD.tw
- 5 ARMD.tw
- 6 or/1-5

This strategy was combined with methodological search filters designed to retrieve the highest levels of evidence, for example:

Guidelines

- 7 guideline.pt
- 8 practice guideline.pt
- 9 health planning guidelines/
- 10 or/1-3

Systematic reviews

- 1 meta-analysis/
- 2 exp review literature/
- 3 (meta-analy\$ or meta analy\$ or metaanaly\$).tw
- 4 meta analysis.pt
- 5 review academic.pt
- 6 review literature.pt
- 7 (systematic\$ adj3 (review\$ or overview\$)).tw
- 8 or/1-7

RCTs

1 clinical trial.pt

The strategy was also combined with the following 'epidemiology' search terms:

- 1 exp epidemiology/
- 2 epidemiolog\$.ti
- 3 inciden\$.ti
- 4 prevalen\$.ti
- 5 incidence/
- 6 prevalence/
- 7 or/1-6

Finally, the strategy was combined with the following 'diagnosis' search terms:

- 1 fluoresein.tw
- 2 amsler.tw
- 3 diagnos\$.ti
- 4 screen\$.ti
- 5 exp *mass screening/
- 6 diagnosis/
- 7 macula\$ computer\$ psychosphysical test\$.tw

- 8 mcpt.tw
- 9 *angiography/
- 10 *fluorescein angiography/
- 11 angiograph\$.ti
- 12 indocyanine green.tw
- 13 or/1-12

No date or language restrictions were applied to the searches.

In terms of cost-effectiveness and utility literature, searches were conducted in MEDLINE, EMBASE, NHS Economic Evaluations Database (EED) and OHE Health Economic Evaluations Database (HEED). Search filters designed to retrieve economic evaluations, economic models and QoL literature were applied to the MEDLINE and EMBASE searches. An example of the filter used in MEDLINE (Ovid) is provided below:

- 1 exp patient acceptance of health care/
- 2 exp "costs and cost analysis"/
- 3 cost\$.ti
- 4 (cost\$ adj2 (effective\$ or util\$ or benefit\$ or minimi\$)).ab
- 5 (economic\$ or pharmacoeconomic\$ or pharmaco-economic\$).tw
- 6 quality adjusted life year/
- 7 quality adjusted life.tw
- 8 (qaly\$ or qald\$ or qale\$ or qtime\$).tw
- 9 disability adjusted life.tw
- 10 daly\$.tw
- 11 health status indicators/
- 12 (sf36 or sf 36 or short form 36 or shortform 36 or sf thirtysix or sf thirty six or shortform thirtysix or shortform thirty six or short form thirtysix or short form thirty six).tw

- 13 (sf6 or sf 6 or short form 6 or shortform 6 or sf six or sfsix or shortform six or short form six).tw
- 14 (sf12 or sf 12 or short form 12 or shortform 12 or sf twelve or sftwelve or shortform twelve or short form twelve).tw
- 15 (sf16 or sf 16 or short form 16 or shortform 16 or sf sixteen or sfsixteen or shortform sixteen or short form sixteen).tw
- 16 (sf20 or sf 20 or short form 20 or shortform 20 or sf twenty or sftwenty or shortform twenty or short form twenty).tw
- 17 (euroqol or euro qol or eq5d or eq 5d).tw
- 18 (hql or hqol or h qol or hrqol or hr qol).tw
- 19 (hye or hyes).tw
- 20 health\$ year\$ equivalent\$.tw
- 21 health utilit\$.tw
- 22 (hui or hui1 or hui2 or hui3).tw
- 23 disutil\$.tw
- 24 rosser.tw
- 25 quality of wellbeing.tw
- 26 qwb.tw
- 27 willingness to pay.tw
- 28 standard gamble\$.tw
- 29 time trade off.tw
- 30 time tradeoff.tw
- 31 tto.tw
- 32 exp models, economic/
- 33 *models, theoretical/
- 34 *models, organizational/
- 35 economic model\$.tw
- 36 markov chains/
- 37 markov\$.tw
- 38 monte carlo method/
- 39 monte carlo.tw
- 40 exp decision theory/
- 41 (decision\$ adj2 (tree\$ or analy\$ or model\$)).tw
- 42 or/1-41

Database	Host/system	Date searched	No. of hits	Location of search strategy	Notes
CDSR	Ovid	4 January 2004	9 (cdsrup.txt)	Paper copy only	
CENTRAL	Ovid	4 January 2004	I3 (centup.txt)	Paper copy only	
MEDLINE 2004-	Ovid	4 January 2004	Guidelines: 4 (mamdupg.txt)	"cbmHTAAMD" in Ovid	
			Systematic reviews: 12 (mamdupsr.txt)		
			Epidemiology: 25 (mamdupep.txt)		
			Treatments (clinical trials): 31 (mamdupct.txt)		
			Diagnosis: 43 (mamdupdiag.txt)		
			Acceptability/utilities: 23 (mamduput.txt)		
CENTRAL/CCTR	Ovid	3 March 2004	297 (cctramd.txt)	Paper copy only	Imported into Ref Man
CDSR	Ovid	3 March 2004	7 (cdsramd.txt)	Paper copy only	Imported into Ref Man
CINAHL 1982-	Ovid				
Citation Indexes (Science and Social Sciences) Author search	Web of Science	3 March 2004	175 (ciamdauthor.txt)	Paper copy only	Imported into Ref Man
CRD Databases	CRD Website http://agatha.york.ac.uk welcome.htm	3 March 2004 /	35 (crdamd.txt)	Paper copy only	Imported into Ref Man
DARE	Ovid	3 March 2004	0	_	
EMBASE 1980-	SilverPlatter WebSPIRS		Guidelines: 43 (eamdg.txt)	Paper copies only	Imported into Ref Man
			Systematic reviews: 15 (eamdsr.txt)		
			Epidemiology: 289 (eamdep.txt)		
			Treatments (clinical trials): 199 (eamdct.txt)		
			Diagnosis: 887 (eamddiag.txt)		
			Acceptability/utilities: 115 (eamdut.txt)		
HMIC	SilverPlatter WinSPIRS				
MEDLINE 1966-	Ovid	3 March 2004	Guidelines: 18 (mamdg.txt)	"cbmHTAAMD" in Ovid	Imported into Ref Man
			Systematic reviews: 92 (mamdsr.txt)		
			Epidemiology: 322 (mamdep.txt)		
			Treatments (clinical trials): 381 (mamdct.txt)		
			Diagnosis: 702 (mamddiag.txt)		
			Acceptability/utilities: 62 (mamdut.txt)		
PreMEDLINE	Ovid				

Other sources searched

Other source	Date searched	No. of hits	Notes
Altavista or Google	3 March 2004	0 additional	
Citation/author searching			
Department of Health Website www.doh.gov.uk	3 March 2004	0	
Handsearching			
Index to Theses			
Macular Disease Society http://www.maculardisease.org/	3 March 2004	0 relevant	
Moorfields Eye Hospital http://www.moorfields.co.uk/Home	3 March 2004	I	Added to Ref Man
National Research Register http://www.update-software.com/national/	3 March 2004	Various	Majority added to Ref Man, others printed off and sent to JK
Research Findings Register (ReFeR) http://www.info.doh.gov.uk/doh/refr_web.nsf/ Home?OpenForm	3 March 2004	0 additional	
RNIB	3 March 2004	0 additional	
ScHARR Library catalogue			
TRIP database	3 March 2004	Various	Relevant abstracts printed off and added to Ref Man

Flow chart of study identification



Appendix 2

Studies of prevalence of AMD

Comments	They also looked at the prevalence of diabetic retinopathy and age-related cataract The FPC lists are out of date, which restricts recruitment 8 patients excluded from the analysis as they were of West Indian origin	continued
	European M 3% (SE 2.7%) 16% (SE 7.3%)	
	European F 3% (SE 3.1%) 28% (SE 7.2%)	
	Asian M 0% 7% (SE 6.4%)	
	Asian F 0% (SE 11.7%) = 0.51	
Results	Age $60-69$ $60-69$ $70+$ $p-Values:$ $p-Values:$ $Page p > 0.001$ Sex $p = 0.08$ Racial group p	
Diagnostic test Confirmation of disease Exclusions	 /A, near and distant Jull refraction Jull refraction Joimicroscopy Macula = area Jacula = area	
Subjects Age range (years) AMD Gender Ethnicity	atients 40 + from PC lists 60-69 00 + M/F <i>n</i> not reported Vigin selected to rigin selected to rigin selected to rigin for the	
Study design Duration of study Location Total number	Prospective, randomised observational study Leicester, UK 377 377	
Study	Das et <i>al.</i> (1994) ⁹⁰	

Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results	Comments
Dickinson et <i>al.</i> (1997) ⁹¹	Retrospective/ prospective randomised 7-year follow-up Melton Mowbray, Leics, UK 529 and 88 at follow-up. Total of 158 eyes with data	Participants in a large epidemiology study in 1982–4 Over 75 in 1981 70% female European	Ophthalmic examination with fundus photography. Mydriatic ophthalmic Examinations in ooth the initial and ollow-up studies. MARMGS VA 6/18 or worse 2 blinded 2 blinded	Prevalence AMD (total eyes $n = 158$):Initial testing (n) 23 (14.6%)% eyes $\geq 6/18$ 50.0Subsequent (n) 38 (24.1%)% eyes $\geq 6/18$ 48.7% eyes $\geq 6/18$ 48.7Prevalence of wet or dry AMD (total eyes $n = 158$):Initial wet (n) 3 (1.9%)Subsequent (n) 5 (3.8%)Initial dry (n) 5 (3.2%)	Also data on drusen and severity of AMD which may be useable Only 50% of survivors in the sample Also broken down by VA available
Vinding (1990) ⁹²	Prospective random Follow-up 10 years District of Copenhagen 946 survivors, 924 in final analysis = 1848 eyes	Participants in large F cardiac epidemiology study 6 (1981–3) 60–80 years European (Funduscopic Examination: Ophthalmoscopy and biomicroscopy with Goldmann 3 mirror lenses. (A 6/9 or worse (Framingham definition). E-chart single observer	AMD prevalence: No. of people % Total AMD 112/924 12 VA 6/9-6/12 (173 eyes) 71.7 VA 6/18-6.36 13.3 VA 6/60 or less 13.3	There are additional graphs representing % of wet and dry AMD
					continued

Study	Study design Duration of study Location Total number	Subjects / Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results					Comments
Klaver et al. (1998) ⁹³	Prospective follow-up of Rotterdam study 6775 subjects	55 + years. 77.7% of those identified as eligible from the municipal register consented to initial interview y y y	"Iedical record eview Diagnostic tests: pphthalmological examination, VA standardised grading protocol MHO: blind AC:0.5 (20/400), /isual impairment <0.3 (20/60)	Prevalence of AM Age (years) 55-74 75-84 85 + All ages (55 +) Percentages of ey	D as cause of bli Blind (64 14% (2 56% (10 78% (37 58% (37 es rather than pe	ndness and vi eyes) V eyes) eyes eyes) eyes ercentages of	sual impairmen isual impairmer 5% (1 e 28% (21 27% (48 25% (48 subjects	t: tr (192 eyes) tye) eyes) eyes) eyes)	No explanation is provided for the 15.2% that did not participate in the ophthalmological examinations The population examined was predominantly white Possible problems with multiple (3) observers. No mention of blinding
Klaver et <i>al.</i> (2001) ⁹⁴	Population-based, prospective cohort study Baseline: 6872 Follow-up: 4953 Rotterdam, The Netherlands	55 + years. 58.5% T were women at the follow-up to the follow-up to the sclusion: AMD at the baseline excluding the progressive to the progressive to the analyses (<i>n</i> = 105) until the follow of the	The screening for the presence of ARM followed the ame protocol at baseline and ollow-up. At ollow-up. At ollow-up side-by- side grading was used with the transparencies of the baseline phase	Age-specific incide incidences of AMI Age Per- (years) ye (years) ye (years) ye 55–64 35 55–64 35 55–64 35 55–64 19 75–84 19 75–84 19 75–84 19 76tal 98	ance rates per IC D in the Rotterdd son- No. ars ars 52 6 40 3 (49 12	000 person-ye Incidence rate 0 0.75 3.07 8.8 1.22	ars and 2-year 95% CI 0 to 1 0.15 to 2.2 1.1 to 6.7 1.8 to 25.8 0.6 to 2.1	cumulative 2-year cumulative ncidence (%) 0 0.15 0.61 1.75 0.24	In addition this paper has details of comparisons between Beaver Dam, Waterman and Rotterdam studies
									continued

Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results						Comments
				Stage	Criteria	Ĕ	At aseline	At follow-up	Selected for detailed grading	
				No AMD Ia	No AMD or drusen Soft, distinct drusen	≪ 63 μm	4038 1485	3238 1160	365 348	
				lb 2a	Pigmentary irreg Soft, indistinct or ret	icular	307 191	217 133	193 133	
				2b	Soft, indistinct druse pigmentary irregul	n with larities	209	I 58	158	
				m	Soft, indistinct or ret drusen with pigme irregularies	ticular entary	83	47	47	
				4	Atrophic or neovasci macular degenerat	ular tion	105	56	0	
Vingerling et <i>al.</i> (1995) ⁸⁹	Follow-up study to the Rotterdam study	55–98 years drawn from municipal register	Photographs graded using WARMGS	Age- and Age	sex-specific prevalence Atrophic AMD	e (%) of atra Neovascula	ophic and ar AMD	Tot	ar AMD: al AMD	Only 6251 (64%) of the eligible 9774 made it to the final analysis
	6251	59.7% women		(years) _N	1ale Female Both sexes	Male Fema	lle Both sexes	Male F	emale Both sexes	with gradable fundus photographs
		63% non-attenders female		65–74 (75–84 75–84 85 + 7 Total (0.2 0 0.1 0.6 0.2 0.4 1.1 1.4 1.3 2.7 4 3.7 0.6 0.6 0.6	0.1 0.1 0.3 0.5 2.4 2.4 6.8 7.5 0.8 1.2	0. 4. 2. 4. 4 1	0.3 9.5 4.	0.1 0.2 0.6 0.8 3.8 3.7 11.5 11 1.9 1.7	Table 2 in the journal article provides AMD and drusen definitions
										continued

 $\textcircled{\sc c}$ Queen's Printer and Controller of HMSO 2008. All rights reserved.

Comments	The results are based on black BES participants with gradable photographs in both eyes The frequency estimates for ARM changes may be less reliable in the older age groups as photograph gradability deteriorated as age increased							
) features: Overall frequency (%) 24.3 24.3 23.6 to 25.0 10, 10, 10, 10, 10, 10, 10, 10, 10, 10,							
	ies for any AME Female 591 (15.6%) 326 (33.7%) 106 (39.7%) 8 (50%) 8 (50%) 9 (95) (95) (95) (95) (95) (95) (95) (95)							
	erall % frequenci Male 482 (16%) 339 (26.6%) 86 (44.2%) 86 (44.2%) 86 (44.2%) 86 (44.2%) 95 (44.2%) 95 (0.1) (0.3 to 0.8) (0.3 to 0.8) (26.4 to 30.9)							
	especific and over Both sexes 1073 (15.8%) 821 (24%) 192 (41.7%) 16 (50%) 16 (50%) 16 (50%) 16 (50%) 26.3 (0.6 0.6 0.6							
Results	Age - and sex Age (years) 40-49 50-59 60-69 70-79 80+ AmD fea AmD fea AmD fea Amy AMD fea							
Diagnostic test Confirmation of disease Exclusions	tandardised xamination utomated efraction, best orrected VA, arly treatment iabetic etinotherapy tudy protocol undus hotographs phthalmological xamination. Drusens were raded from the umber of each typer form the umber of each presentation, trophy and xudative features <i>e</i> rer recorded MD changes: arly (drusen 63 µm), late exudative extures)							
Subjects Age range (years) AMD Gender Ethnicity	Median age of all BES participants was 59 years 57% were female, 33% self-reported 4 40-84 years 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7							
Study design Duration of study Location Total number	Prospective, random sample from resident population Barbados, West Indies Total 3444 Population-based study							
Study	Schachat et <i>al.</i> (1995) ⁹⁶							
Study	Study design Duration of study Location Total number	Subjects Age range (years AMD Gender Ethnicity	Diagnostic test) Confirmation of disease Exclusions	Results				Comments
--	--	--	--	--	--	--	---	---
Weih et <i>al.</i> (2000) ⁹⁷	Prospective Random selection from population- based study 5147	Random, cluster- stratified sample 40+ years	VA assessed using logMAR chart at 4 m. AMD were graded clinically and by photograph graders according to international classification schemes ¹	Cause-specific p Age group (year: 40–49 50–59 60–69 70–79 80–89 ≥90	revalence of vis s) Prevalenc	ion less than driving visi ce of AMD as cause of v (estimate no. of Au: 0 (0) 0 (0) 5.36 (23400 16.56 (10200	ion: dision less than driving stralians)	Figure 2 in the article provides table of severity of AMD: mild, moderate, severe and profound in different age groups
				Visual impairmer Cause	nt classification No. of cases o visual impairme. 364	by VA: f No nt Classified by VA loss criteria 87 (100) 333 (91)	. (%) Classified by visual field loss 0 (0) 24 (9)	
Reidy et <i>al</i> . (1998) ⁹⁸	Prospective Cross-sectional survey North London, England 1547	65+ years and registered with one of the 17 GP groups defined	VA assessed with person's own spectacles and logMAR scale 4 VA groupings: $6/6 + (normal vision), 6/6 - 6/12 (adequate for driving), < 6/12 (bilateral visual impairment), < 6/60 (blindness). AMD: VA < 6/12$	Population prevarable L in the ar Eye No. disorder in AMD	ilence of eye di ticle): of cases sample pol 133 {	sorders (for disorders c Estimated pulation prevalence (95 % CI) 3% (5.8 to 10.8)	ther than AMD, see Estimated no. of cases in population of 13,371 (95% CI) 1108 (776 to 1440)	Also available breakdown by social class/deprivation score and cases of AMD

 $\textcircled{\sc c}$ Queen's Printer and Controller of HMSO 2008. All rights reserved.

Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results		Comments
Mitchell (1993) ⁹⁹	Prospective and retrospective examination of medical records Western Metropolitan Health Region of New South Wales, Australia 3283 (1478 male, 1805 female) participants	Stratified by age, sex and local government area GP practices randomly chosen. 50+ years who attended these practices	Patient history, record review and non invasive diagnosis??	Frequencies, % prevalence for Age (years) 50-64 65-74 75-84 85 + Total	Persons with AMD (bilateral cases): Prevalence of AMD: total persons 132 (9.6%) 176 (14.8%) 91 (18.8%) 29 (25.7%) 428 (13%)	It is difficult to assess what is going on in this study. Unclear what methods other than ophthalmic records to diagnose AMD
Munier e <i>t al.</i> (1998) ¹⁰⁰	Prospective Irish Republic 5002	100% representative of those legally registered as blind in lreland at the time of study. 0.2% of total Irish population More than half of the patients were 65+ years and over, one-third were over 80 years	Legal registration of blindness VA of 6/60 (0.1) or worse (WHO definition of severe visual impairment)	% of those registered as blind Age (years) 16–34 35–54 55–64 65–79 80+ Total	caused by macular degeneration: Macular degeneration as cause of blindness 2.3% of 732 patients 3.1% of 1007 patients 4.5% of 457 patients 19.7% of 1350 patients 19.7% of 5002 patients 16.2% of 5002 patients	Question: when does it become age-related?
						continued

Comments	Details about the sample population are published elsewhere	
9	nent: went: y cause (non-correctable) <20/40 <20/70 <20/200 y cause (non-correctable) <20/40 <20/70 <20/200 y cause (non-correctable) 26 (19.4%) 27 (32.1%) 24 (54.5%) ecific primary attributable cause of incident unilateral visual ment worse than 20/40: y cause: Age Age Age 80+ All aged y cause: Age 60-69 70-79 years 49+ years years years years years et an 20/40 and uses 12 46 63 13 (23.8%) 3 (23.1%) 26 (19.4%) auses 1 (8.3%) 7 (15.2%) 15 (23.8%) 3 (23.1%) 26 (19.4%) y causes of incident unilateral visual impairment worse than 20/40 and y causes of incident unilateral visual impairment worse than 20/40 and y causes of incident unilateral visual impairment worse than 20/40 and y causes 11 (13.6%) 4 (12.5%) 11 (52.4%) 26 (19.4%) auses 81 32 21 134	
Diagnostic test Resul Confirmation of disease Exclusions	LogMAR, Beaver Dam classification impair ARM ARM Prima Dest c best c ARM ARM ARM ARM ARM	
Subjects y Age range (years) AMD Gender Ethnicity	All non- institutionalised, permanent residents 50+ years Females 57.5%, mean age 64.5 years BMES2 BMES2	
Study design Duration of stud) Location Total number	Prospective Blue Mountains, west of Sydney, Australia 2335 2335	
Study	Foran et <i>al</i> . (2002) ¹⁰¹	

		<u> </u>
Comments	At 10-year follow-up (1998–2000) 2962 participated in examination. 59.8% of participants in initial examination Figure 2 in the article shows the relation of age to log odds of ARM for different birth cohorts in the Beaver Dam Eye study	continued
	2000: either eye Age effect 8 45.89 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0001 <0.0000000000	
	th, BDES 1988-2 Late ARM in Birth cohort effect 3.72 0.81 0.81 0.81 1.58 t Age t Age t Late 1.58 2 2 2 2	
	t effect on AF a either eye Age effect 10 41.85 <0.0001 41.85 <0.0001 and and and and and and and and and and	
	or birth cohor Early ARM ir Birth cohort effect 8 22.25 0.005 0.005 0.005 1 lin either eye (9	
Results	Age-cohort model f Likelihood ratio statistic p-Value p-Value Heagerty and Zeger 1998-2000: 1998-2000: Prevalence of ARM i Baseline examination 5-year follow-up 10-year follow-up	
Diagnostic test Confirmation of disease Exclusions	similar procedures were used at baseline and ollow-up examinations Reported alsewhere WARMGS 80° colour fundus onth eyes: onth eyes: orteliminary and detailed grading drusen size, type and severity) Definitions: soft drusen, early and ate ARM	
Subjects Age range (years) AMD Gender Ethnicity	September September Drivate census to dentify residents of Baaver Dam 43–84 years old 1	
Study design Duration of study Location Total number	Prospective population-based 5- and 10-year follow-up Wisconsin, USA 4926	
Study	Huang et <i>al.</i> (2003) ¹⁰²	

Comments		continued
	nd and Wales 0 for males and 1990 F M F 283 2142 4438 466 5085 8481 ales: adjusting ales: adjusting 739 2142 4438	
	blindness in Engla % AMD % AMD 6 14 20 25 24 38 49 1970 1980 1970 1980 1970 1980 1970 1980 1970 1980 1970 1980 1970 1980 1970 1980 21 8 7 1487 1332 21	
	w registrations for gland: for AMD and all re 1960 M F M 1083 1824 114 4338 7343 401 1950–90 gistrations analyse 1960 M F M 1117 1880 116	
Results	Proportion of annual neattributed to AMD in Erattributed to AMD in E	
Diagnostic test) Confirmation of disease Exclusions	Sorsby pre-1980 Post-1980 ICD-9. Macular lesions description in 1970s corresponds to AMD	
Subjects Age range (years) AMD Gender Ethnicity	Registered as blind in England Causes of registered blindness: AMD, cataract, glaucoma and optic atrophy Data taken from the Office of Population Censuses and Surveys Age up to and including 85 + years	
Study design Duration of study Location Total number	Census records review Data abstracted for the years 1950, 1960, 1970, 1980 and the standard year 1990 UK 13,950 records examined in 1990	
Study	Evans and Wormald (1996) ⁸¹	

 $\textcircled{\sc c}$ Queen's Printer and Controller of HMSO 2008. All rights reserved.

Study	Study design Duration of stud) Location Total number	Subjects Y Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results							Comments
Gibson, et <i>al.</i> (1985) ¹⁰³	Prospective Random Melton Mowbray, Leics 1329	75+ years on 31 December 1980 6 on the GP register: (follow-up still living c in area 1982 6 147 male, 337 female. Mean age: 81.7 years	Ophthalmological examinations (optician and ophthalmologist). Criteria Framingham study	Prevalenc Age group (years) _e <85 85+ Total of all ages	ie of senile No. Me xamined 132 (f	macular de n Positive in one or both eyes srevalence, %) 54 (40.9%) 6 (40%)	seneration Wo. No. 260 77	n by age and men Positive in one or both eyes (prevalence, %) 98 (37.7%) 43 (55.8%)	sex: Both No. 392 92 484	l sexes Positive in one or both eyes (prevalence, %) 152 (38.8%) 49 (53.3%) 201 (41.5%)	677 persons available for examination, 484 were actually seen 2 methods were used to calculate prevalence rates, to allow for any discrepancies between the attenders and non- attenders
Cruickshanks, et al. (1997) ¹⁰⁴	Data from 2 population-based studies SLVDS (<i>n</i> = 1541) and Beaver Dam (<i>n</i> = 3999)	Mean age: BDES 58.7, SLV (NHW) 54, SLV Hispanics 53.2 years E 7 7 7 7	Colour stereoscopic fundus photographs. Graded using WARMGS Soth studies used similar methods. The same graders were used in both studies	Prevalenc San Luis v (Wisconsi Age (years) (years) 60-74	e of any A ralley (Colt in) San L (non-His) Male Fr 67 (3.5%) (6 187 (7.9%) (6 135 (14.9%) (1· Age, gende	RM by geog prado) com uis Valley panic whites) emale Both 92 159 159%) (6%%) 193 380 59%) (7.2%) 136 271 136 271 137 272 137 272 1772 1772 1772 1772 1772 1772 1772	graphic loc pared with San (H 71 (10.6%) (149 (10.6%) (104 (12.1%) (104 (15.8%) (2, walence (%))	ation, ethnic Luis Valley lispanics) -emale Both 88 159 (6.7%) 183 332 7.8%) (9.8%) 135 239 23.9%) (20.4%) :	ity, age an' n Eye Stuc Bea' (non-His Male F 982 (12.2%) ({ 843 (19.9%) (2	d gender: ly er Dam panic whites) emale Both 0 0 0 1115 2097 1115 2097 1115 2097 0.3%) (10.3%) 0.3%) (20.1%)	Low rate of AMD in Hispanics – ?due to Iow age limit (74) and may not survive to develop AMD
											continued

Comments	For details on early ARM, see Table 5 in article [weighted prevalence (95% CI)	of early ANY in either eye]. For comparison of the ARM prevalence	with other population- based studies, see	Table 7 in article						continued
	5 0	Eemale 0	0 0	0.20 (0 to 0.58)	1.6 (0.41 to 2.7)	7.0 (1.6 to 12.4)	20.0 (0 to 45.1)	0.76 (0.17 to 1.35)	8	
	eye based o AM	Male	0 0	0.58 (0.02 to 1.1)	1.9 (0.65 to 3.2)	2.0 (0 to 5.6)	58.6 (0 to 120.1)	0.58 (0.29 to 0.88)	0.6 (0.30 tc	
	is in either o c AMD v)	Female	, o	0	0.70 (0 to 1.6)	3.8 (0.56 to 7.1)	20.0 (0 to 45.1)	0.35 (0 to 0.70)	27 5 0.50)	
	AMD lesion Atrophi (dr	Male	0 0	0.23 (0 to 0.68)	0.27 (0 to 0.79)	0	50 (0 to 1)	0.17 (0 to 0.39)	0.0 (0.04 to	
	5% Cl) of <i>I</i> ular ARM et)	Female	0 0	0.20 (0 to 0.58)	0.86 (0 to 1.8)	4.7 (0.54 to 8.8)	0	0.38 (0.02 to 0.73)	39 0.58)	
	valence (95 grading: Neovasci (w	Male	0	0.35 (0 to 0.79)	1.7 (0.37 to 2.9)	.6 (0 to 4.5)	8.6 (0 to 27.8)	0.41 (0.16 to 0.67)	0 (0.20 to	
Results	Weighted pre photographic Age (years)	,	50–59	6069	70-79	8089	+06	Age- standardised rate	Age- standardised rate (both sexes)	
Diagnostic test Confirmation of disease Exclusions	VARMGS was sllowed with iinor nodifications	MD was graded s wet or dry								
Subjects Age range (years) AMD Gender Ethnicity	Mean age V 60.2 years. 55% fc female, 45% male. m Jrban and rural m	population A including NH as residents								
Study design Duration of study Location Total number	Cluster stratified random sampling Victoria, Australia	2 <u>+</u> 10								
Study	VanNewkirk et al. (2000) ¹⁰⁵									

Comments	Comparison between US and WHO definitions	continued
	cording the criteria 20/200) 95% Cl 0.0 to 1.6 0.3 to 3.8 0.3 to 1.4 0.3 to 1.4 0.3 to 1.4 0.3 to 1.4 1.4 41.4%) 44.4%)	
	d by AMD ac AMD US (20/40-: (20/40-: (20/40-: 3 (1.3) 5 (2) 9 (0.9) 0.82 8 (8 (8 (
	ter eye cause O criteria 0.0 to 1.6 0.0 to 1.5 0.1 to 3.1 0.2 to 1.2 0.2 to 1.2 0.2 to 1.2 0.2 to 1.2 0.2 to 1.2 1.7 0.2 to 1.2 0.4 (50%) 4 (50%)	
	airment in bet AMD WH (20/60-: (20/60-: 0 0 0 0 0.69 0.69 0.69 0.69 (40%) (40%) (40%)	
Results	Prevalence of visual impa WHO and US criteria: Age No. of (years) patients Under the age in the next row 249 75–80 249 All 944 Age adjusted % 249 All 944 Age (years) 1 60–69 0 70–80 4 (
Diagnostic test Confirmation of disease Exclusions	Visual impairment defined by the WHO (20/60–20/400) and the most common criteria in the USA (worse than 20/40 but better than 20/200). Amsler chart, slit-lamp examination, direct ophthalmoscopy and biomicroscopy (Goldmanns 3 lenses) and colour fundus photographs	
Subjects Age range (years) AMD Gender Ethnicity	Age- and gender- stratified subjects. 125 men and 125 women randomly selected from each age group (60–64, 65–69, 70–74, 75–80 years) Mean age of participants was 69.8 years (SD ± 5.7) 481 women, 463 male participants male participants the mean age of subjects with visual impairment (US definition) was 74.7 years	
Study design Duration of study Location Total number	Prospective cross-sectional study Copenhagen, 946	
Study	Buch et <i>al.</i> (2001) ¹⁰⁷	

Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results						Comments	
Wang et <i>al.</i> (2000) ¹⁰⁸	Prospective 2 years Blue Mountains, Sydney, Australia 3654	elsewhere ented ed < s = r	WARMGS. Any visual impairment defined as 6/12 or worse. Moderate visual impairment 5/24-6/60 (the definition used in the Beaver Dam Eye Study) and severe 6/60 or worse (Australian definition of blindness)	ARM as cause of impairment: Age (years) ii (to (to (to (to (to (to 0.69 co.69 co.69 co.69 co.79 co	of mild and r Bilateral visu Mild mpairment atal <i>n</i> = 122) i 1 (20%) ow 0 7 (17%) 1 3 (20%)	noderate to lal impairme Modera sever impairm (total <i>n</i> = 0 0 1 (25 8 (80°	severe b ant U te- in eent (toi * 45) %) %) %)	lateral and i nilateral visi Mild npairment al <i>n</i> = 287) 5 (8.9%) 14 (11%) 16 (18%)	unilateral visua ual impairment Moderate- severe impairment (total <i>n</i> = 23 2 (13%) 2 (13%) 19 (28%) 19 (28%) 38 (37%)	BMES participa a slightly higher economic statu a higher propo gaining post-scl qualifications at slightly greater 8) proportion ow their own hom non-participant	nts had socio- s, with trion nool a d a fin s s than s
Mitchell <i>et al.</i> (1995) ¹⁰⁹	Reported elsewhere	Reported elsewhere	As above	Prevalence of <i>A</i> 1992–3: Age (years) Under the age in the next r 55–64 65–74 75–84 85 + Total	ARM (wet or Female 0.3 0.3 0.3 0.3 2.4 2.4 2.4 2.4 2.4 2.4 2.4 2.4 2.4 2.4	- dry) by age at risk	e and sex Male 0 0 1.3 1.3	in the BMES at risk	, Australia, Total % No. at ri: 0.2 1172 0.7 1209 5.4 653 8.5 135 1.9 3654	The study look specifically at ir in risk with age table includes l eve studies. Als prevalence of c in worst eye	ed icrease ate and irusen
										5	ntinued

		_
Comments	This study suffers from the obvious sources of bias and confounding in the sample collection. It is most likely to identify patients with the most advanced AMD Table 4 in the article provides further information for criterion levels 1–3 (CPT codes required for ARM ascertainment)	
	exudative AMD in the Medicare vith Wet ARM (incidence, %) Females Both sexes 222 (0.12) 381 (0.12) % CI: 0.10 to 0.14 95% CI: 0.11 to 0.13 % CI: 0.10 to 0.14 95% CI: 0.11 to 0.13 % CI: 0.22 to 0.27 95% CI: 0.21 to 0.25 % CI: 0.22 to 0.37 95% CI: 0.31 to 0.38 % CI: 0.29 to 0.37 95% CI: 0.31 to 0.38 % CI: 0.22 to 0.37 95% CI: 0.31 to 0.38 % CI: 0.29 to 0.37 95% CI: 0.31 to 0.38 % CI: 0.22 to 0.31 95% CI: 0.31 to 0.38 % CI: 0.22 to 0.31 95% CI: 0.31 to 0.38 % CI: 0.22 to 0.31 95% CI: 0.23 to 0.32 % CI: 0.23 to 0.35 95% CI: 0.22 to 0.31 % CI: 0.23 to 0.25 95% CI: 0.22 to 0.23 % CI: 0.23 to 0.25 95% CI: 0.22 to 0.24	
lesults	hree-year estimated incidence of opulation: ge Laser: no. w Jnder the 159 (0.12) age in 95% CI: 0.1 to 0.13 95 the next 237 (0.21) row 237 (0.21) 0-74 237 (0.21) 95% CI: 0.19, 0.24 95 5-79 244 (0.35) 95% CI: 0.30 to 0.39 95 0-84 130 (0.39) 95 5+ 43 (0.30) 95 6tal 813 (0.22) to 0.24 95 6tal 813 (0.22) to 0.24 95 6tal 813 (0.22) to 0.24 95	
Diagnostic test R Confirmation of disease Exclusions	Wet AMD T ascertained using ascertained using ascertained using diagnosis and aliagnosis and contained in the contained in the claims files (laser, levels 1, 2 and 3) Analysed the surgical and laser procedure claims submitted by the proportion of a AMD defined as the proportion of the proportion of the scriteria to ne of the scriteria to ne of the scriteria to	
Subjects Age range (years) AMD Gender Ethnicity	S-years Medicare beneficiaries' data develop develop ongitudinal cohort incidence measured Age 65 + years	
Study design Duration of study Location Total number	Prospective Population-based USA I,041,009	
Study	Javitt et al. (2003) ¹¹⁰	

Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results							Comments
Klein <i>et al.</i> (2002) ¹¹	Prospective Population-based cohort 10-year incidence Beaver Dam, Wisconsin 5924 eligible, 4926 at baseline examinations (83%). Of the survivors, 81.1% participated in 5-year follow-up and of these survivors 82.9% in 10-year follow-up	13 - 84 years	MARMGS used to grade stereoscopic colour fundus photographs	Relation of age and late ARM Age (years) 43–54 55–64 65–74 75 + Total Age (years) Age (years) 43–54 55–64 65–74 75 + Total	in the Beavarian dender Incidence Men Men 21.7 28.7 10.6 21.7 28.7 10.6 Men Men 7.8 7.8 7.8	er Dam E er Dam E Women 24.8 40.8 10.7 24.8 40.8 13 3.7 10.7 24.8 40.8 13 3.7 24.5 6.6 19.7 34.5 9.8	cipant, to th ye Study: Both Both sexes 36.7 12.1 10.7 23.6 36.7 12.1 12.1 12.1 12.1 12.7 80th sexes 80th sexes 8.9 8.9 8.9	e 10-year hrcidenc Men Men Men Men 1.3 1.4 1.1 1.3 1.3 1.3 1.3 1.3 1.3 1.3	incidence e of late A Women 0.1 0.6 1.3 2.4 2.4 2.4 0 0.6 0.6 9.1 1.6	of early Both 80th 80th 80th 80th 80th 80th 80th 80	82.9% of survivors in 10-years follow-up. 2764 final participants: 56% of all original participants Table 2 in the article: both wet and dry numbers and drusen. Also Table 3, progression regression and disappearance of drusen
											continued

	_	_		_			_	_	_	_	_	_				_		_	_	 	_
Comments																					continued
			ent	Total	70 (100)	246 (100) 316 (100)				AMD	(0) 0		0 (0)	l (1.6)	6 (17.3)	1 (10) 8 (3.3)	in one or				
		e and race:	All causes of al impairme	Blacks	48 (100)	2 (00) 60 (00)		and race:	Blacks	subjects	31		66	513	146	00	as present				
		ment by ag	Avisu	Whites	22 (100)	34 (100) 56 (100)		MD by age		No. of	9		9	9	- m	1	ied cause w				
		f visual impair	: caused	Total	2 (2.9)	43 (17.5) 45 (14.2)	cted	t caused by A	es l	AMD	(0) 0		I (I.6)	I (I.I)	7 (11.1)	11 (53.4) 20 (6.9)	1000). Specifi				
		as cause of	impairment by AMD	Blacks	(0) 0	10 (8.9) 10 (6.3)	of eyes affec	impairment	White	f subjects	542		618	915	630 22	206 119	alence per				
sults		istribution of ARM	ge Visual ears)	Whites	nder the 2 (9.1) age in the	next row 65 33 (24.6) tal 35 (22.4)	llues are no. (%) o	evalence of visual	ge (years)	No. o	nder the age	age in the next row)59)69)–79	80 ital 2	ilues are no. (prev oth eyes				
Diagnostic test R Confirmation of disease Exclusions		'isual impairment D	efined as BCVA A vorse than 20/40 (y	0/200 (different	rom international Uriterion but the ame as the	riteria used by ⇒ nost states to ⊼	or driver	censing) Pi	∢				2	6		M 14	ف لا ا				
Subjects Age range (years) AMD Gender	Ethnicity	v sample of 16 V	luster areas was d elected using a v	cheme designed 2	o obtained fr pproximately c cual numbers of s	lack and white c esidents n	911 blacks, 2389 hites														
Study design Duration of study Location Total number		Sample of 16 🖌	cluster areas c East Baltimore S	5300 s	n d	, <u></u> c															
Study		Rahmani et al.	(1996) ¹¹²																		

Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results				Comments
Bressler <i>et al.</i> (1995) ⁷	Follow-up study 483 Watermen on eastern shore of Maryland	30 + years. Men 7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	-undus ohotographs Eyes in which AMD appeared or disappeared were graded in a side- yy-side fashion. The standardised grading of the undus bhotographs has been described sleewhere	Appearance or di: Age range Pa in 1985 (years) (years) (years) (years) (years) (years) (years) (years) (years) (years) (years) (years) Under the age in the next row 40.49 50.59 60.69 70+ Total AMD level 3: eye: the RPE or non-G	sappearance of AMC intricipants with no Al in 1985 fotal no. No. (%) AMD-3 ir AMD-3 ir 126 9 (7) 95 7 (7) 78 11 (1) 78 11 (1) 78 11 (1) 78 11 (1) 78 11 (1) 78 13 (8) 31 8 (2) 31 8 (2) 31 8 (2) 31 8 (2) 31 8 (2) 31 8 (2) 31 8 (2) 31 8 (2) 413 38 (9) 5 with large or confluence or confluen		nts with AMD-3 in 1985 No. (%) lost AMD-3 in 1990 2 (22) 5 (56) 1 (17) 7 (32) 2 (13) 1 7 (28) arpigmentation of al centre	Only 483 (70%) showed the appearance or disappearance of AMD. Limitation: study composed of only men and biased towards a certain coccupation (watermen). Also, almost half of the population were under 50 years of age
Farber (2003) ¹¹³	Prospective Consecutive From the registry for the blind Israel 41,402	48.6% of the F registered blind E were male and E 51.4% female. 84.3% of blind were aged over 140 years and 21.7% over 80 years (Registered as blind: identified as blind by an pphthalmologist - VA is ≤0.05 (20/400), slightly ess rigid than MHO definition (<0.05)	Number of perso AMD as cause of Age (years) 0-4 5-24 55-74 75 + Total	ns (%) diagnosed as blindness: prevalenci AMD caus 0 122 797 1665 2584	blind from 1987 to 19 e data: e of blindness (4.7%) (30.8%) (64.4%) (14%)	99 by age and Total blind 217 (1.2%) 1258 (6.8%) 3876 (21%) 5914 (32%) 18465 (100%)	
								continued

Number of persons (%) now/ diagonad as blind in 1999 by age and AMD scattee of blindness. Includence atta AMD cause of blindness. Includence atta Re found in 1999 by age and AMD scatter of blindness. Includence atta AMD cause of blindness. Includence atta 44 (8.03, 9(.3, 74)) Scatter of Diagness. Includence rates, by AMD diagness, per 100,000 population, 1999. 5-4 (40, 603, 34), 32.040 (1056), 32.040 (1066), 30.040 (1066), 30.040 (1066), 32.040 (1066	Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results			Comments
Chang et dl.ProspectiveCanadian of Vancouver,Reviewed the Underwent fluoresceinAMD diagnosis for patients of Chinese and European ancestry who underwent fluoresceinAND diagnosis for patients of Chinese and European ancestry who underwent fluoresceinAND diagnosis for patients of Chinese and European ancestry who underwent fluoresceinAlso table by age with regard to with regard to avanticipation rates at anticipation rates at and re-evaluated for primaryAMD diagnosis Interpreted by AMDAMD diagnosis for patients of Chinese and European ancestry who (14.6%) of patientsAlso table by age with regard to with regard to and rates at and rate valuated for primary for primary for primaryAMD among patients of Chinese and European ancestry who underwent fluorescein angiography December 1987-November 1993: Type of AMDAMD Anong patients of Chinese and European ancestry who anticipation rates at for primary for primary for primary for primaryAMD Among patients of Chinese and European ancestry who anticipation rates at for primary 					Number of persons (⁹ as cause of blindness: Age (years) 0-4 5-24 55-74 55-74 75+ Total Blindness incidence ra Age (years) 25-54 55-74 75+ 75+ 75+ 75+ 75+	%) newly diagnosed as blind in I incidence data: AMD cause of blindness 0 99 (19.7%) 404 (80.3%) 503 (20.1%) fes, by AMD diagnosis, per 100, AMD as cause of blindness 12.9 162.7 8.2	999 by age and AMD Total blind 44 (1.8%) 183 (7.3%) 810 (16.3%) 983 (35.4%) 983 (39.2%) 000 population, 1999:	
	Chang et al. (1999) ⁶	Prospective Vancouver, Canada 20,000	Canadian of Chinese ancestry a a a d d d d	keviewed the luorescein ingiograms nterpreted by etinal specialist and re-evaluated or primary liagnosis	AMD diagnosis for par underwent fluoresceit Diagnosis AMD Type of AMD among I Underwent fluoresceit Type of AMD Wet ^a Active choroidal ^b JF, EF and SF, CNVM	tients of Chinese and European 3 angiography December 1987-5 Ancestry: no. (%) of Chinese (n = 417) E 61 (14.6%) 127 61 (14.6%) 127 7 (14.6%) 127 7 (127) 127 8 (14.6%) 127 61 (14.6%) 127 61 (13.8%) 39 (29.2%) 39 (29.2%) 39 (29.2%) 30 (29.2%) 30 (29.2%)	ancestry who September 1990: patients uropean $(n = 417)$ (30.4%), p < 0.001 n ancestry who Vovember 1993: patients patients uropean $(n = 149)$ 94 (63.1%) 55 (36.9%) fisciform scar	Also table by age Methodology is vague with regard to participation rates and exclusion criteria (except for ethnicity)

ments	definitions: Iness, Snellen VA 60: partial sight, 3/60. For further ils see article. Two nitions of AMD: vascular/wet 0) and GAMD 0) and GAMD 0) and GAMD 0)	continued
S	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	
Results	Meta-analysis of GAMD, NVAMConditionConditionGAMD0.60(5 studies)0.61(0.35 to (0.35 to (0.35 to 0))(15 studies)0.61(2 studies)0.61(2 studies)0.61(2 studies)0.15(2 studies)0.15(2 studies)0.15(3 studies)0.15(4 studies)0.15(5 studies)0.15(5 studies)0.03 to (0.03 to 0)(6 studies)0.03 to 0(7 studies)0.01(1 studies)0.01(1 studies)0.01(1 studies)0.01(1 studies)0(1 studies)0(2 studies)0(1 studies)0(1 studies)0(1 studies)0(1 studies)0(1 studies)0(1 studies)0(1 studies)0(1 studies)0<	
Diagnostic test) Confirmation of disease Exclusions	Systematic review was conducted using published papers, abstracts, letters and review articles from the usual sources. 673 references yielded from search, 27 identified as having potential potential relevance; 19 studies were chosen (based on geographically defined, random sample populations, published from 1990 onwards) and the authors written to asking to provide data	
Subjects Age range (years) AMD Gender Ethnicity	Pooled data from 6 studies. Excluded: studies including volunteers or specific professions due to possible selection biases Age groups: < 50, 50–90 (in 5-year bands), and 90+ years	
Study design Duration of study Location Total number	Systematic review followed by a request for data used to establish a pooled prevalence for AMD 6 studies provided data: Blue Mountains Eye Study North London Eye Study North London Eye Study Melbourne Visual Impairment Project	
Study	Owen et al. (2003) ⁴	

Study	Study design Duration of study Location Total number	Subjects Age range (years) AMD Gender Ethnicity	Diagnostic test Confirmation of disease Exclusions	Results									Comments	
				Age range	No. of I	S	No. of P	S/B	No. of	8	No. o	ž		_
				(years)	2001	5011	2001	5011	2001	2011	2001	2011		
				80-84	25 25 (16 to (36)	27 27 18 to (6 39)	 to 9) (6	12 to 21) (6	 to 9) (6	12 5 to 21)	47 (35 to 61)	5 (38 to 67)		
				8589	29 (18 to (33 21 to (2 50)	7 : to 15) (2	8 to 17)	25 (15 to 39)	29 (17 to 44)	61 (45 to 80)	69 (51 to 91)		
				+06	28 (15 to (46)	33 18 to (3 54)	 to 24) (4	13 to 29) (26 (13 to 44)	30 (16 to 51)	64 (44 to 88)	75 (52 to 104)		
				Total	109 (65 to (187)	2 72 to 208)	34 (13 to (92)	38 5 to 03)	71 38 to 140)	80 (43 to 157)	214 (151 to 310)	239 (168 to 346)		
				PS, partial si impairment. on prevalen Only one of predictions	ghted cau See Tabl ce of GA the six st is questio	Ised by e 4 in jo MD and tudies to nable	AMD; PS urnal art NVAMD ook place	/B, part icle for in the L	ial/blind definitio JK, so tl	B, blind ns and fi he validi	d; VI, visi urther d ty of the	ual etails sse		
BC, best correcte Practitioner Com WARMGS, Wisco	ed; BDES, best correc mittee; NH, nursing h nsin Age-Related Mac	:ted eye sight; BES, be nome; NHW, Non-His :ulopathy Grading Syst	st corrected sight; (panic White; SD, st :em; WHO, World I	CNVM, chore andard deviat Health Organ	oidal neov cion; SE, s ization.	/ascular tandard	membra error; S	ne; CPT LV, San I	, Currer uis Vall	it Proce ey; SLVE	dural Te DS, San I	rminolog) -uis Valley	r, FPC, Family r Diabetes Study;	

Appendix 3

Expert elicitation questionnaire

Expert survey to inform the evaluation of a screening programme for AMD

We are building a mathematical model to estimate the cost-effectiveness of a screening programme for AMD. To ensure that the defined screening programme is appropriate, and to overcome some limitations in the data available to populate the model, we are keen to incorporate the views of experts in the field.

The following survey is being distributed to six consultant ophthalmologists and will greatly assist our evaluation. It can either be completed over the telephone with a member of the research team (Carolyn Murray), and is expected to require 30 minutes of your time, or if you prefer we are happy to correspond via email or via the Royal Mail. Your contribution will be acknowledged in subsequent publications, which will include an NHS Health Technology Assessment monograph.

Screening

The following paragraphs lay out a possible screening process. Each paragraph is followed by questions regarding the appropriateness of the defined process.

- 1. A screening test for AMD is defined as providing the general population aged over 50 years with an annual reminder to self-test for AMD by covering one eye, and then the other eye. If they notice an abnormality (visual distortion) in either eye, they are instructed to present at an opticians.
 - Q1.1 How would you describe 'an abnormality' in the context of the population-based screening test, i.e. how would you describe an abnormality in the reminder letter?

Q1.2 We are assuming that the Amsler grid is an INAPPROPRIATE screening tool for the general population? Do you agree, and are there any other approaches that could be used?

Q1.3 Using the 'eye cover' test described above, or any other form of self-test, what proportion of people with the following types of AMD do you think would notice 'an abnormality' when self-testing?

AMD type	Proportion noticing visual distortion after self-completing a cover test	Proportion noticing visual distortion after self-completing another specified test (if any)
No AMD		
Drusen		
Dry AMD		
Extrafoveal CNV		
Juxtafoveal CNV		
Subfoveal CNV		

- 2. In the proposed screening programme, persons who notice an abnormality are instructed to present at an opticians. Opticians are instructed to examine the back of the eye of presenting patients, and if a macular haemorrhage is detected (indicating suspected CNV), the optician is instructed to refer the patient on to an ophthalmologist.
 - Q2.1 Is macular haemorrhage an appropriate indicator of CNV? If not, what other indicators could be used by opticians to identify suspected CNV?

Q2.2 What proportion of people with the following types of AMD do you think would be referred on to a consultant ophthalmologist on the basis of opticians detecting macular haemorrhage (or any specified alternative indicators)?

AMD type	Proportion referred to consultant o	phthalmologist by opticians
	Macular haemorrhage	Other indicator of CNV
No AMD		
Drusen		
Dry AMD		
Extrafoveal CNV		
Juxtafoveal CNV		
Subfoveal CNV		

Q2.2 As consultant ophthalmologists working for the NHS, do you only want to see patients with suspected CNV?

If not referred to a consultant ophthalmologist, how should patients with dry AMD (and no CNV) be treated? Should they be referred to an LVU or monitored by an optician? Should they be advised about diet and lifestyle changes, possible preventative interventions, and told to monitor their eyesight and present again at the opticians upon deterioration of eyesight?

Q2.3 Is a similar process to that for dry AMD recommended for patients presenting with drusen, but no AMD?

Treatment interventions

1. Treatment for CNV

We are describing VA scores over time by patient sub-group based on the patient's age and type of lesion at diagnosis (size – under or over 4 MPS DA; type – occult, minimally classic, or predominantly classic; location – extra-, juxta-, or subfoveal).

We have re-analysed the patient-level data from the TAP and VIP trials to estimate VA profiles for all subfoveal lesions.

Q1.1 What other interventions should we consider for subfoveal lesions?

We have identified studies reporting the use of laser photocoagulation, and PDT, in juxtafoveal lesions, with the PDT study (Frennesson¹¹⁴) stating that PDT is more effective than laser photocoagulation.

Q1.2 What treatment options should we consider for juxtafoveal lesions? Do treatment options differ by lesion size and type?

The only identified treatment for extrafoveal lesions is laser photocoagulation.

Q1.3 Would you treat all extrafoveal lesions with laser photocoagulation? If not, how would you treat extrafoveal AMD?

2. Treatment for dry AMD and drusen (ARM)

We want to model progression from dry AMD and ARM to CNV, as well as to describe VA scores over time for patients with dry AMD.

Q2.1 Are you aware of any data sources describing VA in patients with dry AMD who do not develop CNV? If not, could you estimate an average VA for patients presenting with dry AMD?

Q2.2 How quickly does VA deteriorate in patients with dry AMD who do not develop CNV, for example, what is the average VA loss over a year?

Q2.3 Are there any feasible treatments for dry AMD and ARM? For example, are IRIS Medical[®] OcuLight[®] Infrared Laser Photocoagulator, or the Rheopheresis[®] blood filtration procedure, likely to be treatment options now or in the future?

Q2.4 Are you hopeful that any interventions aimed at preventing progression to CNV will become available? If so, how effective do you think they might be, i.e. what is your estimate of the proportion of cases of CNV they could prevent?

Could the same level of effectiveness be applied to eyes with no AMD and those with early ARM, and to eyes whose fellow eyes have CNV and those whose fellow eyes have no CNV?

Many thanks for your time and opinions.

Appendix 4

Survey of ophthalmologists

ScHARR/PDT Users Group modelling of cost-effectiveness of AMD

Of 100 patients receiving their FIRST diagnosis of AMD or ARM, please estimate in your opinion how many a the following categories:	re in each of
Unilateral ARM	/100
Unilateral dry AMD	/100
Unilateral extrafoveal AMD	/100
Unilateral juxtafoveal AMD	/100
Unilateral subfoveal AMD	/100
Bilateral ARM	/100
Bilateral ARM/AMD, worst eye has dry AMD	/100
Bilateral ARM/AMD, worst eye has extrafoveal AMD	/100
Bilateral ARM/AMD, worst eye has juxtafoveal AMD	/100
Bilateral ARM/AMD, worst eye has subfoveal AMD	/100
Of 100 patients that you diagnose in the following categories, how many do you refer on to a PDT clinic:	
Unilateral juxtafoveal AMD	/100
Unilateral subfoveal AMD	/100
Bilateral ARM/AMD, worst eye has juxtafoveal AMD	/100
Bilateral ARM/AMD, worst eye has subfoveal AMD	/100

Appendix 5

National Screening Committee's criteria for appraising the viability, effectiveness and appropriateness of a screening programme

The criteria, which are set out below, are based on the classic criteria first promulgated in a WHO report in 1966, but take into account both the more rigorous standards of evidence required to improve effectiveness and the greater concern about the adverse effects of healthcare; regrettably, some people who undergo screening will suffer adverse effects without receiving benefit from the programme.

These criteria have been prepared taking into account international work on the appraisal of screening programmes, particularly that in Canada and the USA. It is recognised that not all of the criteria and questions raised in the format will be applicable to every proposed programme, but the more that are answered will obviously assist the NSC to make better evidence-based decisions.

All of the following criteria should be met before screening for a condition is initiated.

The condition

- 1. The condition should be an important health problem. YES.
- 2. The epidemiology and natural history of the condition, including development from latent to declared disease, should be adequately understood and there should be a detectable risk factor, or disease marker and a latent period or early symptomatic stage. THE NATURAL HISTORY IS POORLY QUANTIFIED, FIRST EYE DISEASE MAY BE DEFINED AS A LATENT PERIOD IN SOME INDIVIDUALS.
- 3. All the cost-effective primary prevention interventions should have been implemented as far as practicable.

THERE ARE ONGOING TRIALS OF VITAMIN SUPPLEMENTS, ALTHOUGH THERE IS NO CONCLUSIVE EVIDENCE ON THEIR EFFECTIVENESS AT PRESENT.

The test

- 4. There should be a simple, safe, precise and validated screening test.
 IT IS SIMPLE AND SAFE, BUT NOT PROVEN TO BE PRECISE AND VALIDATED.
- 5. The distribution of test values in the target population should be known and a suitable cut-off level defined and agreed. NOT KNOWN.
- 6. The test should be acceptable to the population. NOT KNOWN, BUT PROBABLE.
- 7. There should be an agreed policy on the further diagnostic investigation of individuals with a positive test result and on the choices available to those individuals. THE DIAGNOSTIC PROCESS IS AGREED.

The treatment

8. There should be an effective treatment or intervention for patients identified through early detection, with evidence of early treatment leading to better outcomes than late treatment. THERE IS SOME EVIDENCE OF EARLY TREATMENT BEING BETTER, ALTHOUGH A RANGE OF NEW INTERVENTIONS MAY HAVE DIFFERENT PROFILES. LONG-TERM TREATMENT EFFECTIVENESS DATA ARE MISSING.

- 9. There should be agreed evidence-based policies covering which individuals should be offered treatment and the appropriate treatment to be offered. CURRENTLY UNDER DEBATE BY NICE.
- 10. Clinical management of the condition and patient outcomes should be optimised by all healthcare providers prior to participation in a screening programme. NOT APPLICABLE.

The screening programme

- 11. There must be evidence from high-quality RCTs that the screening programme is effective in reducing mortality or morbidity. Where screening is aimed solely at providing information to allow the person being screened to make an 'informed choice' (e.g. Down's syndrome, cystic fibrosis carrier screening), there must be evidence from highquality trials that the test accurately measures risk. The information that is provided about the test and its outcome must be of value and readily understood by the individual being screened. NOT AVAILABLE.
- 12. There should be evidence that the complete screening programme (test, diagnostic procedures, treatment/intervention) is clinically, socially and ethically acceptable to health professionals and the public. NOT AVAILABLE.
- 13. The benefit from the screening programme should outweigh the physical and psychological harm (caused by the test, diagnostic procedures and treatment). NOT AVAILABLE.
- 14. The opportunity cost of the screening programme (including testing, diagnosis,

treatment, administration, training and quality assurance) should be economically balanced in relation to expenditure on medical care as a whole (i.e. value for money). COST-EFFECTIVENESS NOT RELIABLY ESTABLISHED (I.E. HIGH UNCERTAINTY).

- 15. There must be a plan for managing and monitoring the screening programme and an agreed set of quality assurance standards. NOT APPLICABLE AT PRESENT.
- 16. Adequate staffing and facilities for testing, diagnosis, treatment and programme management should be made available prior to the commencement of the screening programme. NOT APPLICABLE AT PRESENT.
- 17. All other options for managing the condition should have been considered (e.g. improving treatment, providing other services), to ensure that no more cost-effective intervention could be introduced or current interventions increased within the resources available. TREATMENT OPTIONS CURRENTLY IN A STATE OF FLUX.
- Evidence-based information, explaining the consequences of testing, investigation and treatment, should be made available to potential participants to assist them in making an informed choice.
 NOT APPLICABLE AT PRESENT.
- 19. Public pressure for widening the eligibility criteria, for reducing the screening interval and for increasing the sensitivity of the testing process should be anticipated. Decisions about these parameters should be scientifically justifiable to the public. NOT APPLICABLE AT PRESENT.

Health Technology Assessment reports published to date

Volume 1, 1997

No. 1

Home parenteral nutrition: a systematic review.

By Richards DM, Deeks JJ, Sheldon TA, Shaffer JL.

No. 2

Diagnosis, management and screening of early localised prostate cancer. A review by Selley S, Donovan J,

Faulkner A, Coast J, Gillatt D.

No. 3

The diagnosis, management, treatment and costs of prostate cancer in England and Wales.

A review by Chamberlain J, Melia J, Moss S, Brown J.

No. 4

Screening for fragile X syndrome. A review by Murray J, Cuckle H, Taylor G, Hewison J.

No. 5

A review of near patient testing in primary care.

By Hobbs FDR, Delaney BC, Fitzmaurice DA, Wilson S, Hyde CJ, Thorpe GH, *et al*.

No. 6

Systematic review of outpatient services for chronic pain control.

By McQuay HJ, Moore RA, Eccleston C, Morley S, de C Williams AC.

No. 7

Neonatal screening for inborn errors of metabolism: cost, yield and outcome. A review by Pollitt RJ, Green A, McCabe CJ, Booth A, Cooper NJ, Leonard JV, *et al*.

No. 8

Preschool vision screening. A review by Snowdon SK, Stewart-Brown SL.

No. 9

Implications of socio-cultural contexts for the ethics of clinical trials.

A review by Ashcroft RE, Chadwick DW, Clark SRL, Edwards RHT, Frith L, Hutton JL.

No. 10

A critical review of the role of neonatal hearing screening in the detection of congenital hearing impairment.

By Davis A, Bamford J, Wilson I, Ramkalawan T, Forshaw M, Wright S.

No. 11

Newborn screening for inborn errors of metabolism: a systematic review.

By Seymour CA, Thomason MJ, Chalmers RA, Addison GM, Bain MD, Cockburn F, *et al*.

No. 12

Routine preoperative testing: a systematic review of the evidence. By Munro J, Booth A, Nicholl J.

No. 13

Systematic review of the effectiveness of laxatives in the elderly. By Petticrew M, Watt I, Sheldon T.

No. 14

When and how to assess fast-changing technologies: a comparative study of medical applications of four generic technologies.

A review by Mowatt G, Bower DJ, Brebner JA, Cairns JA, Grant AM, McKee L.

Volume 2, 1998

No. 1

Antenatal screening for Down's syndrome. A review by Wald NJ, Kennard A, Lacksherr A. McCuire A.

Hackshaw A, McGuire A.

No. 2

Screening for ovarian cancer: a systematic review. By Bell R, Petticrew M, Luengo S, Sheldon TA.

No. 3

Consensus development methods, and their use in clinical guideline development.

A review by Murphy MK, Black NA, Lamping DL, McKee CM, Sanderson CFB, Askham J, *et al*.

No. 4

A cost–utility analysis of interferon beta for multiple sclerosis. By Parkin D, McNamee P, Jacoby A, Miller P, Thomas S, Bates D.

No. 5

Effectiveness and efficiency of methods of dialysis therapy for end-stage renal disease: systematic reviews.

By MacLeod A, Grant A, Donaldson C, Khan I, Campbell M, Daly C, *et al*.

No. 6

Effectiveness of hip prostheses in primary total hip replacement: a critical review of evidence and an economic model.

By Faulkner A, Kennedy LG, Baxter K, Donovan J, Wilkinson M, Bevan G.

No. 7

Antimicrobial prophylaxis in colorectal surgery: a systematic review of randomised controlled trials. By Song F, Glenny AM.

No. 8

Bone marrow and peripheral blood stem cell transplantation for malignancy. A review by Johnson PWM, Simnett SJ, Sweetenham JW, Morgan GJ, Stewart LA.

No. 9

Screening for speech and language delay: a systematic review of the literature.

By Law J, Boyle J, Harris F, Harkness A, Nye C.

No. 10

Resource allocation for chronic stable angina: a systematic review of effectiveness, costs and cost-effectiveness of alternative interventions. By Sculpher MJ, Petticrew M, Kolland H, Elliott PA, Holdwicht DP

Kelland JL, Elliott RA, Holdright DR, Buxton MJ.

No. 11

Detection, adherence and control of hypertension for the prevention of stroke: a systematic review. By Ebrahim S.

No. 12

Postoperative analgesia and vomiting, with special reference to day-case surgery: a systematic review. By McQuay HJ, Moore RA.

No. 13

Choosing between randomised and nonrandomised studies: a systematic review.

By Britton A, McKee M, Black N, McPherson K, Sanderson C, Bain C.

No. 14

Evaluating patient-based outcome measures for use in clinical trials.

A review by Fitzpatrick R, Davey C, Buxton MJ, Jones DR.

Ethical issues in the design and conduct of randomised controlled trials.

A review by Edwards SJL, Lilford RJ, Braunholtz DA, Jackson JC, Hewison J, Thornton J.

No. 16

Qualitative research methods in health technology assessment: a review of the literature.

By Murphy E, Dingwall R, Greatbatch D, Parker S, Watson P.

No. 17

The costs and benefits of paramedic skills in pre-hospital trauma care. By Nicholl J, Hughes S, Dixon S, Turner J, Yates D.

No. 18

Systematic review of endoscopic ultrasound in gastro-oesophageal cancer

By Harris KM, Kelly S, Berry E, Hutton J, Roderick P, Cullingworth J, et al.

No. 19

Systematic reviews of trials and other studies.

By Sutton AJ, Abrams KR, Jones DR, Sheldon TA, Song F.

No. 20

Primary total hip replacement surgery: a systematic review of outcomes and modelling of cost-effectiveness associated with different prostheses.

A review by Fitzpatrick R, Shortall E, Sculpher M, Murray D, Morris R, Lodge M. et al.

Volume 3, 1999

No. 1

Informed decision making: an annotated bibliography and systematic review.

By Bekker H, Thornton JG, Airey CM, Connelly JB, Hewison J,

Robinson MB, et al.

No. 2

Handling uncertainty when performing economic evaluation of healthcare interventions.

A review by Briggs AH, Gray AM.

No. 3

The role of expectancies in the placebo effect and their use in the delivery of health care: a systematic review.

By Crow R, Gage H, Hampson S, Hart J, Kimber A, Thomas H.

No. 4

A randomised controlled trial of different approaches to universal antenatal HIV testing: uptake and acceptability. Annex: Antenatal HIV testing - assessment of a routine voluntary approach.

By Simpson WM, Johnstone FD, Boyd FM, Goldberg DJ, Hart GJ, Gormley SM, et al.

No. 5

Methods for evaluating area-wide and organisation-based interventions in health and health care: a systematic review

By Ukoumunne OC, Gulliford MC, Chinn S, Sterne JAC, Burney PGJ.

No. 6

Assessing the costs of healthcare technologies in clinical trials. A review by Johnston K, Buxton MJ, Jones DR, Fitzpatrick R.

No. 7

Cooperatives and their primary care emergency centres: organisation and impact.

By Hallam L, Henthorne K.

No. 8

Screening for cystic fibrosis. A review by Murray J, Cuckle H, Taylor G, Littlewood J, Hewison J.

No. 9

A review of the use of health status measures in economic evaluation. By Brazier J, Deverill M, Green C, Harper R, Booth A.

No. 10

Methods for the analysis of quality-oflife and survival data in health technology assessment.

A review by Billingham LJ, Abrams KR, Jones DR.

No. 11

Antenatal and neonatal haemoglobinopathy screening in the

UK: review and economic analysis. By Zeuner D, Ades AE, Karnon J,

Brown J, Dezateux C, Anionwu EN.

No. 12

Assessing the quality of reports of randomised trials: implications for the conduct of meta-analyses. A review by Moher D, Cook DJ, Jadad

AR, Tugwell P, Moher M, Jones A, et al.

No. 13

'Early warning systems' for identifying new healthcare technologies. By Robert G, Stevens A, Gabbay J.

No. 14

A systematic review of the role of human papillomavirus testing within a cervical screening programme.

By Cuzick J, Sasieni P, Davies P, Adams J, Normand C, Frater A, et al.

No. 15

Near patient testing in diabetes clinics: appraising the costs and outcomes. By Grieve R, Beech R, Vincent J, Mazurkiewicz J.

No. 16

Positron emission tomography: establishing priorities for health technology assessment.

A review by Robert G, Milne R.

No. 17 (Pt 1)

The debridement of chronic wounds: a systematic review.

By Bradley M, Cullum N, Sheldon T.

No. 17 (Pt 2)

Systematic reviews of wound care management: (2) Dressings and topical agents used in the healing of chronic wounds.

By Bradley M, Cullum N, Nelson EA, Petticrew M, Sheldon T, Torgerson D.

No. 18

A systematic literature review of spiral and electron beam computed tomography: with particular reference to clinical applications in hepatic lesions, pulmonary embolus and coronary artery disease.

By Berry E, Kelly S, Hutton J, Harris KM, Roderick P, Boyce JC, et al.

No. 19

What role for statins? A review and economic model.

By Ebrahim S, Davey Smith G, McCabe C, Payne N, Pickin M, Sheldon TA. et al.

No. 20

Factors that limit the quality, number and progress of randomised controlled trials.

A review by Prescott RJ, Counsell CE, Gillespie WJ, Grant AM, Russell IT, Kiauka S, et al.

No. 21

Antimicrobial prophylaxis in total hip replacement: a systematic review. By Glenny AM, Song F.

No. 22

Health promoting schools and health promotion in schools: two systematic reviews.

By Lister-Sharp D, Chapman S, Stewart-Brown S. Sowden A.

No. 23

Economic evaluation of a primary carebased education programme for patients with osteoarthritis of the knee. A review by Lord J, Victor C,

Littlejohns P, Ross FM, Axford JS.

Volume 4, 2000

No. 1

The estimation of marginal time preference in a UK-wide sample (TEMPUS) project.

A review by Cairns JA, van der Pol MM.

No. 2

Geriatric rehabilitation following fractures in older people: a systematic review.

By Cameron I, Crotty M, Currie C, Finnegan T, Gillespie L, Gillespie W, et al.



Screening for sickle cell disease and thalassaemia: a systematic review with supplementary research. By Davies SC, Cronin E, Gill M,

Greengross P, Hickman M, Normand C.

No. 4

Community provision of hearing aids and related audiology services. A review by Reeves DJ, Alborz A, Hickson FS, Bamford JM.

No. 5

False-negative results in screening programmes: systematic review of impact and implications.

By Petticrew MP, Sowden AJ, Lister-Sharp D, Wright K.

No. 6

Costs and benefits of community postnatal support workers: a randomised controlled trial. By Morrell CJ, Spiby H, Stewart P, Walters S, Morgan A.

No. 7

Implantable contraceptives (subdermal implants and hormonally impregnated intrauterine systems) versus other forms of reversible contraceptives: two systematic reviews to assess relative effectiveness, acceptability, tolerability and cost-effectiveness.

By French RS, Cowan FM, Mansour DJA, Morris S, Procter T, Hughes D, *et al.*

No. 8

An introduction to statistical methods for health technology assessment. A review by White SJ, Ashby D,

Brown PJ.

No. 9

Disease-modifying drugs for multiple sclerosis: a rapid and systematic review.

By Clegg A, Bryant J, Milne R.

No. 10

Publication and related biases. A review by Song F, Eastwood AJ, Gilbody S, Duley L, Sutton AJ.

No. 11

Cost and outcome implications of the organisation of vascular services. By Michaels J, Brazier J, Palfreyman S, Shackley P, Slack R.

No. 12

Monitoring blood glucose control in diabetes mellitus: a systematic review. By Coster S, Gulliford MC, Seed PT, Powrie JK, Swaminathan R.

No. 13

The effectiveness of domiciliary health visiting: a systematic review of international studies and a selective review of the British literature.

By Elkan R, Kendrick D, Hewitt M, Robinson JJA, Tolley K, Blair M, *et al.*

No. 14

The determinants of screening uptake and interventions for increasing uptake: a systematic review.

By Jepson R, Clegg A, Forbes C, Lewis R, Sowden A, Kleijnen J.

No. 15

The effectiveness and cost-effectiveness of prophylactic removal of wisdom teeth.

A rapid review by Song F, O'Meara S, Wilson P, Golder S, Kleijnen J.

No. 16

Ultrasound screening in pregnancy: a systematic review of the clinical effectiveness, cost-effectiveness and women's views.

By Bricker L, Garcia J, Henderson J, Mugford M, Neilson J, Roberts T, et al.

No. 17

A rapid and systematic review of the effectiveness and cost-effectiveness of the taxanes used in the treatment of advanced breast and ovarian cancer. By Lister-Sharp D, McDonagh MS, Khan KS, Kleijnen J.

No. 18

Liquid-based cytology in cervical screening: a rapid and systematic review. By Payne N, Chilcott J, McGoogan E.

No. 19

Randomised controlled trial of nondirective counselling, cognitive-behaviour therapy and usual general practitioner care in the management of depression as well as mixed anxiety and depression in primary care.

By King M, Sibbald B, Ward E, Bower P, Lloyd M, Gabbay M, *et al*.

No. 20

Routine referral for radiography of patients presenting with low back pain: is patients' outcome influenced by GPs' referral for plain radiography?

By Kerry S, Hilton S, Patel S, Dundas D, Rink E, Lord J.

No. 21

Systematic reviews of wound care management: (3) antimicrobial agents for chronic wounds; (4) diabetic foot ulceration.

By O'Meara S, Cullum N, Majid M, Sheldon T.

No. 22

Using routine data to complement and enhance the results of randomised controlled trials.

By Lewsey JD, Leyland AH, Murray GD, Boddy FA.

No. 23

Coronary artery stents in the treatment of ischaemic heart disease: a rapid and systematic review.

By Meads C, Cummins C, Jolly K, Stevens A, Burls A, Hyde C.

No. 24

Outcome measures for adult critical care: a systematic review. By Hayes JA, Black NA, Jenkinson C, Young JD, Rowan KM, Daly K, *et al.*

No. 25

A systematic review to evaluate the effectiveness of interventions to promote the initiation of breastfeeding.

By Fairbank L, O'Meara S, Renfrew MJ, Woolridge M, Sowden AJ, Lister-Sharp D.

No. 26

Implantable cardioverter defibrillators: arrhythmias. A rapid and systematic review.

By Parkes J, Bryant J, Milne R.

No. 27

Treatments for fatigue in multiple sclerosis: a rapid and systematic review.

By Brañas P, Jordan R, Fry-Smith A, Burls A, Hyde C.

No. 28

Early asthma prophylaxis, natural history, skeletal development and economy (EASE): a pilot randomised controlled trial.

By Baxter-Jones ADG, Helms PJ, Russell G, Grant A, Ross S, Cairns JA, *et al.*

No. 29

Screening for hypercholesterolaemia versus case finding for familial hypercholesterolaemia: a systematic review and cost-effectiveness analysis.

By Marks D, Wonderling D, Thorogood M, Lambert H, Humphries SE, Neil HAW.

No. 30

A rapid and systematic review of the clinical effectiveness and costeffectiveness of glycoprotein IIb/IIIa antagonists in the medical management of unstable angina.

By McDonagh MS, Bachmann LM, Golder S, Kleijnen J, ter Riet G.

No. 31

A randomised controlled trial of prehospital intravenous fluid replacement therapy in serious trauma. By Turner J, Nicholl J, Webber L, Cox H, Dixon S, Yates D.

No. 32

Intrathecal pumps for giving opioids in chronic pain: a systematic review. By Williams JE, Louw G, Towlerton G.

No. 33

Combination therapy (interferon alfa and ribavirin) in the treatment of chronic hepatitis C: a rapid and systematic review.

By Shepherd J, Waugh N, Hewitson P.

A systematic review of comparisons of effect sizes derived from randomised and non-randomised studies.

By MacLehose RR, Reeves BC, Harvey IM, Sheldon TA, Russell IT, Black AMS.

No. 35

Intravascular ultrasound-guided interventions in coronary artery disease: a systematic literature review, with decision-analytic modelling, of outcomes and cost-effectiveness.

By Berry E, Kelly S, Hutton J, Lindsay HSJ, Blaxill JM, Evans JA, et al.

No. 36

A randomised controlled trial to evaluate the effectiveness and costeffectiveness of counselling patients with chronic depression.

By Simpson S, Corney R, Fitzgerald P, Beecham J.

No. 37

Systematic review of treatments for atopic eczema.

By Hoare C, Li Wan Po A, Williams H.

No. 38

Bayesian methods in health technology assessment: a review. By Spiegelhalter DJ, Myles JP,

Jones DR, Abrams KR.

No. 39

The management of dyspepsia: a systematic review. By Delaney B, Moayyedi P, Deeks J, Innes M, Soo S, Barton P, *et al.*

No. 40

A systematic review of treatments for severe psoriasis. By Griffiths CEM, Clark CM, Chalmers RJG, Li Wan Po A, Williams HC.

Volume 5, 2001

No. 1

Clinical and cost-effectiveness of donepezil, rivastigmine and galantamine for Alzheimer's disease: a rapid and systematic review.

By Clegg A, Bryant J, Nicholson T, McIntyre L, De Broe S, Gerard K, *et al.*

No. 2

The clinical effectiveness and costeffectiveness of riluzole for motor neurone disease: a rapid and systematic review.

By Stewart A, Sandercock J, Bryan S, Hyde C, Barton PM, Fry-Smith A, *et al*.

No. 3

Equity and the economic evaluation of healthcare.

By Sassi F, Archard L, Le Grand J.

No. 4 Quality-of-life measures in chronic

diseases of childhood. By Eiser C, Morse R.

No. 5

Eliciting public preferences for healthcare: a systematic review of techniques.

By Ryan M, Scott DA, Reeves C, Bate A, van Teijlingen ER, Russell EM, *et al*.

No. 6

General health status measures for people with cognitive impairment: learning disability and acquired brain injury.

By Riemsma RP, Forbes CA, Glanville JM, Eastwood AJ, Kleijnen J.

No. 7

An assessment of screening strategies for fragile X syndrome in the UK. By Pembrey ME, Barnicoat AJ,

Carmichael B, Bobrow M, Turner G.

No. 8

Issues in methodological research: perspectives from researchers and

commissioners.

By Lilford RJ, Richardson A, Stevens A, Fitzpatrick R, Edwards S, Rock F, et al.

No. 9

Systematic reviews of wound care management: (5) beds; (6) compression; (7) laser therapy, therapeutic ultrasound, electrotherapy and electromagnetic therapy. By Cullum N, Nelson EA, Flemming K, Sheldon T.

No. 10

Effects of educational and psychosocial interventions for adolescents with diabetes mellitus: a systematic review. By Hampson SE, Skinner TC, Hart J,

Storey L, Gage H, Foxcroft D, *et al*.

No. 11

Effectiveness of autologous chondrocyte transplantation for hyaline cartilage defects in knees: a rapid and systematic review.

By Jobanputra P, Parry D, Fry-Smith A, Burls A.

No. 12

Statistical assessment of the learning curves of health technologies.

By Ramsay CR, Grant AM, Wallace SA, Garthwaite PH, Monk AF, Russell IT.

No. 13

The effectiveness and cost-effectiveness of temozolomide for the treatment of recurrent malignant glioma: a rapid and systematic review.

By Dinnes J, Cave C, Huang S, Major K, Milne R.

No. 14

A rapid and systematic review of the clinical effectiveness and costeffectiveness of debriding agents in treating surgical wounds healing by secondary intention.

By Lewis R, Whiting P, ter Riet G, O'Meara S, Glanville J.

No. 15

Home treatment for mental health problems: a systematic review. By Burns T, Knapp M, Catty J, Healey A, Henderson J, Watt H, *et al.*

No. 16

How to develop cost-conscious guidelines. By Eccles M, Mason J.

, , ,

No. 17

The role of specialist nurses in multiple sclerosis: a rapid and systematic review. By De Broe S, Christopher F, Waugh N.

No. 18

A rapid and systematic review of the clinical effectiveness and costeffectiveness of orlistat in the management of obesity. By O'Meara S, Riemsma R,

Shirran L, Mather L, ter Riet G.

No. 19

The clinical effectiveness and costeffectiveness of pioglitazone for type 2 diabetes mellitus: a rapid and systematic review.

By Chilcott J, Wight J, Lloyd Jones M, Tappenden P.

No. 20

Extended scope of nursing practice: a multicentre randomised controlled trial of appropriately trained nurses and preregistration house officers in pre-operative assessment in elective general surgery.

By Kinley H, Czoski-Murray C, George S, McCabe C, Primrose J, Reilly C, *et al*.

No. 21

Systematic reviews of the effectiveness of day care for people with severe mental disorders: (1) Acute day hospital versus admission; (2) Vocational rehabilitation; (3) Day hospital versus outpatient care.

By Marshall M, Crowther R, Almaraz-Serrano A, Creed F, Sledge W, Kluiter H, *et al.*

No. 22

The measurement and monitoring of surgical adverse events.

By Bruce J, Russell EM, Mollison J, Krukowski ZH.

No. 23

Action research: a systematic review and guidance for assessment. By Waterman H, Tillen D, Dickson R,

de Koning K.

No. 24

A rapid and systematic review of the clinical effectiveness and costeffectiveness of gemcitabine for the treatment of pancreatic cancer.

By Ward S, Morris E, Bansback N, Calvert N, Crellin A, Forman D, et al.

A rapid and systematic review of the evidence for the clinical effectiveness and cost-effectiveness of irinotecan, oxaliplatin and raltitrexed for the treatment of advanced colorectal cancer.

By Lloyd Jones M, Hummel S, Bansback N, Orr B, Seymour M.

No. 26

Comparison of the effectiveness of inhaler devices in asthma and chronic obstructive airways disease: a systematic review of the literature.

By Brocklebank D, Ram F, Wright J, Barry P, Cates C, Davies L, *et al*.

No. 27

The cost-effectiveness of magnetic resonance imaging for investigation of the knee joint.

By Bryan S, Weatherburn G, Bungay H, Hatrick C, Salas C, Parry D, *et al.*

No. 28

A rapid and systematic review of the clinical effectiveness and costeffectiveness of topotecan for ovarian cancer.

By Forbes C, Shirran L, Bagnall A-M, Duffy S, ter Riet G.

No. 29

Superseded by a report published in a later volume.

No. 30

The role of radiography in primary care patients with low back pain of at least 6 weeks duration: a randomised (unblinded) controlled trial.

By Kendrick D, Fielding K, Bentley E, Miller P, Kerslake R, Pringle M.

No. 31

Design and use of questionnaires: a review of best practice applicable to surveys of health service staff and patients.

By McColl E, Jacoby A, Thomas L, Soutter J, Bamford C, Steen N, *et al*.

No. 32

A rapid and systematic review of the clinical effectiveness and costeffectiveness of paclitaxel, docetaxel, gemcitabine and vinorelbine in nonsmall-cell lung cancer.

By Clegg A, Scott DA, Sidhu M, Hewitson P, Waugh N.

No. 33

Subgroup analyses in randomised controlled trials: quantifying the risks of false-positives and false-negatives.

By Brookes ST, Whitley E, Peters TJ, Mulheran PA, Egger M, Davey Smith G.

No. 34

Depot antipsychotic medication in the treatment of patients with schizophrenia: (1) Meta-review; (2) Patient and nurse attitudes.

By David AS, Adams C.

No. 35

A systematic review of controlled trials of the effectiveness and costeffectiveness of brief psychological treatments for depression.

By Churchill R, Hunot V, Corney R, Knapp M, McGuire H, Tylee A, *et al*.

No. 36

Cost analysis of child health surveillance.

By Sanderson D, Wright D, Acton C, Duree D.

Volume 6, 2002

No. 1

A study of the methods used to select review criteria for clinical audit.

By Hearnshaw H, Harker R, Cheater F, Baker R, Grimshaw G.

No. 2

Fludarabine as second-line therapy for B cell chronic lymphocytic leukaemia: a technology assessment.

By Hyde C, Wake B, Bryan S, Barton P, Fry-Smith A, Davenport C, *et al*.

No. 3

Rituximab as third-line treatment for refractory or recurrent Stage III or IV follicular non-Hodgkin's lymphoma: a systematic review and economic evaluation.

By Wake B, Hyde C, Bryan S, Barton P, Song F, Fry-Smith A, *et al*.

No. 4

A systematic review of discharge arrangements for older people.

By Parker SG, Peet SM, McPherson A, Cannaby AM, Baker R, Wilson A, *et al.*

No. 5

The clinical effectiveness and costeffectiveness of inhaler devices used in the routine management of chronic asthma in older children: a systematic review and economic evaluation.

By Peters J, Stevenson M, Beverley C, Lim J, Smith S.

No. 6

The clinical effectiveness and costeffectiveness of sibutramine in the management of obesity: a technology assessment.

By O'Meara S, Riemsma R, Shirran L, Mather L, ter Riet G.

No. 7

The cost-effectiveness of magnetic resonance angiography for carotid artery stenosis and peripheral vascular disease: a systematic review.

By Berry E, Kelly S, Westwood ME, Davies LM, Gough MJ, Bamford JM, *et al.*

No. 8

Promoting physical activity in South Asian Muslim women through 'exercise on prescription'. By Carroll B, Ali N, Azam N. No. 9

Zanamivir for the treatment of influenza in adults: a systematic review and economic evaluation. By Burls A, Clark W, Stewart T, Preston C, Bryan S, Jefferson T, *et al*.

No. 10

A review of the natural history and epidemiology of multiple sclerosis: implications for resource allocation and health economic models. By Richards RG, Sampson FC,

Beard SM, Tappenden P.

No. 11

Screening for gestational diabetes: a systematic review and economic evaluation.

By Scott DA, Loveman E, McIntyre L, Waugh N.

No. 12

The clinical effectiveness and costeffectiveness of surgery for people with morbid obesity: a systematic review and economic evaluation.

By Clegg AJ, Colquitt J, Sidhu MK, Royle P, Loveman E, Walker A.

No. 13

The clinical effectiveness of trastuzumab for breast cancer: a systematic review. By Lewis R, Bagnall A-M, Forbes C, Shirran E, Duffy S, Kleijnen J, *et al.*

No. 14

The clinical effectiveness and costeffectiveness of vinorelbine for breast cancer: a systematic review and economic evaluation.

By Lewis R, Bagnall A-M, King S, Woolacott N, Forbes C, Shirran L, et al.

No. 15

A systematic review of the effectiveness and cost-effectiveness of metal-on-metal hip resurfacing arthroplasty for treatment of hip disease.

By Vale L, Wyness L, McCormack K, McKenzie L, Brazzelli M, Stearns SC.

No. 16

The clinical effectiveness and costeffectiveness of bupropion and nicotine replacement therapy for smoking cessation: a systematic review and economic evaluation.

By Woolacott NF, Jones L, Forbes CA, Mather LC, Sowden AJ, Song FJ, et al.

No. 17

A systematic review of effectiveness and economic evaluation of new drug treatments for juvenile idiopathic arthritis: etanercept.

By Cummins C, Connock M, Fry-Smith A, Burls A.

No. 18

Clinical effectiveness and costeffectiveness of growth hormone in children: a systematic review and economic evaluation.

By Bryant J, Cave C, Mihaylova B, Chase D, McIntyre L, Gerard K, *et al.*

Clinical effectiveness and costeffectiveness of growth hormone in adults in relation to impact on quality of life: a systematic review and economic evaluation.

By Bryant J, Loveman E, Chase D, Mihaylova B, Cave C, Gerard K, *et al.*

No. 20

Clinical medication review by a pharmacist of patients on repeat prescriptions in general practice: a randomised controlled trial.

By Zermansky AG, Petty DR, Raynor DK, Lowe CJ, Freementle N, Vail A.

No. 21

The effectiveness of infliximab and etanercept for the treatment of rheumatoid arthritis: a systematic review and economic evaluation.

By Jobanputra P, Barton P, Bryan S, Burls A.

No. 22

A systematic review and economic evaluation of computerised cognitive behaviour therapy for depression and anxiety.

By Kaltenthaler E, Shackley P, Stevens K, Beverley C, Parry G, Chilcott J.

No. 23

A systematic review and economic evaluation of pegylated liposomal doxorubicin hydrochloride for ovarian cancer.

By Forbes C, Wilby J, Richardson G, Sculpher M, Mather L, Reimsma R.

No. 24

A systematic review of the effectiveness of interventions based on a stages-ofchange approach to promote individual behaviour change.

By Riemsma RP, Pattenden J, Bridle C, Sowden AJ, Mather L, Watt IS, *et al*.

No. 25

A systematic review update of the clinical effectiveness and costeffectiveness of glycoprotein IIb/IIIa antagonists.

By Robinson M, Ginnelly L, Sculpher M, Jones L, Riemsma R, Palmer S, et al.

No. 26

A systematic review of the effectiveness, cost-effectiveness and barriers to implementation of thrombolytic and neuroprotective therapy for acute ischaemic stroke in the NHS.

By Sandercock P, Berge E, Dennis M, Forbes J, Hand P, Kwan J, *et al.*

No. 27

A randomised controlled crossover trial of nurse practitioner versus doctor-led outpatient care in a bronchiectasis clinic.

By Caine N, Sharples LD, Hollingworth W, French J, Keogan M, Exley A, *et al*.

No. 28

Clinical effectiveness and cost – consequences of selective serotonin reuptake inhibitors in the treatment of sex offenders.

By Adi Y, Ashcroft D, Browne K, Beech A, Fry-Smith A, Hyde C.

No. 29

Treatment of established osteoporosis: a systematic review and cost–utility analysis.

By Kanis JA, Brazier JE, Stevenson M, Calvert NW, Lloyd Jones M.

No. 30

Which anaesthetic agents are costeffective in day surgery? Literature review, national survey of practice and randomised controlled trial.

By Elliott RA Payne K, Moore JK, Davies LM, Harper NJN, St Leger AS, *et al.*

No. 31

Screening for hepatitis C among injecting drug users and in genitourinary medicine clinics: systematic reviews of effectiveness, modelling study and national survey of current practice.

By Stein K, Dalziel K, Walker A, McIntyre L, Jenkins B, Horne J, et al.

No. 32

The measurement of satisfaction with healthcare: implications for practice from a systematic review of the literature.

By Crow R, Gage H, Hampson S, Hart J, Kimber A, Storey L, *et al*.

No. 33

The effectiveness and cost-effectiveness of imatinib in chronic myeloid leukaemia: a systematic review.

By Garside R, Round A, Dalziel K, Stein K, Royle R.

No. 34

A comparative study of hypertonic saline, daily and alternate-day rhDNase in children with cystic fibrosis.

By Suri R, Wallis C, Bush A, Thompson S, Normand C, Flather M, *et al.*

No. 35

A systematic review of the costs and effectiveness of different models of paediatric home care.

By Parker G, Bhakta P, Lovett CA, Paisley S, Olsen R, Turner D, et al.

Volume 7, 2003

No. 1

How important are comprehensive literature searches and the assessment of trial quality in systematic reviews? Empirical study.

By Egger M, Jüni P, Bartlett C, Holenstein F, Sterne J.

No. 2

Systematic review of the effectiveness and cost-effectiveness, and economic evaluation, of home versus hospital or satellite unit haemodialysis for people with end-stage renal failure.

By Mowatt G, Vale L, Perez J, Wyness L, Fraser C, MacLeod A, *et al*.

No. 3

Systematic review and economic evaluation of the effectiveness of infliximab for the treatment of Crohn's disease.

By Clark W, Raftery J, Barton P, Song F, Fry-Smith A, Burls A.

No. 4

A review of the clinical effectiveness and cost-effectiveness of routine anti-D prophylaxis for pregnant women who are rhesus negative.

By Chilcott J, Lloyd Jones M, Wight J, Forman K, Wray J, Beverley C, et al.

No. 5

Systematic review and evaluation of the use of tumour markers in paediatric oncology: Ewing's sarcoma and neuroblastoma.

By Riley RD, Burchill SA, Abrams KR, Heney D, Lambert PC, Jones DR, *et al*.

No. 6

The cost-effectiveness of screening for *Helicobacter pylori* to reduce mortality and morbidity from gastric cancer and peptic ulcer disease: a discrete-event simulation model.

By Roderick P, Davies R, Raftery J, Crabbe D, Pearce R, Bhandari P, *et al.*

No. 7

The clinical effectiveness and costeffectiveness of routine dental checks: a systematic review and economic evaluation.

By Davenport C, Elley K, Salas C, Taylor-Weetman CL, Fry-Smith A, Bryan S, *et al*.

No. 8

A multicentre randomised controlled trial assessing the costs and benefits of using structured information and analysis of women's preferences in the management of menorrhagia.

By Kennedy ADM, Sculpher MJ, Coulter A, Dwyer N, Rees M, Horsley S, *et al*.

No. 9

Clinical effectiveness and cost–utility of photodynamic therapy for wet age-related macular degeneration: a systematic review and economic evaluation.

By Meads C, Salas C, Roberts T, Moore D, Fry-Smith A, Hyde C.

No. 10

Evaluation of molecular tests for prenatal diagnosis of chromosome abnormalities.

By Grimshaw GM, Szczepura A, Hultén M, MacDonald F, Nevin NC, Sutton F, *et al*.



First and second trimester antenatal screening for Down's syndrome: the results of the Serum, Urine and Ultrasound Screening Study (SURUSS).

By Wald NJ, Rodeck C, Hackshaw AK, Walters J, Chitty L, Mackinson AM.

No. 12

The effectiveness and cost-effectiveness of ultrasound locating devices for central venous access: a systematic review and economic evaluation.

By Calvert N, Hind D, McWilliams RG, Thomas SM, Beverley C, Davidson A.

No. 13

A systematic review of atypical antipsychotics in schizophrenia. By Bagnall A-M, Jones L, Lewis R, Ginnelly L, Glanville J, Torgerson D, *et al.*

No. 14

Prostate Testing for Cancer and Treatment (ProtecT) feasibility study.

By Donovan J, Hamdy F, Neal D, Peters T, Oliver S, Brindle L, *et al*.

No. 15

Early thrombolysis for the treatment of acute myocardial infarction: a systematic review and economic evaluation.

By Boland A, Dundar Y, Bagust A, Haycox A, Hill R, Mujica Mota R, *et al.*

No. 16

Screening for fragile X syndrome: a literature review and modelling. By Song FJ, Barton P, Sleightholme V, Yao GL, Fry-Smith A.

No. 17

Systematic review of endoscopic sinus surgery for nasal polyps. By Dalziel K, Stein K, Round A, Garside R, Royle P.

our side K

No. 18

Towards efficient guidelines: how to monitor guideline use in primary care.

By Hutchinson A, McIntosh A, Cox S, Gilbert C.

No. 19

Effectiveness and cost-effectiveness of acute hospital-based spinal cord injuries services: systematic review. By Bagnall A-M, Jones L,

Richardson G, Duffy S, Riemsma R.

No. 20

Harper G.

Prioritisation of health technology assessment. The PATHS model: methods and case studies. By Townsend J, Buxton M,

No. 21

Systematic review of the clinical effectiveness and cost-effectiveness of tension-free vaginal tape for treatment of urinary stress incontinence.

By Cody J, Wyness L, Wallace S, Glazener C, Kilonzo M, Stearns S, *et al.*

No. 22

The clinical and cost-effectiveness of patient education models for diabetes: a systematic review and economic evaluation.

By Loveman E, Cave C, Green C, Royle P, Dunn N, Waugh N.

No. 23

The role of modelling in prioritising and planning clinical trials.

By Chilcott J, Brennan A, Booth A, Karnon J, Tappenden P.

No. 24

Cost-benefit evaluation of routine influenza immunisation in people 65–74 years of age.

By Allsup S, Gosney M, Haycox A, Regan M.

No. 25

The clinical and cost-effectiveness of pulsatile machine perfusion versus cold storage of kidneys for transplantation retrieved from heart-beating and nonheart-beating donors.

By Wight J, Chilcott J, Holmes M, Brewer N.

No. 26

Can randomised trials rely on existing electronic data? A feasibility study to explore the value of routine data in health technology assessment.

By Williams JG, Cheung WY, Cohen DR, Hutchings HA, Longo MF, Russell IT.

No. 27

Evaluating non-randomised intervention studies.

By Deeks JJ, Dinnes J, D'Amico R, Sowden AJ, Sakarovitch C, Song F, *et al*.

No. 28

A randomised controlled trial to assess the impact of a package comprising a patient-orientated, evidence-based selfhelp guidebook and patient-centred consultations on disease management and satisfaction in inflammatory bowel disease.

By Kennedy A, Nelson E, Reeves D, Richardson G, Roberts C, Robinson A, *et al.*

No. 29

The effectiveness of diagnostic tests for the assessment of shoulder pain due to soft tissue disorders: a systematic review.

By Dinnes J, Loveman E, McIntyre L, Waugh N.

No. 30

The value of digital imaging in diabetic retinopathy.

By Sharp PF, Olson J, Strachan F, Hipwell J, Ludbrook A, O'Donnell M, *et al.*

No. 31

Lowering blood pressure to prevent myocardial infarction and stroke: a new preventive strategy.

By Law M, Wald N, Morris J.

No. 32

Clinical and cost-effectiveness of capecitabine and tegafur with uracil for the treatment of metastatic colorectal cancer: systematic review and economic evaluation.

By Ward S, Kaltenthaler E, Cowan J, Brewer N.

No. 33

Clinical and cost-effectiveness of new and emerging technologies for early localised prostate cancer: a systematic review.

By Hummel S, Paisley S, Morgan A, Currie E, Brewer N.

No. 34

Literature searching for clinical and cost-effectiveness studies used in health technology assessment reports carried out for the National Institute for Clinical Excellence appraisal system.

By Royle P, Waugh N.

No. 35

Systematic review and economic decision modelling for the prevention and treatment of influenza A and B.

By Turner D, Wailoo A, Nicholson K, Cooper N, Sutton A, Abrams K.

No. 36

A randomised controlled trial to evaluate the clinical and costeffectiveness of Hickman line insertions in adult cancer patients by nurses.

By Boland A, Haycox A, Bagust A, Fitzsimmons L.

No. 37

Redesigning postnatal care: a randomised controlled trial of protocol-based midwifery-led care focused on individual women's physical and psychological health needs.

By MacArthur C, Winter HR, Bick DE, Lilford RJ, Lancashire RJ, Knowles H, *et al*.

No. 38

Grimley Evans J.

Estimating implied rates of discount in healthcare decision-making. By West RR, McNabb R, Thompson AGH, Sheldon TA,

Systematic review of isolation policies in the hospital management of methicillinresistant *Staphylococcus aureus*: a review of the literature with epidemiological and economic modelling.

By Cooper BS, Stone SP, Kibbler CC, Cookson BD, Roberts JA, Medley GF, *et al.*

No. 40

Treatments for spasticity and pain in multiple sclerosis: a systematic review. By Beard S, Hunn A, Wight J.

No. 41

The inclusion of reports of randomised trials published in languages other than English in systematic reviews.

By Moher D, Pham B, Lawson ML, Klassen TP.

No. 42

The impact of screening on future health-promoting behaviours and health beliefs: a systematic review.

By Bankhead CR, Brett J, Bukach C, Webster P, Stewart-Brown S, Munafo M, *et al.*

Volume 8, 2004

No. 1

What is the best imaging strategy for acute stroke?

By Wardlaw JM, Keir SL, Seymour J, Lewis S, Sandercock PAG, Dennis MS, *et al.*

No. 2

Systematic review and modelling of the investigation of acute and chronic chest pain presenting in primary care.

By Mant J, McManus RJ, Oakes RAL, Delaney BC, Barton PM, Deeks JJ, et al.

No. 3

The effectiveness and cost-effectiveness of microwave and thermal balloon endometrial ablation for heavy menstrual bleeding: a systematic review and economic modelling.

By Garside R, Stein K, Wyatt K, Round A, Price A.

No. 4

A systematic review of the role of bisphosphonates in metastatic disease.

By Ross JR, Saunders Y, Edmonds PM, Patel S, Wonderling D, Normand C, et al.

No. 5

Systematic review of the clinical effectiveness and cost-effectiveness of capecitabine (Xeloda[®]) for locally advanced and/or metastatic breast cancer.

By Jones L, Hawkins N, Westwood M, Wright K, Richardson G, Riemsma R.

No. 6

Effectiveness and efficiency of guideline dissemination and implementation strategies.

By Grimshaw JM, Thomas RE, MacLennan G, Fraser C, Ramsay CR, Vale L, *et al.*

No. 7

Clinical effectiveness and costs of the Sugarbaker procedure for the treatment of pseudomyxoma peritonei.

By Bryant J, Clegg AJ, Sidhu MK, Brodin H, Royle P, Davidson P.

No. 8

Psychological treatment for insomnia in the regulation of long-term hypnotic drug use.

By Morgan K, Dixon S, Mathers N, Thompson J, Tomeny M.

No. 9

Improving the evaluation of therapeutic interventions in multiple sclerosis: development of a patient-based measure of outcome.

By Hobart JC, Riazi A, Lamping DL, Fitzpatrick R, Thompson AJ.

No. 10

A systematic review and economic evaluation of magnetic resonance cholangiopancreatography compared with diagnostic endoscopic retrograde cholangiopancreatography.

By Kaltenthaler E, Bravo Vergel Y, Chilcott J, Thomas S, Blakeborough T, Walters SJ, *et al*.

No. 11

The use of modelling to evaluate new drugs for patients with a chronic condition: the case of antibodies against tumour necrosis factor in rheumatoid arthritis.

By Barton P, Jobanputra P, Wilson J, Bryan S, Burls A.

No. 12

Clinical effectiveness and costeffectiveness of neonatal screening for inborn errors of metabolism using tandem mass spectrometry: a systematic review.

By Pandor A, Eastham J, Beverley C, Chilcott J, Paisley S.

No. 13

Clinical effectiveness and costeffectiveness of pioglitazone and rosiglitazone in the treatment of type 2 diabetes: a systematic review and economic

evaluation.

By Czoski-Murray C, Warren E, Chilcott J, Beverley C, Psyllaki MA, Cowan J.

No. 14

Routine examination of the newborn: the EMREN study. Evaluation of an extension of the midwife role including a randomised controlled trial of appropriately trained midwives and paediatric senior house officers.

By Townsend J, Wolke D, Hayes J, Davé S, Rogers C, Bloomfield L, *et al.*

No. 15

Involving consumers in research and development agenda setting for the NHS: developing an evidence-based approach.

By Oliver S, Clarke-Jones L, Rees R, Milne R, Buchanan P, Gabbay J, *et al.*

No. 16

A multi-centre randomised controlled trial of minimally invasive direct coronary bypass grafting versus percutaneous transluminal coronary angioplasty with stenting for proximal stenosis of the left anterior descending coronary artery.

By Reeves BC, Angelini GD, Bryan AJ, Taylor FC, Cripps T, Spyt TJ, et al.

No. 17

Does early magnetic resonance imaging influence management or improve outcome in patients referred to secondary care with low back pain? A pragmatic randomised controlled trial.

By Gilbert FJ, Grant AM, Gillan MGC, Vale L, Scott NW, Campbell MK, *et al.*

No. 18

The clinical and cost-effectiveness of anakinra for the treatment of rheumatoid arthritis in adults: a

systematic review and economic analysis. By Clark W, Jobanputra P, Barton P, Burls A.

No. 19

A rapid and systematic review and economic evaluation of the clinical and cost-effectiveness of newer drugs for treatment of mania associated with bipolar affective disorder.

By Bridle C, Palmer S, Bagnall A-M, Darba J, Duffy S, Sculpher M, *et al*.

No. 20

Liquid-based cytology in cervical screening: an updated rapid and systematic review and economic analysis.

By Karnon J, Peters J, Platt J, Chilcott J, McGoogan E, Brewer N.

No. 21

Systematic review of the long-term effects and economic consequences of treatments for obesity and implications for health improvement.

By Avenell A, Broom J, Brown TJ, Poobalan A, Aucott L, Stearns SC, *et al.*

No. 22

Autoantibody testing in children with newly diagnosed type 1 diabetes mellitus.

By Dretzke J, Cummins C, Sandercock J, Fry-Smith A, Barrett T, Burls A.


Clinical effectiveness and costeffectiveness of prehospital intravenous fluids in trauma patients. By Dretzke J, Sandercock J, Bayliss S,

Burls A.

No. 24

Newer hypnotic drugs for the shortterm management of insomnia: a systematic review and economic evaluation.

By Dündar Y, Boland A, Strobl J, Dodd S, Haycox A, Bagust A, *et al.*

No. 25

Development and validation of methods for assessing the quality of diagnostic accuracy studies.

By Whiting P, Rutjes AWS, Dinnes J, Reitsma JB, Bossuyt PMM, Kleijnen J.

No. 26

EVALUATE hysterectomy trial: a multicentre randomised trial comparing abdominal, vaginal and laparoscopic methods of hysterectomy.

By Garry R, Fountain J, Brown J, Manca A, Mason S, Sculpher M, *et al*.

No. 27

Methods for expected value of information analysis in complex health economic models: developments on the health economics of interferon- β and glatiramer acetate for multiple sclerosis.

By Tappenden P, Chilcott JB, Eggington S, Oakley J, McCabe C.

No. 28

Effectiveness and cost-effectiveness of imatinib for first-line treatment of chronic myeloid leukaemia in chronic phase: a systematic review and economic analysis.

By Dalziel K, Round A, Stein K, Garside R, Price A.

No. 29

VenUS I: a randomised controlled trial of two types of bandage for treating venous leg ulcers.

By Iglesias C, Nelson EA, Cullum NA, Torgerson DJ on behalf of the VenUS Team.

No. 30

Systematic review of the effectiveness and cost-effectiveness, and economic evaluation, of myocardial perfusion scintigraphy for the diagnosis and management of angina and myocardial infarction.

By Mowatt G, Vale L, Brazzelli M, Hernandez R, Murray A, Scott N, *et al.*

No. 31

A pilot study on the use of decision theory and value of information analysis as part of the NHS Health Technology Assessment programme.

By Claxton K, Ginnelly L, Sculpher M, Philips Z, Palmer S.

No. 32

The Social Support and Family Health Study: a randomised controlled trial and economic evaluation of two alternative forms of postnatal support for mothers living in disadvantaged inner-city areas.

By Wiggins M, Oakley A, Roberts I, Turner H, Rajan L, Austerberry H, et al.

No. 33

Psychosocial aspects of genetic screening of pregnant women and newborns:

a systematic review.

By Green JM, Hewison J, Bekker HL, Bryant, Cuckle HS.

No. 34

Evaluation of abnormal uterine bleeding: comparison of three outpatient procedures within cohorts defined by age and menopausal status.

By Critchley HOD, Warner P, Lee AJ, Brechin S, Guise J, Graham B.

No. 35

Coronary artery stents: a rapid systematic review and economic evaluation.

By Hill R, Bagust A, Bakhai A, Dickson R, Dündar Y, Haycox A, *et al*.

No. 36

Review of guidelines for good practice in decision-analytic modelling in health technology assessment.

By Philips Z, Ginnelly L, Sculpher M, Claxton K, Golder S, Riemsma R, et al.

No. 37

Rituximab (MabThera[®]) for aggressive non-Hodgkin's lymphoma: systematic review and economic evaluation. By Knight C, Hind D, Brewer N,

Abbott V.

No. 38

Clinical effectiveness and costeffectiveness of clopidogrel and modified-release dipyridamole in the secondary prevention of occlusive vascular events: a systematic review and economic evaluation.

By Jones L, Griffin S, Palmer S, Main C, Orton V, Sculpher M, *et al.*

No. 39

Pegylated interferon α -2a and -2b in combination with ribavirin in the treatment of chronic hepatitis C: a systematic review and economic evaluation.

By Shepherd J, Brodin H, Cave C, Waugh N, Price A, Gabbay J.

No. 40

Clopidogrel used in combination with aspirin compared with aspirin alone in the treatment of non-ST-segmentelevation acute coronary syndromes: a systematic review and economic evaluation.

By Main C, Palmer S, Griffin S, Jones L, Orton V, Sculpher M, et al.

No. 41

Provision, uptake and cost of cardiac rehabilitation programmes: improving services to under-represented groups. By Beswick AD, Rees K, Griebsch I,

Taylor FC, Burke M, West RR, et al.

No. 42

Involving South Asian patients in clinical trials.

By Hussain-Gambles M, Leese B, Atkin K, Brown J, Mason S, Tovey P.

No. 43

Clinical and cost-effectiveness of continuous subcutaneous insulin infusion for diabetes.

By Colquitt JL, Green C, Sidhu MK, Hartwell D, Waugh N.

No. 44

Identification and assessment of ongoing trials in health technology assessment reviews.

By Song FJ, Fry-Smith A, Davenport C, Bayliss S, Adi Y, Wilson JS, *et al.*

No. 45

Systematic review and economic evaluation of a long-acting insulin analogue, insulin glargine By Warren E, Weatherley-Jones E, Chilcott J, Beverley C.

No. 46

Supplementation of a home-based exercise programme with a class-based programme for people with osteoarthritis of the knees: a randomised controlled trial and health economic analysis.

By McCarthy CJ, Mills PM, Pullen R, Richardson G, Hawkins N, Roberts CR, *et al*.

No. 47

Clinical and cost-effectiveness of oncedaily versus more frequent use of same potency topical corticosteroids for atopic eczema: a systematic review and economic evaluation.

By Green C, Colquitt JL, Kirby J, Davidson P, Payne E.

No. 48

Acupuncture of chronic headache disorders in primary care: randomised controlled trial and economic analysis.

By Vickers AJ, Rees RW, Zollman CE, McCarney R, Smith CM, Ellis N, *et al*.

No. 49

Generalisability in economic evaluation studies in healthcare: a review and case studies.

By Sculpher MJ, Pang FS, Manca A, Drummond MF, Golder S, Urdahl H, *et al.*

No. 50

Virtual outreach: a randomised controlled trial and economic evaluation of joint teleconferenced medical consultations.

By Wallace P, Barber J, Clayton W, Currell R, Fleming K, Garner P, et al.

Volume 9, 2005

No. 1

Randomised controlled multiple treatment comparison to provide a cost-effectiveness rationale for the selection of antimicrobial therapy in acne.

By Ozolins M, Eady EA, Avery A, Cunliffe WJ, O'Neill C, Simpson NB, *et al.*

No. 2

Do the findings of case series studies vary significantly according to methodological characteristics?

By Dalziel K, Round A, Stein K, Garside R, Castelnuovo E, Payne L.

No. 3

Improving the referral process for familial breast cancer genetic counselling: findings of three randomised controlled trials of two interventions.

By Wilson BJ, Torrance N, Mollison J, Wordsworth S, Gray JR, Haites NE, et al.

No. 4

Randomised evaluation of alternative electrosurgical modalities to treat bladder outflow obstruction in men with benign prostatic hyperplasia.

By Fowler C, McAllister W, Plail R, Karim O, Yang Q.

No. 5

A pragmatic randomised controlled trial of the cost-effectiveness of palliative therapies for patients with inoperable oesophageal cancer.

By Shenfine J, McNamee P, Steen N, Bond J, Griffin SM.

No. 6

Impact of computer-aided detection prompts on the sensitivity and specificity of screening mammography.

By Taylor P, Champness J, Given-Wilson R, Johnston K, Potts H.

No. 7

Issues in data monitoring and interim analysis of trials.

By Grant AM, Altman DG, Babiker AB, Campbell MK, Clemens FJ, Darbyshire JH, *et al.*

No. 8

Lay public's understanding of equipoise and randomisation in randomised controlled trials.

By Robinson EJ, Kerr CEP, Stevens AJ, Lilford RJ, Braunholtz DA, Edwards SJ, *et al.*

No. 9

Clinical and cost-effectiveness of electroconvulsive therapy for depressive illness, schizophrenia, catatonia and mania: systematic reviews and economic modelling studies.

By Greenhalgh J, Knight C, Hind D, Beverley C, Walters S.

No. 10

Measurement of health-related quality of life for people with dementia: development of a new instrument (DEMQOL) and an evaluation of current methodology.

By Smith SC, Lamping DL, Banerjee S, Harwood R, Foley B, Smith P, *et al*.

No. 11

Clinical effectiveness and costeffectiveness of drotrecogin alfa (activated) (Xigris[®]) for the treatment of severe sepsis in adults: a systematic review and economic evaluation.

By Green C, Dinnes J, Takeda A, Shepherd J, Hartwell D, Cave C, *et al.*

No. 12

A methodological review of how heterogeneity has been examined in systematic reviews of diagnostic test accuracy.

By Dinnes J, Deeks J, Kirby J, Roderick P.

No. 13

Cervical screening programmes: can automation help? Evidence from systematic reviews, an economic analysis and a simulation modelling exercise applied to the UK.

By Willis BH, Barton P, Pearmain P, Bryan S, Hyde C.

No. 14

Laparoscopic surgery for inguinal hernia repair: systematic review of effectiveness and economic evaluation.

By McCormack K, Wake B, Perez J, Fraser C, Cook J, McIntosh E, *et al*.

No. 15

Clinical effectiveness, tolerability and cost-effectiveness of newer drugs for epilepsy in adults: a systematic review and economic evaluation.

By Wilby J, Kainth A, Hawkins N, Epstein D, McIntosh H, McDaid C, et al.

No. 16

A randomised controlled trial to compare the cost-effectiveness of tricyclic antidepressants, selective serotonin reuptake inhibitors and lofepramine.

By Peveler R, Kendrick T, Buxton M, Longworth L, Baldwin D, Moore M, *et al.*

No. 17

Clinical effectiveness and costeffectiveness of immediate angioplasty for acute myocardial infarction: systematic review and economic evaluation.

By Hartwell D, Colquitt J, Loveman E, Clegg AJ, Brodin H, Waugh N, *et al.*

No. 18

A randomised controlled comparison of alternative strategies in stroke care. By Kalra L, Evans A, Perez I, Knapp M, Swift C, Donaldson N.

No. 19

The investigation and analysis of critical incidents and adverse events in healthcare.

By Woloshynowych M, Rogers S, Taylor-Adams S, Vincent C.

No. 20

Potential use of routine databases in health technology assessment. By Raftery J, Roderick P, Stevens A.

No. 21

Clinical and cost-effectiveness of newer immunosuppressive regimens in renal transplantation: a systematic review and modelling study.

By Woodroffe R, Yao GL, Meads C, Bayliss S, Ready A, Raftery J, et al.

No. 22

A systematic review and economic evaluation of alendronate, etidronate, risedronate, raloxifene and teriparatide for the prevention and treatment of postmenopausal osteoporosis.

By Stevenson M, Lloyd Jones M, De Nigris E, Brewer N, Davis S, Oakley J.

No. 23

A systematic review to examine the impact of psycho-educational interventions on health outcomes and costs in adults and children with difficult asthma.

By Smith JR, Mugford M, Holland R, Candy B, Noble MJ, Harrison BDW, et al.

No. 24

An evaluation of the costs, effectiveness and quality of renal replacement therapy provision in renal satellite units in England and Wales.

By Roderick P, Nicholson T, Armitage A, Mehta R, Mullee M, Gerard K, *et al*.

No. 25

Imatinib for the treatment of patients with unresectable and/or metastatic gastrointestinal stromal tumours: systematic review and economic evaluation.

By Wilson J, Connock M, Song F, Yao G, Fry-Smith A, Raftery J, *et al*.

No. 26

Indirect comparisons of competing interventions.

By Glenny AM, Altman DG, Song F, Sakarovitch C, Deeks JJ, D'Amico R, *et al.*

No. 27

Cost-effectiveness of alternative strategies for the initial medical management of non-ST elevation acute coronary syndrome: systematic review and decision-analytical modelling.

By Robinson M, Palmer S, Sculpher M, Philips Z, Ginnelly L, Bowens A, *et al.*



Outcomes of electrically stimulated gracilis neosphincter surgery.

By Tillin T, Chambers M, Feldman R.

No. 29

The effectiveness and cost-effectiveness of pimecrolimus and tacrolimus for atopic eczema: a systematic review and economic evaluation.

By Garside R, Stein K, Castelnuovo E, Pitt M, Ashcroft D, Dimmock P, et al.

No. 30

Systematic review on urine albumin testing for early detection of diabetic complications.

By Newman DJ, Mattock MB, Dawnay ABS, Kerry S, McGuire A, Yaqoob M, *et al.*

No. 31

Randomised controlled trial of the costeffectiveness of water-based therapy for lower limb osteoarthritis.

By Cochrane T, Davey RC, Matthes Edwards SM.

No. 32

Longer term clinical and economic benefits of offering acupuncture care to patients with chronic low back pain. By Thomas KJ, MacPherson H, Ratcliffe J, Thorpe L, Brazier J, Campbell M, *et al.*

No. 33

Cost-effectiveness and safety of epidural steroids in the management of sciatica. By Price C, Arden N, Coglan L, Rogers P.

No. 34

The British Rheumatoid Outcome Study Group (BROSG) randomised controlled trial to compare the effectiveness and cost-effectiveness of aggressive versus symptomatic therapy in established rheumatoid arthritis.

By Symmons D, Tricker K, Roberts C, Davies L, Dawes P, Scott DL.

No. 35

Conceptual framework and systematic review of the effects of participants' and professionals' preferences in randomised controlled trials.

By King M, Nazareth I, Lampe F, Bower P, Chandler M, Morou M, *et al*.

No. 36

The clinical and cost-effectiveness of implantable cardioverter defibrillators: a systematic review.

By Bryant J, Brodin H, Loveman E, Payne E, Clegg A.

No. 37

A trial of problem-solving by community mental health nurses for anxiety, depression and life difficulties among general practice patients. The CPN-GP study.

By Kendrick T, Simons L, Mynors-Wallis L, Gray A, Lathlean J, Pickering R, *et al*.

No. 38

The causes and effects of sociodemographic exclusions from clinical trials.

By Bartlett C, Doyal L, Ebrahim S, Davey P, Bachmann M, Egger M, *et al.*

No. 39

Is hydrotherapy cost-effective? A randomised controlled trial of combined hydrotherapy programmes compared with physiotherapy land techniques in children with juvenile idiopathic arthritis.

By Epps H, Ginnelly L, Utley M, Southwood T, Gallivan S, Sculpher M, *et al.*

No. 40

A randomised controlled trial and costeffectiveness study of systematic screening (targeted and total population screening) versus routine practice for the detection of atrial fibrillation in people aged 65 and over. The SAFE study.

By Hobbs FDR, Fitzmaurice DA, Mant J, Murray E, Jowett S, Bryan S, *et al.*

No. 41

Displaced intracapsular hip fractures in fit, older people: a randomised comparison of reduction and fixation, bipolar hemiarthroplasty and total hip arthroplasty.

By Keating JF, Grant A, Masson M, Scott NW, Forbes JF.

No. 42

Long-term outcome of cognitive behaviour therapy clinical trials in central Scotland.

By Durham RC, Chambers JA, Power KG, Sharp DM, Macdonald RR, Major KA, *et al*.

No. 43

The effectiveness and cost-effectiveness of dual-chamber pacemakers compared with single-chamber pacemakers for bradycardia due to atrioventricular block or sick sinus syndrome: systematic review and economic evaluation.

By Castelnuovo E, Stein K, Pitt M, Garside R, Payne E.

No. 44

Newborn screening for congenital heart defects: a systematic review and cost-effectiveness analysis.

By Knowles R, Griebsch I, Dezateux C, Brown J, Bull C, Wren C.

No. 45

The clinical and cost-effectiveness of left ventricular assist devices for end-stage heart failure: a systematic review and economic evaluation.

By Clegg AJ, Scott DA, Loveman E, Colquitt J, Hutchinson J, Royle P, *et al.*

No. 46

The effectiveness of the Heidelberg Retina Tomograph and laser diagnostic glaucoma scanning system (GDx) in detecting and monitoring glaucoma.

By Kwartz AJ, Henson DB, Harper RA, Spencer AF, McLeod D.

No. 47

Clinical and cost-effectiveness of autologous chondrocyte implantation for cartilage defects in knee joints: systematic review and economic evaluation.

By Clar C, Cummins E, McIntyre L, Thomas S, Lamb J, Bain L, et al.

No. 48

Systematic review of effectiveness of different treatments for childhood retinoblastoma.

By McDaid C, Hartley S, Bagnall A-M, Ritchie G, Light K, Riemsma R.

No. 49

Towards evidence-based guidelines for the prevention of venous thromboembolism: systematic reviews of mechanical methods, oral anticoagulation, dextran and regional anaesthesia as thromboprophylaxis.

By Roderick P, Ferris G, Wilson K, Halls H, Jackson D, Collins R, *et al.*

No. 50

The effectiveness and cost-effectiveness of parent training/education programmes for the treatment of conduct disorder, including oppositional defiant disorder, in children.

By Dretzke J, Frew E, Davenport C, Barlow J, Stewart-Brown S, Sandercock J, *et al.*

Volume 10, 2006

No. 1

The clinical and cost-effectiveness of donepezil, rivastigmine, galantamine and memantine for Alzheimer's disease.

By Loveman E, Green C, Kirby J, Takeda A, Picot J, Payne E, *et al.*

No. 2

FOOD: a multicentre randomised trial evaluating feeding policies in patients admitted to hospital with a recent stroke.

By Dennis M, Lewis S, Cranswick G, Forbes J.

No. 3

The clinical effectiveness and costeffectiveness of computed tomography screening for lung cancer: systematic reviews.

By Black C, Bagust A, Boland A, Walker S, McLeod C, De Verteuil R, *et al.*

A systematic review of the effectiveness and cost-effectiveness of neuroimaging assessments used to visualise the seizure focus in people with refractory epilepsy being considered for surgery.

By Whiting P, Gupta R, Burch J, Mujica Mota RE, Wright K, Marson A, *et al.*

No. 5

Comparison of conference abstracts and presentations with full-text articles in the health technology assessments of rapidly evolving technologies.

By Dundar Y, Dodd S, Dickson R, Walley T, Haycox A, Williamson PR.

No. 6

Systematic review and evaluation of methods of assessing urinary incontinence.

By Martin JL, Williams KS, Abrams KR, Turner DA, Sutton AJ, Chapple C, *et al.*

No. 7

The clinical effectiveness and costeffectiveness of newer drugs for children with epilepsy. A systematic review.

By Connock M, Frew E, Evans B-W, Bryan S, Cummins C, Fry-Smith A, *et al.*

No. 8

Surveillance of Barrett's oesophagus: exploring the uncertainty through systematic review, expert workshop and economic modelling.

By Garside R, Pitt M, Somerville M, Stein K, Price A, Gilbert N.

No. 9

Topotecan, pegylated liposomal doxorubicin hydrochloride and paclitaxel for second-line or subsequent treatment of advanced ovarian cancer: a systematic review and economic evaluation.

By Main C, Bojke L, Griffin S, Norman G, Barbieri M, Mather L, *et al.*

No. 10

Evaluation of molecular techniques in prediction and diagnosis of cytomegalovirus disease in immunocompromised patients. By Szczepura A, Westmoreland D, Vinogradova Y, Fox J, Clark M.

No. 11

Screening for thrombophilia in high-risk situations: systematic review and costeffectiveness analysis. The Thrombosis: Risk and Economic Assessment of Thrombophilia Screening (TREATS) study.

By Wu O, Robertson L, Twaddle S, Lowe GDO, Clark P, Greaves M, *et al.*

No. 12

A series of systematic reviews to inform a decision analysis for sampling and treating infected diabetic foot ulcers.

By Nelson EA, O'Meara S, Craig D, Iglesias C, Golder S, Dalton J, *et al.*

No. 13

Randomised clinical trial, observational study and assessment of costeffectiveness of the treatment of varicose veins (REACTIV trial).

By Michaels JA, Campbell WB, Brazier JE, MacIntyre JB, Palfreyman SJ, Ratcliffe J, *et al*.

No. 14

The cost-effectiveness of screening for oral cancer in primary care.

By Speight PM, Palmer S, Moles DR, Downer MC, Smith DH, Henriksson M *et al.*

No. 15

Measurement of the clinical and costeffectiveness of non-invasive diagnostic testing strategies for deep vein thrombosis.

By Goodacre S, Sampson F, Stevenson M, Wailoo A, Sutton A, Thomas S, *et al*.

No. 16

Systematic review of the effectiveness and cost-effectiveness of HealOzone[®] for the treatment of occlusal pit/fissure caries and root caries.

By Brazzelli M, McKenzie L, Fielding S, Fraser C, Clarkson J, Kilonzo M, *et al.*

No. 17

Randomised controlled trials of conventional antipsychotic versus new atypical drugs, and new atypical drugs versus clozapine, in people with schizophrenia responding poorly to, or intolerant of, current drug treatment.

By Lewis SW, Davies L, Jones PB, Barnes TRE, Murray RM, Kerwin R, *et al.*

No. 18

Diagnostic tests and algorithms used in the investigation of haematuria: systematic reviews and economic evaluation.

By Rodgers M, Nixon J, Hempel S, Aho T, Kelly J, Neal D, *et al*.

No. 19

Cognitive behavioural therapy in addition to antispasmodic therapy for irritable bowel syndrome in primary care: randomised controlled trial.

By Kennedy TM, Chalder T, McCrone P, Darnley S, Knapp M, Jones RH, *et al*.

No. 20

A systematic review of the clinical effectiveness and cost-effectiveness of enzyme replacement therapies for Fabry's disease and

mucopolysaccharidosis type 1.

By Connock M, Juarez-Garcia A, Frew E, Mans A, Dretzke J, Fry-Smith A, *et al.*

No. 21

Health benefits of antiviral therapy for mild chronic hepatitis C: randomised controlled trial and economic evaluation.

By Wright M, Grieve R, Roberts J, Main J, Thomas HC on behalf of the UK Mild Hepatitis C Trial Investigators.

No. 22

Pressure relieving support surfaces: a randomised evaluation.

By Nixon J, Nelson EA, Cranny G, Iglesias CP, Hawkins K, Cullum NA, et al.

No. 23

A systematic review and economic model of the effectiveness and costeffectiveness of methylphenidate, dexamfetamine and atomoxetine for the treatment of attention deficit hyperactivity disorder in children and adolescents.

By King S, Griffin S, Hodges Z, Weatherly H, Asseburg C, Richardson G, *et al.*

No. 24

The clinical effectiveness and costeffectiveness of enzyme replacement therapy for Gaucher's disease: a systematic review.

By Connock M, Burls A, Frew E, Fry-Smith A, Juarez-Garcia A, McCabe C, *et al*.

No. 25

Effectiveness and cost-effectiveness of salicylic acid and cryotherapy for cutaneous warts. An economic decision model.

By Thomas KS, Keogh-Brown MR, Chalmers JR, Fordham RJ, Holland RC, Armstrong SJ, *et al*.

No. 26

A systematic literature review of the effectiveness of non-pharmacological interventions to prevent wandering in dementia and evaluation of the ethical implications and acceptability of their use.

By Robinson L, Hutchings D, Corner L, Beyer F, Dickinson H, Vanoli A, *et al.*

No. 27

A review of the evidence on the effects and costs of implantable cardioverter defibrillator therapy in different patient groups, and modelling of costeffectiveness and cost-utility for these groups in a UK context.

By Buxton M, Caine N, Chase D, Connelly D, Grace A, Jackson C, et al.



Adefovir dipivoxil and pegylated interferon alfa-2a for the treatment of chronic hepatitis B: a systematic review and economic evaluation.

By Shepherd J, Jones J, Takeda A, Davidson P, Price A.

No. 29

An evaluation of the clinical and costeffectiveness of pulmonary artery catheters in patient management in intensive care: a systematic review and a randomised controlled trial.

By Harvey S, Stevens K, Harrison D, Young D, Brampton W, McCabe C, *et al.*

No. 30

Accurate, practical and cost-effective assessment of carotid stenosis in the UK.

By Wardlaw JM, Chappell FM, Stevenson M, De Nigris E, Thomas S, Gillard J, *et al*.

No. 31

Etanercept and infliximab for the treatment of psoriatic arthritis: a systematic review and economic evaluation.

By Woolacott N, Bravo Vergel Y, Hawkins N, Kainth A, Khadjesari Z, Misso K, *et al*.

No. 32

The cost-effectiveness of testing for hepatitis C in former injecting drug users.

By Castelnuovo E, Thompson-Coon J, Pitt M, Cramp M, Siebert U, Price A, *et al.*

No. 33

Computerised cognitive behaviour therapy for depression and anxiety update: a systematic review and economic evaluation.

By Kaltenthaler E, Brazier J, De Nigris E, Tumur I, Ferriter M, Beverley C, *et al*.

No. 34

Cost-effectiveness of using prognostic information to select women with breast cancer for adjuvant systemic therapy. By Williams C, Brunskill S, Altman D,

Briggs A, Campbell H, Clarke M, et al.

No. 35

Psychological therapies including dialectical behaviour therapy for borderline personality disorder: a systematic review and preliminary economic evaluation.

By Brazier J, Tumur I, Holmes M, Ferriter M, Parry G, Dent-Brown K, et al.

No. 36

Clinical effectiveness and costeffectiveness of tests for the diagnosis and investigation of urinary tract infection in children: a systematic review and economic model.

By Whiting P, Westwood M, Bojke L, Palmer S, Richardson G, Cooper J, *et al.*

No. 37

Cognitive behavioural therapy in chronic fatigue syndrome: a randomised controlled trial of an outpatient group programme.

By O'Dowd H, Gladwell P, Rogers CA, Hollinghurst S, Gregory A.

No. 38

A comparison of the cost-effectiveness of five strategies for the prevention of nonsteroidal anti-inflammatory drug-induced gastrointestinal toxicity: a systematic review with economic modelling.

By Brown TJ, Hooper L, Elliott RA, Payne K, Webb R, Roberts C, *et al.*

No. 39

The effectiveness and cost-effectiveness of computed tomography screening for coronary artery disease: systematic review.

By Waugh N, Black C, Walker S, McIntyre L, Cummins E, Hillis G.

No. 40

What are the clinical outcome and costeffectiveness of endoscopy undertaken by nurses when compared with doctors? A Multi-Institution Nurse Endoscopy Trial (MINuET).

By Williams J, Russell I, Durai D, Cheung W-Y, Farrin A, Bloor K, et al.

No. 41

The clinical and cost-effectiveness of oxaliplatin and capecitabine for the adjuvant treatment of colon cancer: systematic review and economic evaluation.

By Pandor A, Eggington S, Paisley S, Tappenden P, Sutcliffe P.

No. 42

A systematic review of the effectiveness of adalimumab, etanercept and infliximab for the treatment of rheumatoid arthritis in adults and an economic evaluation of their costeffectiveness.

By Chen Y-F, Jobanputra P, Barton P, Jowett S, Bryan S, Clark W, *et al*.

No. 43

Telemedicine in dermatology: a randomised controlled trial. By Bowns IR, Collins K, Walters SJ, McDonagh AJG.

No. 44

Cost-effectiveness of cell salvage and alternative methods of minimising perioperative allogeneic blood transfusion: a systematic review and economic model.

By Davies L, Brown TJ, Haynes S, Payne K, Elliott RA, McCollum C.

No. 45

Clinical effectiveness and costeffectiveness of laparoscopic surgery for colorectal cancer: systematic reviews and economic evaluation.

By Murray A, Lourenco T, de Verteuil R, Hernandez R, Fraser C, McKinley A, *et al.*

No. 46

Etanercept and efalizumab for the treatment of psoriasis: a systematic review.

By Woolacott N, Hawkins N, Mason A, Kainth A, Khadjesari Z, Bravo Vergel Y, *et al*.

No. 47

Systematic reviews of clinical decision tools for acute abdominal pain. By Liu JLY, Wyatt JC, Deeks JJ, Clamp S, Keen J, Verde P, *et al*.

No. 48

Evaluation of the ventricular assist device programme in the UK. By Sharples L, Buxton M, Caine N, Cafferty F, Demiris N, Dyer M, *et al.*

No. 49

A systematic review and economic model of the clinical and costeffectiveness of immunosuppressive therapy for renal transplantation in children.

By Yao G, Albon E, Adi Y, Milford D, Bayliss S, Ready A, et al.

No. 50

Amniocentesis results: investigation of anxiety. The ARIA trial. By Hewison J, Nixon J, Fountain J,

Cocks K, Jones C, Mason G, et al.

Volume 11, 2007

No. 1

Pemetrexed disodium for the treatment of malignant pleural mesothelioma: a systematic review and economic evaluation.

By Dundar Y, Bagust A, Dickson R, Dodd S, Green J, Haycox A, *et al*.

No. 2

A systematic review and economic model of the clinical effectiveness and cost-effectiveness of docetaxel in combination with prednisone or prednisolone for the treatment of hormone-refractory metastatic prostate cancer.

By Collins R, Fenwick E, Trowman R, Perard R, Norman G, Light K, *et al.*

No. 3

A systematic review of rapid diagnostic tests for the detection of tuberculosis infection.

By Dinnes J, Deeks J, Kunst H, Gibson A, Cummins E, Waugh N, *et al*.

No. 4

The clinical effectiveness and costeffectiveness of strontium ranelate for the prevention of osteoporotic fragility fractures in postmenopausal women.

By Stevenson M, Davis S, Lloyd-Jones M, Beverley C.

A systematic review of quantitative and qualitative research on the role and effectiveness of written information available to patients about individual medicines.

By Raynor DK, Blenkinsopp A, Knapp P, Grime J, Nicolson DJ, Pollock K, *et al*.

No. 6

Oral naltrexone as a treatment for relapse prevention in formerly opioid-dependent drug users: a systematic review and economic evaluation.

By Adi Y, Juarez-Garcia A, Wang D, Jowett S, Frew E, Day E, *et al*.

No. 7

Glucocorticoid-induced osteoporosis: a systematic review and cost-utility analysis.

By Kanis JA, Stevenson M, McCloskey EV, Davis S, Lloyd-Jones M.

No. 8

Epidemiological, social, diagnostic and economic evaluation of population screening for genital chlamydial infection.

By Low N, McCarthy A, Macleod J, Salisbury C, Campbell R, Roberts TE, *et al.*

No. 9

Methadone and buprenorphine for the management of opioid dependence: a systematic review and economic evaluation.

By Connock M, Juarez-Garcia A, Jowett S, Frew E, Liu Z, Taylor RJ, *et al.*

No. 10

Exercise Evaluation Randomised Trial (EXERT): a randomised trial comparing GP referral for leisure centre-based exercise, community-based walking and advice only.

By Isaacs AJ, Critchley JA, See Tai S, Buckingham K, Westley D, Harridge SDR, *et al*.

No. 11

Interferon alfa (pegylated and non-pegylated) and ribavirin for the treatment of mild chronic hepatitis C: a systematic review and economic evaluation.

By Shepherd J, Jones J, Hartwell D, Davidson P, Price A, Waugh N.

No. 12

Systematic review and economic evaluation of bevacizumab and cetuximab for the treatment of metastatic colorectal cancer.

By Tappenden P, Jones R, Paisley S, Carroll C.

No. 13

A systematic review and economic evaluation of epoetin alfa, epoetin beta and darbepoetin alfa in anaemia associated with cancer, especially that attributable to cancer treatment.

By Wilson J, Yao GL, Raftery J, Bohlius J, Brunskill S, Sandercock J, *et al.*

No. 14

A systematic review and economic evaluation of statins for the prevention of coronary events.

By Ward S, Lloyd Jones M, Pandor A, Holmes M, Ara R, Ryan A, *et al*.

No. 15

A systematic review of the effectiveness and cost-effectiveness of different models of community-based respite care for frail older people and their carers.

By Mason A, Weatherly H, Spilsbury K, Arksey H, Golder S, Adamson J, *et al.*

No. 16

Additional therapy for young children with spastic cerebral palsy: a randomised controlled trial.

By Weindling AM, Cunningham CC, Glenn SM, Edwards RT, Reeves DJ.

No. 17

Screening for type 2 diabetes: literature review and economic modelling. By Waugh N, Scotland G, McNamee P, Gillett M, Brennan A, Goyder E, *et al.*

No. 18

The effectiveness and cost-effectiveness of cinacalcet for secondary hyperparathyroidism in end-stage renal disease patients on dialysis: a systematic review and economic evaluation.

By Garside R, Pitt M, Anderson R, Mealing S, Roome C, Snaith A, *et al.*

No. 19

The clinical effectiveness and cost-effectiveness of gemcitabine for metastatic breast cancer: a systematic review and economic evaluation. By Takeda AL, Jones J, Loveman E, Tan SC, Clegg AJ.

No. 20

A systematic review of duplex ultrasound, magnetic resonance angiography and computed tomography angiography for the diagnosis and assessment of symptomatic, lower limb peripheral arterial disease.

By Collins R, Cranny G, Burch J, Aguiar-Ibáñez R, Craig D, Wright K, *et al*.

No. 21

The clinical effectiveness and costeffectiveness of treatments for children with idiopathic steroid-resistant nephrotic syndrome: a systematic review. By Colquitt JL, Kirby J, Green C,

Cooper K, Trompeter RS.

No. 22

A systematic review of the routine monitoring of growth in children of primary school age to identify growth-related conditions. By Fayter D, Nixon J, Hartley S,

Rithalia A, Butler G, Rudolf M, *et al.*

No. 23

Systematic review of the effectiveness of preventing and treating *Staphylococcus aureus* carriage in reducing peritoneal catheter-related infections.

By McCormack K, Rabindranath K, Kilonzo M, Vale L, Fraser C, McIntyre L, *et al.*

No. 24

The clinical effectiveness and cost of repetitive transcranial magnetic stimulation versus electroconvulsive therapy in severe depression: a multicentre pragmatic randomised controlled trial and economic analysis.

By McLoughlin DM, Mogg A, Eranti S, Pluck G, Purvis R, Edwards D, *et al.*

No. 25

A randomised controlled trial and economic evaluation of direct versus indirect and individual versus group modes of speech and language therapy for children with primary language impairment.

By Boyle J, McCartney E, Forbes J, O'Hare A.

No. 26

Hormonal therapies for early breast cancer: systematic review and economic evaluation.

By Hind D, Ward S, De Nigris E, Simpson E, Carroll C, Wyld L.

No. 27

Cardioprotection against the toxic effects of anthracyclines given to children with cancer: a systematic review.

By Bryant J, Picot J, Levitt G, Sullivan I, Baxter L, Clegg A.

No. 28

Adalimumab, etanercept and infliximab for the treatment of ankylosing spondylitis: a systematic review and economic evaluation.

By McLeod C, Bagust A, Boland A, Dagenais P, Dickson R, Dundar Y, et al.



Prenatal screening and treatment strategies to prevent group B streptococcal and other bacterial infections in early infancy: costeffectiveness and expected value of information analyses.

By Colbourn T, Asseburg C, Bojke L, Philips Z, Claxton K, Ades AE, *et al.*

No. 30

Clinical effectiveness and costeffectiveness of bone morphogenetic proteins in the non-healing of fractures and spinal fusion: a systematic review.

By Garrison KR, Donell S, Ryder J, Shemilt I, Mugford M, Harvey I, *et al.*

No. 31

A randomised controlled trial of postoperative radiotherapy following breast-conserving surgery in a minimum-risk older population. The PRIME trial.

By Prescott RJ, Kunkler IH, Williams LJ, King CC, Jack W, van der Pol M, *et al.*

No. 32

Current practice, accuracy, effectiveness and cost-effectiveness of the school entry hearing screen.

By Bamford J, Fortnum H, Bristow K, Smith J, Vamvakas G, Davies L, *et al*.

No. 33

The clinical effectiveness and cost-effectiveness of inhaled insulin in diabetes mellitus: a systematic review and economic evaluation.

By Black C, Cummins E, Royle P, Philip S, Waugh N.

No. 34

Surveillance of cirrhosis for hepatocellular carcinoma: systematic review and economic analysis.

By Thompson Coon J, Rogers G, Hewson P, Wright D, Anderson R, Cramp M, *et al*.

No. 35

The Birmingham Rehabilitation Uptake Maximisation Study (BRUM). Homebased compared with hospital-based cardiac rehabilitation in a multi-ethnic population: cost-effectiveness and patient adherence.

By Jolly K, Taylor R, Lip GYH, Greenfield S, Raftery J, Mant J, et al.

No. 36

A systematic review of the clinical, public health and cost-effectiveness of rapid diagnostic tests for the detection and identification of bacterial intestinal pathogens in faeces and food.

By Abubakar I, Irvine L, Aldus CF, Wyatt GM, Fordham R, Schelenz S, *et al.*

No. 37

A randomised controlled trial examining the longer-term outcomes of standard versus new antiepileptic drugs. The SANAD trial.

By Marson AG, Appleton R, Baker GA, Chadwick DW, Doughty J, Eaton B, *et al*.

No. 38

Clinical effectiveness and costeffectiveness of different models of managing long-term oral anticoagulation therapy: a systematic review and economic modelling.

By Connock M, Stevens C, Fry-Smith A, Jowett S, Fitzmaurice D, Moore D, *et al*.

No. 39

A systematic review and economic model of the clinical effectiveness and cost-effectiveness of interventions for preventing relapse in people with bipolar disorder.

By Soares-Weiser K, Bravo Vergel Y, Beynon S, Dunn G, Barbieri M, Duffy S, *et al.*

No. 40

Taxanes for the adjuvant treatment of early breast cancer: systematic review and economic evaluation.

By Ward S, Simpson E, Davis S, Hind D, Rees A, Wilkinson A.

No. 41

The clinical effectiveness and costeffectiveness of screening for open angle glaucoma: a systematic review and economic evaluation.

By Burr JM, Mowatt G, Hernández R, Siddiqui MAR, Cook J, Lourenco T, *et al.*

No. 42

Acceptability, benefit and costs of early screening for hearing disability: a study of potential screening tests and models.

By Davis A, Smith P, Ferguson M, Stephens D, Gianopoulos I.

No. 43

Contamination in trials of educational interventions.

By Keogh-Brown MR, Bachmann MO, Shepstone L, Hewitt C, Howe A, Ramsay CR, *et al*.

No. 44

Overview of the clinical effectiveness of positron emission tomography imaging in selected cancers.

By Facey K, Bradbury I, Laking G, Payne E.

No. 45

The effectiveness and cost-effectiveness of carmustine implants and temozolomide for the treatment of newly diagnosed high-grade glioma: a systematic review and economic evaluation.

By Garside R, Pitt M, Anderson R, Rogers G, Dyer M, Mealing S, *et al.*

No. 46

Drug-eluting stents: a systematic review and economic evaluation.

By Hill RA, Boland A, Dickson R, Dündar Y, Haycox A, McLeod C, *et al.*

No. 47

The clinical effectiveness and costeffectiveness of cardiac resynchronisation (biventricular pacing) for heart failure: systematic review and economic model.

By Fox M, Mealing S, Anderson R, Dean J, Stein K, Price A, *et al*.

No. 48

Recruitment to randomised trials: strategies for trial enrolment and participation study. The STEPS study.

By Campbell MK, Snowdon C, Francis D, Elbourne D, McDonald AM, Knight R, *et al*.

No. 49

Cost-effectiveness of functional cardiac testing in the diagnosis and management of coronary artery disease: a randomised controlled trial. The CECaT trial.

By Sharples L, Hughes V, Crean A, Dyer M, Buxton M, Goldsmith K, *et al.*

No. 50

Evaluation of diagnostic tests when there is no gold standard. A review of methods.

By Rutjes AWS, Reitsma JB, Coomarasamy A, Khan KS, Bossuyt PMM.

No. 51

Systematic reviews of the clinical effectiveness and cost-effectiveness of proton pump inhibitors in acute upper gastrointestinal bleeding.

By Leontiadis GI, Sreedharan A, Dorward S, Barton P, Delaney B, Howden CW, *et al*.

No. 52

A review and critique of modelling in prioritising and designing screening programmes.

By Karnon J, Goyder E, Tappenden P, McPhie S, Towers I, Brazier J, *et al*.

No. 53

An assessment of the impact of the NHS Health Technology Assessment Programme.

By Hanney S, Buxton M, Green C, Coulson D, Raftery J.

Volume 12, 2008

No. 1

A systematic review and economic model of switching from nonglycopeptide to glycopeptide antibiotic prophylaxis for surgery.

By Cranny G, Elliott R, Weatherly H, Chambers D, Hawkins N, Myers L, *et al.*

'Cut down to quit' with nicotine replacement therapies in smoking cessation: a systematic review of effectiveness and economic analysis.

By Wang D, Connock M, Barton P, Fry-Smith A, Aveyard P, Moore D.

No. 3

A systematic review of the effectiveness of strategies for reducing fracture risk in children with juvenile idiopathic arthritis with additional data on longterm risk of fracture and cost of disease management.

By Thornton J, Ashcroft D, O'Neill T, Elliott R, Adams J, Roberts C, *et al.*

No. 4

Does befriending by trained lay workers improve psychological well-being and quality of life for carers of people with dementia, and at what cost? A randomised controlled trial.

By Charlesworth G, Shepstone L, Wilson E, Thalanany M, Mugford M, Poland F.

No. 5

A multi-centre retrospective cohort study comparing the efficacy, safety and costeffectiveness of hysterectomy and uterine artery embolisation for the treatment of symptomatic uterine fibroids. The HOPEFUL study.

By Hirst A, Dutton S, Wu O, Briggs A, Edwards C, Waldenmaier L, *et al.*

No. 6

Methods of prediction and prevention of pre-eclampsia: systematic reviews of accuracy and effectiveness literature with economic modelling.

By Meads CA, Cnossen JS, Meher S, Juarez-Garcia A, ter Riet G, Duley L, *et al.*

No. 7

The use of economic evaluations in NHS decision-making: a review and empirical investigation.

By Williams I, McIver S, Moore D, Bryan S.

No. 8

Stapled haemorrhoidectomy (haemorrhoidopexy) for the treatment of haemorrhoids: a systematic review and economic evaluation.

By Burch J, Epstein D, Baba-Akbari A, Weatherly H, Fox D, Golder S, *et al*.

No. 9

The clinical effectiveness of diabetes education models for Type 2 diabetes: a systematic review.

By Loveman E, Frampton GK, Clegg AJ.

No. 10

Payment to healthcare professionals for patient recruitment to trials: systematic review and qualitative study. By Raftery J, Bryant J, Powell J, Kerr C, Hawker S.

No. 11

Cyclooxygenase-2 selective non-steroidal anti-inflammatory drugs (etodolac, meloxicam, celecoxib, rofecoxib, etoricoxib, valdecoxib and lumiracoxib) for osteoarthritis and rheumatoid arthritis: a systematic review and economic evaluation.

By Chen Y-F, Jobanputra P, Barton P, Bryan S, Fry-Smith A, Harris G, *et al.*

No. 12

The clinical effectiveness and costeffectiveness of central venous catheters treated with anti-infective agents in preventing bloodstream infections: a systematic review and economic evaluation.

By Hockenhull JC, Dwan K, Boland A, Smith G, Bagust A, Dündar Y, et al.

No. 13

Stepped treatment of older adults on laxatives. The STOOL trial. By Mihaylov S, Stark C, McColl E,

Steen N, Vanoli A, Rubin G, *et al.*

No. 14

A randomised controlled trial of cognitive behaviour therapy in adolescents with major depression treated by selective serotonin reuptake inhibitors. The ADAPT trial.

By Goodyer IM, Dubicka B, Wilkinson P, Kelvin R, Roberts C, Byford S, *et al*.

No. 15

The use of irinotecan, oxaliplatin and raltitrexed for the treatment of advanced colorectal cancer: systematic review and economic evaluation. By Hind D, Tappenden P, Tumur I,

Eggington E, Sutcliffe P, Ryan A.

No. 16

Ranibizumab and pegaptanib for the treatment of age-related macular degeneration: a systematic review and economic evaluation.

By Colquitt JL, Jones J, Tan SC, Takeda A, Clegg AJ, Price A.

No. 17

Systematic review of the clinical effectiveness and cost-effectiveness of 64-slice or higher computed tomography angiography as an alternative to invasive coronary angiography in the investigation of coronary artery disease.

By Mowatt G, Cummins E, Waugh N, Walker S, Cook J, Jia X, et al.

No. 18

Structural neuroimaging in psychosis: a systematic review and economic evaluation.

By Albon E, Tsourapas A, Frew E, Davenport C, Oyebode F, Bayliss S, *et al.*

No. 19

Systematic review and economic analysis of the comparative effectiveness of different inhaled corticosteroids and their usage with long-acting beta₂ agonists for the treatment of chronic asthma in adults and children aged 12 years and over.

By Shepherd J, Rogers G, Anderson R, Main C, Thompson-Coon J, Hartwell D, *et al.*

No. 20

Systematic review and economic analysis of the comparative effectiveness of different inhaled corticosteroids and their usage with long-acting beta₂ agonists for the treatment of chronic asthma in children under the age of 12 years.

By Main C, Shepherd J, Anderson R, Rogers G, Thompson-Coon J, Liu Z, *et al.*

No. 21

Ezetimibe for the treatment of hypercholesterolaemia: a systematic review and economic evaluation.

By Ara R, Tumur I, Pandor A, Duenas A, Williams R, Wilkinson A, *et al.*

No. 22

Topical or oral ibuprofen for chronic knee pain in older people. The TOIB study.

By Underwood M, Ashby D, Carnes D, Castelnuovo E, Cross P, Harding G, *et al.*

No. 23

A prospective randomised comparison of minor surgery in primary and secondary care. The MiSTIC trial.

By George S, Pockney P, Primrose J, Smith H, Little P, Kinley H, et al.

No. 24

A review and critical appraisal of measures of therapist–patient

interactions in mental health settings. By Cahill J, Barkham M, Hardy G, Gilbody S, Richards D, Bower P, *et al.*

No. 25

The clinical effectiveness and costeffectiveness of screening programmes for amblyopia and strabismus in children up to the age of 4–5 years: a systematic review and economic evaluation.

By Carlton J, Karnon J, Czoski-Murray C, Smith KJ, Marr J.



A systematic review of the clinical effectiveness and cost-effectiveness and economic modelling of minimal incision total hip replacement approaches in the management of arthritic disease of the hip.

By de Verteuil R, Imamura M, Zhu S, Glazener C, Fraser C, Munro N, *et al*.

No. 27

A preliminary model-based assessment of the cost–utility of a screening programme for early age-related macular degeneration.

By Karnon J, Czoski-Murray C, Smith K, Brand C, Chakravarthy U, Davis S, *et al.*



Director,

Professor Tom Walley, Director, NHS HTA Programme, Department of Pharmacology & Therapeutics, University of Liverpool Deputy Director, Professor Jon Nicholl, Director, Medical Care Research Unit, University of Sheffield, School of Health and Related Research

Prioritisation Strategy Group

Members

Members

Chair.

Professor Tom Walley, Director, NHS HTA Programme, Department of Pharmacology & Therapeutics, University of Liverpool Professor Bruce Campbell, Consultant Vascular & General Surgeon, Royal Devon & Exeter Hospital

Professor Robin E Ferner, Consultant Physician and Director, West Midlands Centre for Adverse Drug Reactions, City Hospital NHS Trust, Birmingham Dr Edmund Jessop, Medical Adviser, National Specialist, Commissioning Advisory Group (NSCAG), Department of Health, London

Professor Jon Nicholl, Director, Medical Care Research Unit, University of Sheffield, School of Health and Related Research Dr Ron Zimmern, Director, Public Health Genetics Unit, Strangeways Research Laboratories, Cambridge

HTA Commissioning Board

Programme Director, Professor Tom Walley, Director, NHS HTA Programme, Department of Pharmacology & Therapeutics, University of Liverpool

Chair, Professor Jon Nicholl, Director, Medical Care Research Unit, University of Sheffield, School of Health and Related Research

Deputy Chair, Dr Andrew Farmer, University Lecturer in General Practice, Department of Primary Health Care, University of Oxford

Dr Jeffrey Aronson, Reader in Clinical Pharmacology, Department of Clinical Pharmacology, Radcliffe Infirmary, Oxford Professor Deborah Ashby, Professor of Medical Statistics, Department of Environmental and Preventative Medicine, Queen Mary University of London

Professor Ann Bowling, Professor of Health Services Research, Primary Care and Population Studies, University College London

Professor John Cairns, Professor of Health Economics, Public Health Policy, London School of Hygiene and Tropical Medicine, London

Professor Nicky Cullum, Director of Centre for Evidence Based Nursing, Department of Health Sciences, University of York

Professor Jon Deeks, Professor of Health Statistics, University of Birmingham Professor Jenny Donovan, Professor of Social Medicine, Department of Social Medicine, University of Bristol

Professor Freddie Hamdy, Professor of Urology, University of Sheffield

Professor Allan House, Professor of Liaison Psychiatry, University of Leeds

Professor Sallie Lamb, Director, Warwick Clinical Trials Unit, University of Warwick

Professor Stuart Logan, Director of Health & Social Care Research, The Peninsula Medical School, Universities of Exeter & Plymouth

Professor Miranda Mugford, Professor of Health Economics, University of East Anglia

Dr Linda Patterson, Consultant Physician, Department of Medicine, Burnley General Hospital Professor Ian Roberts, Professor of Epidemiology & Public Health, Intervention Research Unit, London School of Hygiene and Tropical Medicine

Professor Mark Sculpher, Professor of Health Economics, Centre for Health Economics, Institute for Research in the Social Services, University of York

Professor Kate Thomas, Professor of Complementary and Alternative Medicine, University of Leeds

Professor David John Torgerson, Director of York Trial Unit, Department of Health Sciences, University of York

Professor Hywel Williams, Professor of Dermato-Epidemiology, University of Nottingham

Current and past membership details of all HTA 'committees' are available from the HTA website (www.hta.ac.uk)

Diagnostic Technologies & Screening Panel

Members

Chair, Dr Ron Zimmern, Director of the Public Health Genetics Unit, Strangeways Research Laboratories, Cambridge

Ms Norma Armston, Freelance Consumer Advocate, Bolton

Professor Max Bachmann, Professor of Health Care Interfaces, Department of Health Policy and Practice, University of East Anglia

Professor Rudy Bilous Professor of Clinical Medicine & Consultant Physician, The Academic Centre, South Tees Hospitals NHS Trust

Ms Dea Birkett, Service User Representative, London Dr Paul Cockcroft, Consultant Medical Microbiologist and Clinical Director of Pathology, Department of Clinical Microbiology, St Mary's Hospital, Portsmouth

Professor Adrian K Dixon, Professor of Radiology, University Department of Radiology, University of Cambridge Clinical School

Dr David Elliman, Consultant in Community Child Health, Islington PCT & Great Ormond Street Hospital, London

Professor Glyn Elwyn, Research Chair, Centre for Health Sciences Research, Cardiff University, Department of General Practice, Cardiff

Professor Paul Glasziou, Director, Centre for Evidence-Based Practice, University of Oxford Dr Jennifer J Kurinczuk, Consultant Clinical Epidemiologist, National Perinatal Epidemiology Unit, Oxford

Dr Susanne M Ludgate, Clinical Director, Medicines & Healthcare Products Regulatory Agency, London

Mr Stephen Pilling, Director, Centre for Outcomes, Research & Effectiveness, Joint Director, National Collaborating Centre for Mental Health, University College London

Mrs Una Rennard, Service User Representative, Oxford

Dr Phil Shackley, Senior Lecturer in Health Economics, Academic Vascular Unit, University of Sheffield Dr Margaret Somerville, Director of Public Health Learning, Peninsula Medical School, University of Plymouth

Dr Graham Taylor, Scientific Director & Senior Lecturer, Regional DNA Laboratory, The Leeds Teaching Hospitals

Professor Lindsay Wilson Turnbull, Scientific Director, Centre for MR Investigations & YCR Professor of Radiology, University of Hull

Professor Martin J Whittle, Clinical Co-director, National Co-ordinating Centre for Women's and Childhealth

Dr Dennis Wright, Consultant Biochemist & Clinical Director, The North West London Hospitals NHS Trust, Middlesex

Pharmaceuticals Panel

Members

Chair,

Professor Robin Ferner, Consultant Physician and Director, West Midlands Centre for Adverse Drug Reactions, City Hospital NHS Trust, Birmingham

Ms Anne Baileff, Consultant Nurse in First Contact Care, Southampton City Primary Care Trust, University of Southampton Professor Imti Choonara, Professor in Child Health, Academic Division of Child Health, University of Nottingham

Professor John Geddes, Professor of Epidemiological Psychiatry, University of Oxford

Mrs Barbara Greggains, Non-Executive Director, Greggains Management Ltd

Dr Bill Gutteridge, Medical Adviser, National Specialist Commissioning Advisory Group (NSCAG), London

Mrs Sharon Hart, Consultant Pharmaceutical Adviser, Reading Dr Jonathan Karnon, Senior Research Fellow, Health Economics and Decision Science, University of Sheffield

Dr Yoon Loke, Senior Lecturer in Clinical Pharmacology, University of East Anglia

Ms Barbara Meredith, Lay Member, Epsom

Dr Andrew Prentice, Senior Lecturer and Consultant Obstetrician & Gynaecologist, Department of Obstetrics & Gynaecology, University of Cambridge

Dr Frances Rotblat, CPMP Delegate, Medicines & Healthcare Products Regulatory Agency, London Dr Martin Shelly, General Practitioner, Leeds

Mrs Katrina Simister, Assistant Director New Medicines, National Prescribing Centre, Liverpool

Dr Richard Tiner, Medical Director, Medical Department, Association of the British Pharmaceutical Industry, London



Therapeutic Procedures Panel

Members Chair, Professor Bruce Campbell, Consultant Vascular and General Surgeon, Department of Surgery, Royal Devon & Exeter Hospital

Dr Mahmood Adil, Deputy Regional Director of Public Health, Department of Health, Manchester

Dr Aileen Clarke, Consultant in Public Health, Public Health Resource Unit, Oxford Professor Matthew Cooke, Professor of Emergency Medicine, Warwick Emergency Care and Rehabilitation, University of Warwick

Mr Mark Emberton, Senior Lecturer in Oncological Urology, Institute of Urology, University College Hospital

Professor Paul Gregg, Professor of Orthopaedic Surgical Science, Department of General Practice and Primary Care, South Tees Hospital NHS Trust, Middlesbrough

Ms Maryann L Hardy, Lecturer, Division of Radiography, University of Bradford Dr Simon de Lusignan, Senior Lecturer, Primary Care Informatics, Department of Community Health Sciences, St George's Hospital Medical School, London

Dr Peter Martin, Consultant Neurologist, Addenbrooke's Hospital, Cambridge

Professor Neil McIntosh, Edward Clark Professor of Child Life & Health, Department of Child Life & Health, University of Edinburgh

Professor Jim Neilson, Professor of Obstetrics and Gynaecology, Department of Obstetrics and Gynaecology, University of Liverpool Dr John C Pounsford, Consultant Physician, Directorate of Medical Services, North Bristol NHS Trust

Dr Karen Roberts, Nurse Consultant, Queen Elizabeth Hospital, Gateshead

Dr Vimal Sharma, Consultant Psychiatrist/Hon. Senior Lecturer, Mental Health Resource Centre, Cheshire and Wirral Partnership NHS Trust, Wallasey

Professor Scott Weich, Professor of Psychiatry, Division of Health in the Community, University of Warwick

Disease Prevention Panel

Members

Chair, Dr Edmund Jessop, Medical Adviser, National Specialist Commissioning Advisory Group (NSCAG), London

Mrs Sheila Clark, Chief Executive, St James's Hospital, Portsmouth

Mr Richard Copeland, Lead Pharmacist: Clinical Economy/Interface, Wansbeck General Hospital, Northumberland Dr Elizabeth Fellow-Smith, Medical Director, West London Mental Health Trust, Middlesex

Mr Ian Flack, Director PPI Forum Support, Council of Ethnic Minority Voluntary Sector Organisations, Stratford

Dr John Jackson, General Practitioner, Newcastle upon Tyne

Mrs Veronica James, Chief Officer, Horsham District Age Concern, Horsham

Professor Mike Kelly, Director, Centre for Public Health Excellence, National Institute for Health and Clinical Excellence, London Professor Yi Mien Koh, Director of Public Health and Medical Director, London NHS (North West London Strategic Health Authority), London

Ms Jeanett Martin, Director of Clinical Leadership & Quality, Lewisham PCT, London

Dr Chris McCall, General Practitioner, Dorset

Dr David Pencheon, Director, Eastern Region Public Health Observatory, Cambridge

Dr Ken Stein, Senior Clinical Lecturer in Public Health, Director, Peninsula Technology Assessment Group, University of Exeter, Exeter Dr Carol Tannahill, Director, Glasgow Centre for Population Health, Glasgow

Professor Margaret Thorogood, Professor of Epidemiology, University of Warwick, Coventry

Dr Ewan Wilkinson, Consultant in Public Health, Royal Liverpool University Hospital, Liverpool

Expert Advisory Network

Members

Professor Douglas Altman, Professor of Statistics in Medicine, Centre for Statistics in Medicine, University of Oxford

Professor John Bond, Director, Centre for Health Services Research, University of Newcastle upon Tyne, School of Population & Health Sciences, Newcastle upon Tyne

Professor Andrew Bradbury, Professor of Vascular Surgery, Solihull Hospital, Birmingham

Mr Shaun Brogan, Chief Executive, Ridgeway Primary Care Group, Aylesbury

Mrs Stella Burnside OBE, Chief Executive, Regulation and Improvement Authority, Belfast

Ms Tracy Bury, Project Manager, World Confederation for Physical Therapy, London

Professor Iain T Cameron, Professor of Obstetrics and Gynaecology and Head of the School of Medicine, University of Southampton

Dr Christine Clark, Medical Writer & Consultant Pharmacist, Rossendale

Professor Collette Clifford, Professor of Nursing & Head of Research, School of Health Sciences, University of Birmingham, Edgbaston, Birmingham

Professor Barry Cookson, Director, Laboratory of Healthcare Associated Infection, Health Protection Agency, London

Dr Carl Counsell, Clinical Senior Lecturer in Neurology, Department of Medicine & Therapeutics, University of Aberdeen

Professor Howard Cuckle, Professor of Reproductive Epidemiology, Department of Paediatrics, Obstetrics & Gynaecology, University of Leeds

Dr Katherine Darton, Information Unit, MIND – The Mental Health Charity, London Professor Carol Dezateux, Professor of Paediatric Epidemiology, London

Dr Keith Dodd, Consultant Paediatrician, Derby

Mr John Dunning, Consultant Cardiothoracic Surgeon, Cardiothoracic Surgical Unit, Papworth Hospital NHS Trust, Cambridge

Mr Jonothan Earnshaw, Consultant Vascular Surgeon, Gloucestershire Royal Hospital, Gloucester

Professor Martin Eccles, Professor of Clinical Effectiveness, Centre for Health Services Research, University of Newcastle upon Tyne

Professor Pam Enderby, Professor of Community Rehabilitation, Institute of General Practice and Primary Care, University of Sheffield

Professor Gene Feder, Professor of Primary Care Research & Development, Centre for Health Sciences, Barts & The London Queen Mary's School of Medicine & Dentistry, London

Mr Leonard R Fenwick, Chief Executive, Newcastle upon Tyne Hospitals NHS Trust

Mrs Gillian Fletcher, Antenatal Teacher & Tutor and President, National Childbirth Trust, Henfield

Professor Jayne Franklyn, Professor of Medicine, Department of Medicine, University of Birmingham, Queen Elizabeth Hospital, Edgbaston, Birmingham

Dr Neville Goodman, Consultant Anaesthetist, Southmead Hospital, Bristol

Professor Robert E Hawkins, CRC Professor and Director of Medical Oncology, Christie CRC Research Centre, Christie Hospital NHS Trust, Manchester

Professor Allen Hutchinson, Director of Public Health & Deputy Dean of ScHARR, Department of Public Health, University of Sheffield

Professor Peter Jones, Professor of Psychiatry, University of Cambridge, Cambridge Professor Stan Kaye, Cancer Research UK Professor of Medical Oncology, Section of Medicine, Royal Marsden Hospital & Institute of Cancer Research. Surrev

Dr Duncan Keeley, General Practitioner (Dr Burch & Ptnrs), The Health Centre, Thame

Dr Donna Lamping, Research Degrees Programme Director & Reader in Psychology, Health Services Research Unit, London School of Hygiene and Tropical Medicine, London

Mr George Levvy, Chief Executive, Motor Neurone Disease Association, Northampton

Professor James Lindesay, Professor of Psychiatry for the Elderly, University of Leicester, Leicester General Hospital

Professor Julian Little, Professor of Human Genome Epidemiology, Department of Epidemiology & Community Medicine, University of Ottawa

Professor Rajan Madhok, Consultant in Public Health, South Manchester Primary Care Trust, Manchester

Professor Alexander Markham, Director, Molecular Medicine Unit, St James's University Hospital, Leeds

Professor Alistaire McGuire, Professor of Health Economics, London School of Economics

Dr Peter Moore, Freelance Science Writer, Ashtead

Dr Andrew Mortimore, Public Health Director, Southampton City Primary Care Trust, Southampton

Dr Sue Moss, Associate Director, Cancer Screening Evaluation Unit, Institute of Cancer Research, Sutton

Mrs Julietta Patnick, Director, NHS Cancer Screening Programmes, Sheffield

Professor Robert Peveler, Professor of Liaison Psychiatry, Royal South Hants Hospital, Southampton Professor Chris Price, Visiting Professor in Clinical Biochemistry, University of Oxford

Professor William Rosenberg, Professor of Hepatology and Consultant Physician, University of Southampton, Southampton

Professor Peter Sandercock, Professor of Medical Neurology, Department of Clinical Neurosciences, University of Edinburgh

Dr Susan Schonfield, Consultant in Public Health, Hillingdon PCT, Middlesex

Dr Eamonn Sheridan, Consultant in Clinical Genetics, Genetics Department, St James's University Hospital, Leeds

Professor Sarah Stewart-Brown, Professor of Public Health, University of Warwick, Division of Health in the Community Warwick Medical School, LWMS, Coventry

Professor Ala Szczepura, Professor of Health Service Research, Centre for Health Services Studies, University of Warwick

Dr Ross Taylor, Senior Lecturer, Department of General Practice and Primary Care, University of Aberdeen

Mrs Joan Webster, Consumer member, HTA – Expert Advisory Network



Feedback

The HTA Programme and the authors would like to know your views about this report.

The Correspondence Page on the HTA website (http://www.hta.ac.uk) is a convenient way to publish your comments. If you prefer, you can send your comments to the address below, telling us whether you would like us to transfer them to the website.

We look forward to hearing from you.

The National Coordinating Centre for Health Technology Assessment, Alpha House, Enterprise Road Southampton Science Park Chilworth Southampton SO16 7NS, UK. Fax: +44 (0) 23 8059 5639 Email: hta@hta.ac.uk http://www.hta.ac.uk